#### **ORDINANCE NO. 1508**

#### AN ORDINANCE AUTHORIZING THE MAYOR AND CITY ADMINISTRATOR TO EXECUTE A CONTRACT WITH EAGLE- ELSNER, INC. IN THE AMOUNT OF \$1,042,657.50 FOR CONSTRUCTION OF THE 2019 STREET MAINTENANCE PROGRAM; AND DECLARING AN EMERGENCY

WHEREAS, the City of Canby has heretofore advertised and received seven (7) bids for the 2019 Street Maintenance; and

**WHEREAS**, the notice of call for bids was duly and regularly published in the Oregon Daily Journal of Commerce on March 27, 2019; and

WHEREAS, bids were received and opened on April 18, 2019 at 2:00 pm in the City Hall Conference Room of the City of Canby and the bids were read aloud; and

WHEREAS, the bidders are as listed below and a detailed tabulation of all items is attached herein and summarized as follows:

The summary of cost for each of the seven (7) bidders is shown on the attached tabulation and listed below:

1.	Eagle-Elsner, Inc.	\$1,042,657.50
2.	K & E Paving, Inc.	\$1,153,369.75
3.	North Santiam Paving Company	\$1,276,510.00
4.	S-2 Contractors, Inc.	\$1,278,335.00
5.	Brix Paving NW, Inc.	\$1,297,018.00
6.	Knife River Corporation	\$1,340,647.00
7.	Pacific Excavation, Inc.	\$1, 492,00.00

WHEREAS, the Canby City Council, acting as the City's Contract Review Board, met on Wednesday, May 1, 2019, and considered the bids and reports and recommendations of the City staff, including the staff recommendation that the low responsive bid be selected; and

WHEREAS, the Canby City Council determined that the low responsive bid was that of Eagle-Elsner, Inc.

#### NOW, THEREFORE, THE CITY OF CANBY ORDAINS AS FOLLOWS:

<u>Section 1.</u> The Mayor and/or City Administrator are hereby authorized and directed to make, execute, and declare in the name of the City of Canby and on its behalf, an appropriate

contract with Eagle-Elsner, Inc. for the 2019 Street Maintenance in the amount of \$1,042,657.50. A copy of the contract with, Eagle-Elsner, Inc. is attached hereto and incorporated herein.

<u>Section 2.</u> Due to the large number of ADA ramp replacements, which are part of the project, initiating the project in June ensures it can be completed during the summer construction season. Because of this, an emergency is hereby declared to exist and this ordinance shall therefore take effect immediately upon its enactment after final reading.

**SUBMITTED** to the Canby City Council and read the first time at a regular meeting thereof on Wednesday, May 1, 2019 and ordered posted in three (3) public and conspicuous places in the City of Canby as specified in the Canby City Charter and to come before the City Council for final reading and action at a regular meeting thereof on Wednesday, May 15, 2019, commencing at the hour of 7:00 PM in the Council Meeting Chambers located at 222 NE 2nd Avenue, 1<sup>st</sup> Avenue, Canby, Oregon.

Kimberly Scheafer, MMC City Recorder

**PASSED** on second and final reading by the Canby City Council at a regular meeting thereof on the 15th day of May 2019, by the following vote:

YEAS NAYS ( Brian Hodson Mayor

ATTEST:

Kimberly Scheafer City Recorder

Contract Documents, General Conditions, & Technical Specifications

# **CITY OF CANBY**

# **2019 STREET MAINTENANCE**

Clackamas County, Oregon

April, 2019

CURRAN-MCLEOD, INC., Consulting Engineers 6655 SW Hampton Street, Suite 210 Portland, OR 97223



F1 78 8.2 ر هراي <sup>ا</sup> . العرب ي د ن م د ما ار ا ل ع 

#### CITY OF CANBY 2019 STREET MAINTENANCE

#### CURRAN-McLEOD, INC. CONSULTING ENGINEERS

Addendum No. 1 ************************************	Page 1 of 1	April 5, 2019 *******
No. ITEM ************************************	CHANGE OR CLARIF	ICATION

1. Bid Proposal

2. March 15, 2019 Memorandum

ALL BIDDERS SHALL USE THE ATTACHED REVISED BID PROPOSAL (pages 13 through 19) IN PLACE OF THE BOUND BID PROPOSAL IN THE CONTRACT DOCUMENTS BOOK.

**<u>CLARIFICATION</u>**: S. Elm Street item 3 and S Pine Street item 4, the paving shall be made in two panels with the screed tilted so that the AC depth is 2" at the gutter line to 3" at the crown of the street. One lift of pavement is acceptable.



#### REVISED BID PROPOSAL

TO: Mr. Jerry Nelzen Public Works Operations Supervisor City of Canby 222 NE 2<sup>nd</sup> Avenue Canby, OR 97013

The undersigned is a (indicate one):

#### RESIDENT D NON RESIDENT

bidder as defined by ORS 279A.120, structured as (check one of the following and insert information requested):

□ a. A Corporation organized and existing under laws of the State of \_\_\_\_\_; or

□ b. A Partnership registered under the laws of the State of \_\_\_\_\_; or

□ c. An individual doing business under an assumed name registered under the laws of the State of \_\_\_\_\_\_.

The undersigned bidder declares that the only persons or parties interested in this proposal are those named herein, that this proposal is in all respects fair and without fraud and that it is made without collusion with any representatives of the OWNER.

The bidder further declares that: a) he has examined the plans, specifications, and other proposed contract documents; b) he has determined the extent, character, and location of the proposed Work, the nature and type of excavation to be done, the location and condition of existing streets and roadways giving access to the site of the Work, and topography of the site of the Work; and c) he has personally inspected the site of the Work and has satisfied himself as to the conditions of the Work and understands the listing of materials as included herein is brief and is intended only to associate the said quantities with detailed requirements of the contract documents.

The bidder does hereby propose to furnish all materials, tools, equipment, and appliances, and to perform all labor and work necessary to construct and complete the project entitled:

#### City of Canby 2019 Street Maintenance

and all specified work appurtenant thereto, and in connection with this project for the OWNER with the time limit specified, and in accordance with plans, specifications and change order documents prepared by the ENGINEER for the sums set forth in the following schedule of prices, it being understood that the unit prices are independent of the exact quantities involved and that they represent a true measure of the labor and materials required to perform the Work.

The successful bidder agrees that if this proposal is accepted, bidder will execute the required documents and supply the required submittal information as specified herein, within the time frames established herein.

13

The successful bidder agrees to be substantially complete with all work within 45 days of the Notice to Proceed and achieve final completion within 15 calendar days after issuance of the Certificate of Substantial Completion including punch list items.

In the event the successful bidder fails to achieve Substantial Completion or Final Completion of the project within the time limits specified or extended time limits as agreed upon, liquidated damages shall be paid to the OWNER or deducted from amounts due the Contractor, at the rate of Four Hundred Dollars (\$400.00) per calendar day until Substantial Completion is achieved, and/or until Final Completion is achieved. Sundays and legal holidays shall be excluded in determining the number of days in default.

The City has an estimated available budget not to exceed \$700,000 for this work. The City reserves the right to adjust the basic bid quantities if the submitted low responsive bid exceeds the available budget. All adjusted quantities will be paid based on the bid unit prices.

### BASIC BID SCHEDULE - 2019 Street Resurfacing:

	Cents ( \$ ) per ton.	\$
		Ŧ
	Dollars and	
	joints, at the unit price of:	
	equipment, material tack coat, placement and sand seal	
	PAVEMENT OVERLAY, 4,500 tons, including all labor,	
€.А	2" THICK, ONE LIFT, 1/2" ASPHALT CONCRETE	
	Cents ( \$ ) per ton.	\$
		φ
	Dollars and	
	installation, at the unit price of:	
	<u>300 tons, including all labor, equipment, material and</u>	
2.7	15. VSPHALT CONCRETE PAVEMENT PRELEVEL,	
	Cents ( \$ ) per tump sum.	
	Cents ( \$ ) per lump sum.	\$
	Dollars and	
	at the lump strice of:	
	items listed in G.1 thru G.10 of the Special Requirements,	
	landscaping, clean up, all bonding and insurance costs	
	notifications, final site restoration, incidental grading &	
	mobilization and demobilization, traffic control, public	
	start-up costs, equipment move-in, all associated costs for	
	schedule and contract special requirements, miscellaneous	
	material to complete the work as listed in the basic bid	
Ľ	MOBILIZATION, including all labor, equipment,	

A.4	<u>3" THICK, TWO LIFTS, ½" ASPHALT CONCRETE</u>	
	PAVEMENT OVERLAY, 3,700 tons, including all labor,	
	equipment, material tack coat, placement and sand seal joints, at the unit price of:	
	Dollars and	
	Cents ( <u>\$</u> ) per ton.	\$
A.5	<u>GRIND EXISTING PAVEMENT, (6' WIDE PANEL, 0''-</u> <u>2'' MAX DEPTH), 26.000 lineal feet, including all labor,</u> <u>equipment, material, and disposal of excess pavement</u> <u>material</u> at the unit price of:	
•	Dollars and	
	Cents ( _\$) per lineal foot.	\$
A.6	GRIND EXISTING PAVEMENT NW & NE 2 <sup>nd</sup> AVE, 2" MAX DEPTH, 2,640 square yards, including all labor, equipment, material, and disposal of excess pavement material at the unit price of:	
	Dollars and	
	Cents ( <u>\$</u> ) per square yard.	\$
A.7	EXISTING PAVEMENT REMOVAL/ EXCAVATION, NW & NE 2 <sup>nd</sup> AVE, 2" MAX DEPTH, 710 square yards, including all labor, equipment, material, and disposal of excess pavement material at the unit price of:	
	Dollars and	
	Cents ( \$) per square yard.	\$
A.8	ASPHALT PAVEMENT REPAIR, 150 square yards, including all labor, equipment, material, AC sawcut, removal and disposal of existing pavement, fine grading, placement 4" thick of <sup>1</sup> / <sub>2</sub> " Asphalt Concrete pavement (ACP), ACP will be paid under bid item A.3), sand seal joints, at the unit price of:	
•	Dollars and	
	Cents ( \$ ) per square yard.	\$

\$	Cents ( \$ ) per lineal foot.	
	Dollars and	
	CONCRETE SHORT CURB TYPE "C" at the back of the material, sawcut concrete, remove existing curb, dispose excess material off-site, grade and place 2" of 34"-0" of: of:	EI
\$	Cents ( \$ ) per lineal foot.	
	Dollars and	
	CONCRETE CURB TYPE "C" at the front of the ADA ramp, 750 lineal feet, including all labor, equipment, material, sawcut concrete, remove existing curb, dispose excess material off-site, grade and place 2" of ¾"-0" Crushed Rock, compact and pour new curb, at the unit price of:	7
\$	Cents ( \$ ) per each.	
• •	Dollars and	
	TRUNCATED DOME DETECTABLE WARNING CAST- IN-PLACE MAT, 78 each, including all labor, equipment, material and installation, at the unit price of:	I
\$	Cents ( \$ ) per lineal foot.	
· · · ·	Dollars and	
	<u>CRACK SEAL REPAIR, 50,000 lineal feet, including all</u> <u>labor, equipment, material, pressure cleaning and placement</u> <u>of sealant, at the unit price of:</u>	0
\$	Cents ( \$ ) per square yard.	
	Dollars and	
· · · · · · · · · · · · · · · · · · ·	ASPHALT RECONSTRUCTION, 385 square vards, including all labor, equipment, material, AC sawcut, excavation, disposal of existing pavement and unsuitable material, placement of geotechnical fabric and 12" of compacted Crushed Rock, placement 4" thick of 1/2" Asphalt Concrete pavement (ACP), ACP will be paid under bid item A.3), sand seal joints, at the unit price of:	
	shiev arenos 285 NOITOLIATONODAD TILIIDA	

A.14 <u>4" CONCRETE SIDEWALK, 450 square yards, including</u> <u>all labor, equipment, material, sawcut concrete, remove</u> <u>existing sidewalk, dispose excess material off-site, grade</u> <u>and place 2" of 3/4"-0" Crushed Rock, compact and pour</u> <u>new sidewalk, at the unit price of:</u>

	Dollars and	
	Cents ( _\$) per square yard.	\$
A.15	<u>4" WIDE WHITE OR YELLOW THERMOPLASTIC</u> <u>STRIPE, 11,100 lineal feet including all labor, equipment,</u> <u>material and installation. All material to be thermoplastic</u> (installed complete including layout in the field), at the unit price of: Dollars and	
	Cents ( <u>\$</u> ) per lineal foot.	\$ 
A.16	<u>4" YELLOW BIDIRECTIONAL, TYPE 1 REFLECTORS,</u> <u>10 each, including all labor, equipment, material and</u> <u>installation. (installed complete including layout in the</u> <u>field), at the unit price of:</u>	
	Dollars and	
	Cents ( <u>\$</u> ) per each.	\$
A.17	8 <u>" WIDE WHITE THERMOPLASTIC STRIPE, 7,000</u> lineal feet including all labor, equipment, material and installation. All material to be thermoplastic (installed complete including layout in the field), at the unit price of:	
	Dollars and	
	Cents ( <u>\$</u> ) per lineal foot.	\$ 
A.18	12" or 24" WIDE WHITE THERMOPLASTIC STOP AND CROSSWALK BARS, 2' WIDE x 9' LONG CONTINENTAL CROSSWALK (NON-SKID FINISH) 1,500 square feet, including all labor, equipment, material, and installation. All material to be thermoplastic (installed complete including layout in the field), at the unit price of:	
	Dollars and	
	Cents ( _\$) per square foot.	\$ 

······	(Agent Name & Phone Number)	
	(Name of Surety Company)	
and to one hundred percent	dersigned agrees, if awarded the Contract, to execute and delated agrees, if awarded the Contract forms, an Agreement a contract forms, an Agreement and Construction Payment Bonds each in an amount ed of the Contract sum, using forms provided therefor by the OW of the Construction Performance Bond will be:	(%) ULU U
fo innoms latot shi fo $(\% \mathcal{E})$	canying herewith is Bid Security which is equal to five percent ic Bid.	
\$	<b>LOTAL BASIC BID SCHEDULE:</b>	
\$	Cents ( \$ ) per each.	
	Dollars and	
	installed complete, at the unit price of:	
	plumb V-loc bases and posts, remove, salvage and reinstall existing signs on new posts to the size, kind and all associated accessories such as bolts, nuts, washers, etc.	
	SIGNS POST AND V-LOC BASES, 4 each, including all labor, equipment, and material necessary to provide and	
\$	Cents ( \$ ) per each.	•
	Dollars and	
	<u>LEFT OR RIGHT TURN ARROWS, 4 each, including all</u> <u>labor, material, equipment and installation. All material to</u> <u>be thermoplastic (installed complete including layout in the</u> <u>field)</u> , at the unit price of:	(
\$	Cents ( \$ ) per each.	-
	Dollars and	-
	each, including all labor, material, equipment and installation. All material to be thermoplastic (installed complete including layout in the field), at the unit price of:	
	BIKE ZLENCIL'S' SLOW & HUMP, ADA LEGENDS, 17	

81

The undersigned hereby authorizes said Surety to disclose any information to the OWNER concerning the undersigned's ability to supply a Construction Performance Bond in the amount of the Contract.

The undersigned certifies that (1) this Bid has been arrived at independently and is being submitted without collusion with, and without any agreement, understanding, or planned common course of action with, any other vendor of materials, supplies, equipment, or services described in the Advertisement for Bids designed to limit independent bidding or competition, (2) the contents of this Bid have not been communicated by the undersigned or its employees or agents to any person not an employee or agent of the undersigned or its surety on any bond furnished with the Bid and will not be communicated to such person prior to the official opening of the bids, and (3) if awarded the Contract, he will commence work within ten (10) calendar days after the date of Notice to Proceed and that he will complete the Work within the specified number of days set forth in the Agreement.

The undersigned certifies that he has received and duly considered the following Addenda to the specifications.

Addenda: No.\_\_\_\_\_ to No.\_\_\_\_\_ inclusive.

The undersigned agrees if awarded the Contract, that he will comply with the provisions of the Contract Documents and will comply with the provisions of ORS 279C.800 and Oregon Prevailing Wage Rates.

Name of Firm				
Signature				
Name				-
Address	· · ·			
Telephone No	• .	······································	·	
Construction Contractors Board No.	#			
If Corporation, Attest:				
·····	Secretary of Corpo	ration		

\*If bid is by a partnership, then one of the partners must sign the bid.

# Annla H9J Ylhnoitn9tnl 98nA zihT

- -

CONTRACT DOCUMENTS, GENERAL CONDITIONS AND TECHNICAL SPECIFICATIONS

# **City of Canby**

# **2019 STREET MAINTENANCE**

**Clackamas County, Oregon** 



April 2019

CURRAN-McLEOD, INC. CONSULTING ENGINEERS 6655 S.W. Hampton Street, Suite 210 Portland, Oregon 97223

## City of Canby 2019 Street Maintenance

# TABLE OF CONTENTS

Notice to Contractors1
Instructions to Bidders
Bid Proposal13
Bid Bond 21
First-Tier Subcontractor Disclosure Form
Contract for Construction
Construction Performance Bond
Construction Payment Bond
General Conditions of the Construction Contract
Supplementary General Conditions73
Prevailing Wage Rate Determination

#### **TECHNICAL SPECIFICATIONS**

DIVISI	ON 1 – GENERAL REQUIREMENTS	
	General Requirements	

- 01500 Contractor Facilities and Temporary Controls
- 01700 Project Closeout

DIVISION 2 – SITE WORK

- 02100 Site Preparation
- 02200 Earthwork
- 02500 Paving and Surfacing

#### ATTACHMENTS

- 2019 Street Asphalt Overlay List
- Street Maintenance Scope
- Street Location Map
- Construction Details

i

This Page Intentionally Left Blank

зJ

120

ند ...

Ĵ

 $\mathcal{L} \subseteq \mathcal{L} = \{ x_i, x_i, x_i \}$ 

\$

-

-

# NOTICE TO CONTRACTORS

# NOTICE TO CONTRACTORS

d

37

لين ا

بر -راح .

ت م لد ا

-1

ر.

ġ

#### NOTICE TO CONTRACTORS

Sealed bids for *2019 Street Maintenance* will be received by the Public Works Supervisor of the City of Canby until 2:00 P.M on Thursday, April 18, 2019 at which time bidding will be closed.

Pursuant to ORS 279C.370(2), within two (2) working hours of the Bid Closing all Bidders shall submit to the City a disclosure form as furnished with these bid documents identifying any first-tier subcontractors (those Entities that would be contracting directly with the prime contractor) that will be furnishing labor or materials on the Contract, if awarded, whose subcontract value would be equal to or greater than:

- Five percent of the total Contract Price, but at least \$15,000, or
- \$350,000 regardless of the percentage of the total Contract Price.

Bids and disclosure forms shall be addressed to Mr. Jerry Nelzen, Public Works Operations Supervisor, City of Canby, 222 NE 2<sup>nd</sup> Avenue, Canby, Oregon 97013. At 2:00 P.M., bids will be opened and read aloud in the City Council Chambers, Canby, Oregon. Only bids for which a disclosure form has been submitted within two (2) working hours of the Bid Closing date and time will be considered responsive.

The scope of work generally consists of the following:

- Approximately 8,000 tons of asphaltic concrete pavement at 2" or 3" thickness.
- Approximately 26,000 lineal feet of 6-foot wide panel grinding at 0"-2" depth and approximately 24,000 SF area at 2" depth.
- Retrofit an estimated 42 ADA ramps, including curbs, sidewalks, and 3 pedestrian midblocks.
- Restore approximately 17,000 lineal feet of 4" and 8" wide stripes, including legends such as arrows, continental crosswalks, stop bars, bike stencils, etc.
- Repair an approximate 50,000 lineal feet of joined crack sealing.

The City has an estimated available budget not to exceed \$700,000 for this work. The City reserves the right to adjust the basic bid quantities if the submitted low responsive bid exceeds the available budget. All adjusted quantities will be paid based on the bid unit prices.

Specifications are available at the offices of CURRAN-McLEOD, INC., Consulting Engineers, 6655 S.W. Hampton Street, Suite 210, Portland, Oregon, 97223, telephone: (503) 684-3478. All questions related to the project shall be directed to Hassan Ibrahim of CURRAN-McLEOD, INC. Electronic .PDF copies are available without charge for contractors and material suppliers upon registration with the Engineer. (Please email cmi@curran-mcleod.com to register and request an electronic copy). A forty- dollar (\$40.00) non-refundable fee is required with each request for hard copies. Specifications may be examined at the following locations:

City of Canby	222 NE 2 <sup>nd</sup> Avenue	Canby, OR
Central Oregon Builders Exchange	1902 NE 4 <sup>th</sup>	Bend, OR
Daily Journal of Commerce – OR	Electronic Only - (503) 274-0624	Portland, OR
Eugene Builders Exchange	2460 West 11 <sup>th</sup>	Eugene, OR
McGraw-Hill Construction/Dodge	3461 NW Yeon Ave	Portland, OR
Oregon Contractor Plan Center	5468 SE International Way	Clackamas, OR

Corvallis, OR	33862 SE Eastgate Cr.	Willamette Valley Bid Center
Vancouver, WA	J01J N'E' H <sup>M</sup> 66' #714	SW Washington Contractors Assoc.
Salem, OK	5226 Judson St. S.E.	Salem Contractors Exchange

Bidders must be pre-qualified to bid on this project. Bidders not currently pre-qualified by the Owner, Engineer or the Oregon Department of Transportation may apply for pre-qualification with the Owner or Engineer prior to the bid opening on forms available or similar to those available from the Oregon Department of Transportation.

No bid shall be received or considered unless the Bidder is registered with the Construction Contractors Board as required by ORS chapter 701.021.

To be considered, bids must be completed on the bidding forms provided, in the manner prescribed in the bidding documents and accompanied by a 5% bid security in favor of the City of Canby. Each bidder must indicate in the space provided on the Bid Form, whether they are a Resident or Non-Resident Bidder pursuant to ORS 279A.120.

This public works project is subject to the provisions of ORS 279C.800 to ORS 279C.870. By prices. Contractors need not be licensed for asbestos handling per ORS 468A.720 to complete this project. A pre-bid conference will not be held for this project.

أينه

The City of Canby reserves the right to reject any or all bids not in compliance with all prescribed public bidding procedures and requirements, may reject for good cause any and all bids upon a finding of the agency it is in the public interest to do so, and may waive all informalities. No bidder may withdraw or modify his bid prior to the lapse of thirty (30) days after bid opening.

By Order of the City Council of the City of Canby.

Published in the Portland Daily Journal of Commerce, March 27, 2019

**INSTRUCTIONS TO BIDDERS** 

INSTRUCTIONS TO BIDDERS

أيد

3

ز. ا

(ت ک لیر ب

لى يا

. ....

i. ......

- *2* 

#### **INSTRUCTIONS TO BIDDERS**

#### **ARTICLE 1 - DEFINED TERMS**

1.1 Terms used in these Instructions to Bidders which are defined in the Standard General Conditions of the Construction Contract (No. C-700, 2013 Ed.) have the meanings assigned to them in the General Conditions. The term "Bidder" means one who submits a Bid directly to OWNER, as distinct from a subbidder, who submits a bid to a Bidder. The term "Successful Bidder" means the lowest, qualified, responsible and responsive Bidder to whom OWNER (on the basis of OWNER'S evaluation as hereinafter provided) makes an award. The term "Bidding Documents" includes the Advertisement or Invitation to Bid, Instructions to Bidders, the Bid Form, and the proposed Contract Documents (including all Addenda issued prior to receipt of Bids).

#### **ARTICLE 2 - COPIES OF BIDDING DOCUMENTS**

- 2.1 Complete sets of Bidding Documents in the number and for the sum, if any, stated in the Advertisement or Invitation to Bid may be obtained from ENGINEER.
- 2.2 Complete sets of Bidding Documents must be used in preparing Bids. Neither OWNER nor ENGINEER assume any responsibility for errors or misinterpretations resulting from the use of incomplete sets of Bidding Documents.
- 2.3 OWNER and ENGINEER in making copies of Bidding Documents available on the above terms do so only for the purpose of obtaining Bids on the Work and do not confer a license or grant for any other use.

#### **ARTICLE 3 - QUALIFICATIONS OF BIDDERS**

3.1 Bidders must be pre-qualified to bid on this project. Bidders not currently pre-qualified for this Contract by the OWNER, ENGINEER or the Oregon State Department of Transportation may apply for pre-qualification by filing a completed pre-qualification statement with the ENGINEER on a form prescribed by the Oregon State Department of Transportation. Bidders filing pre-qualification applications with the ENGINEER should file the statement no later than the scheduled date for receipt of Bids.

#### **ARTICLE 4 - EXAMINATION OF CONTRACT DOCUMENTS AND SITE**

4.1 It is the responsibility of each Bidder before submitting a Bid, to (a) examine the Contract Documents thoroughly, (b) visit the site to become familiar with local conditions that may affect cost, progress, performance, or furnishing of the Work, (c) consider Federal, State and Local Laws and Regulations that may affect cost, progress, performance or furnishing of the Work, (d) study

and carefully correlate Bidder's observations with the Contract Documents, and (e) notify ENGINEER of all conflicts, errors, or discrepancies in the Contract Documents.

Information and data reflected in the Contract Documents with respect to Underground Facilities at or contiguous to the site is based upon information and data furnished to OWNER and ENGINEER by owners of such Underground Facilities or others, and OWNER does not assume responsibility for the accuracy or completeness thereof unless it is expressly provided otherwise in the Supplementary Conditions.

4.3 Provisions concerning responsibilities for the adequacy of data furnished to prospective Bidders on subsurface conditions, underground facilities and other physical conditions, and possible changes in the Contract Documents due to differing conditions appear in Article 5 of the General Conditions.

Before submitting a Bid each Bidder will be responsible to make or obtain such explorations, tests, and data concerning physical conditions (surface, subsurface, and Underground Facilities) at or contiguous to the site, or otherwise which may affect cost, progress, performance, or furnishing of the Work and which Bidder deems necessary to determine its Bid for performing and furnishing the Work in accordance with the time, price, and other terms and conditions of the Contract Documents.

On request in advance, OWNER will provide each bidder access to the site to conduct such explorations and tests as Bidder deems necessary for submission of a Bid. Bidder shall fill all holes, clean up, and restore the site to its former condition upon completion of such explorations.

The lands upon which the Work is to be performed, rights-of-way and easements for access thereto and other land designated for use by the Contractor in performing the Work are identified in the Contract Documents. All additional lands and access thereto required for temporary construction facilities or storage of materials and equipment are to be provided by Contractor. Easements for permanent structures or permanent changes in existing structures are to be obtained and paid for by OWNER unless otherwise provided in the Contract Documents.

The submission of a bid will constitute an incontrovertible representation by Bidder that Bidder has complied with every requirement of this ARTICLE 4, that without exception the Bid is premised upon performing and furnishing the Work required by the Contract Documents and such means, methods, techniques, sequences, or procedures of construction as may be indicated in or required by the Contract Documents, and that the Contract Documents are sufficient in scope and detail to indicate and convey understanding of all terms and conditions for performance and furnishing of the Work to complete the project.

**VELICEE 5 - INTERPRETATIONS AND ADDENDA** 

L't

9.4

5.4

ヤヤ

4.2

 $\mathbf{v}$ 

- 5.1 All questions about the meaning or intent of the Contract Documents are to be directed to the ENGINEER. Interpretations or clarifications considered necessary by the ENGINEER in response to such questions will be issued by Addenda mailed or delivered to all parties recorded by ENGINEER as having received the Bidding Documents. Questions received less than three days prior to the date for the opening of Bids may not be answered. Only questions answered by formal written Addenda will be binding. Oral and other interpretations or clarifications will be without legal effect.
- 5.2 Addenda may also be issued to modify the Bidding Documents as deemed advisable by OWNER or ENGINEER.

#### **ARTICLE 6 - BID SECURITY**

6.1

6.2

Bid Security in the amount of five percent (5%) of the Base Bid must accompany each Bid proposal. Bid Security may be in the form of a surety bond or cashier's check, but if it is a surety bond, it must comply with the requirements of Article 6 of the General Conditions. If a surety bond is submitted as Bid Security, it shall be written on EJCDC NO. C-430, Bid Bond, or similar instrument, and the attorney-in-fact who executes the bond shall affix to the bond a current copy of his power of attorney. Surety companies executing bonds must appear on the U.S. Treasury Department's most current list (Circular 570 as amended) and be authorized to transact business in the state where the project is located.

The Bid security of the top three Bidder will be retained until the successful Bidder has executed the Agreement and furnished the required contract security, whereupon the Bid Securities will be returned. If the Successful Bidder fails to execute and deliver the Agreement and furnish the required contract security within ten (10) days after the Notice of Award, OWNER may annul the Notice of Award and the Bid Security of that Bidder will be forfeited. The Bid Security of other Bidders whom the OWNER believes to have a reasonable chance of receiving the award` may be retained by the OWNER until the earlier of the seventh day (7th) day after the Effective Date of the Agreement or the thirty first (31<sup>st</sup>) day after the Bid opening, whereupon Bid Security furnished by such Bidders will be returned. Bid Security and Bids which are not competitive will be returned within seven (7) days after the Bid opening.

#### **ARTICLE 7 - CONTRACT TIME**

7.1 The number of days within which, or the dates by which, the Work is to be substantially completed and also completed and ready for final payment (the Contract Time) are set forth in the Agreement.

#### **ARTICLE 8 - LIQUIDATED DAMAGES**

Provisions for liquidated damages are set forth in the Agreement. Delay damages will be assessed for failure to achieve substantial completion AND/OR final completion as defined in the General Conditions. Liquidated damages shall be paid by the Contractor to the Owner or the OWNER may withhold damages from any amounts due to the Contractor.

### VELICIE 9 - SUBSTITUTING "OR-EQUAL" ITEMS

1.6

1.8

manner. prospective Bidders. Bidders shall not rely upon approvals made in any other substitution, such approval will be set forth in an Addendum issued to all substitution shall be final. IT ENGINEER approves any proposed The ENGINEER'S decision of approval or disapproval of a proposed burden of proof of the merit of the proposed substitution is upon the Bidder. that incorporation of the substitution would require shall be included. The statement setting forth any changes in other materials, equipment, or Work and test data and any other information necessary for an evaluation. A description of the proposed substitute, including drawings, cuts, performance of the material or equipment for which it is to be substituted and a complete prior to the date for receipt of Bids. Each such request shall include the name the Bidder and has been received by the ENGINEER at least seven (7) days will be considered unless written request for approval has been submitted by verbiage in the specific specification unless noted otherwise. No substitution specified "or equal", without regard to the presence or lack of "or equal" by any proposed substitution. All materials and equipment shall be as standard of required function, dimension, appearance, and quality to be met The materials and equipment described in the Bidding Documents establish a

#### **ARTICLE 10 - SUBCONTRACTORS, SUPPLIERS AND OTHERS**

- 10.1 The OWNER may require the identity of certain Subcontractors, Suppliers, and other material and equipment) to be submitted to OWNER in advance of the Notice of Matd. The apparent Successful Bidder, and any other Bidder if requested shall within seven (7) days after the Bid opening submit to the OWNER a list of all such portions of the Work for which such identification is required. Such list shall be accompanied by an experience statement with pertinent information regarding similar projects and other evidence of qualification of each such Subcontractor, Suppliers, and other bersons and organizations proposed for those portions of the Work for which such identification is required. Such list shall be provident and other evidence of qualification of each such Subcontractor, Supplier, person or organization if requested by the OWNER.
- 10.2 If OWNER or ENGINEER after due investigation has reasonable objection to any proposed Subcontractor, Supplier, other person or organization, either may before the Notice of Award is given request the apparent Successful Bidder to submit an acceptable substitute in which case the apparent Successful Bidder shall submit an acceptable substitute, that Bidder's bid price will be increased (or decreased) by the

difference in cost occasioned by such substitution and OWNER may consider such price adjustment in evaluating Bids and making the contract award. If apparent Successful Bidder declines to make any such substitution, OWNER may award the contract to the next lowest Bidder that proposes to use acceptable Subcontractors, Suppliers, and other persons and organizations. The declining to make requested substitutions will not constitute grounds for sacrificing the Bid Security of any Bidder.

- 10.3 Any Subcontractor, Supplier, other person or organization listed and to whom OWNER or ENGINEER does not make written objection prior to the giving of the Notice of Award will be deemed acceptable to OWNER and ENGINEER, subject to revocation of such acceptance after the Effective Date of the Agreement as provided in Paragraph 7.06 of the General Conditions.
- 10.4 CONTRACTOR shall not be required to employ any Subcontractor, Supplier, individual, or entity against whom CONTRACTOR has reasonable objection.

#### **ARTICLE 11 - BID FORM**

- 11.1 The Bid Form is included within the Contract Documents.
- 11.2 Bid Forms must be completed in ink or by typewriter. The Bid price of each item on the form must be stated in numerals.
- 11.3 Bids by corporations must be executed in the corporate name by the president or a vice-president (or other corporate officer accompanied by evidence of authority to sign) and the corporate seal must be affixed and attested by the secretary or an assistant secretary. The corporate address and state of incorporation must be shown below the signature.
- 11.4 Bids by partnerships must be executed in the partnership name and signed by a partner, whose title must appear under the signature and the official address of the partnership must be shown below the signature.
- 11.5 A Bid by a limited liability company shall be executed in the name of the firm by a member and accompanied by evidence of authority to sign. The state of formation of the firm and the official address of the firm must be shown below the signature.
- 11.6 A Bid by an individual shall show the Bidder's name and official address.
- 11.7. A Bid by a joint venture shall be executed by each joint venturer in the manner indicated on the Bid form. The official address of the joint venture must be shown below the signature.
- 11.8 All names must be typed or printed below the signature.
- 11.9 The bid shall contain an acknowledgment of receipt of all Addenda (the numbers of which must be filled in on the Bid Form).

11.10 The CCB Registration, business address and telephone number for communications regarding the Bid must be shown.

#### **VELICIE 12 - SUBMISSION OF BIDS**

12.1 To be responsive, bids shall be submitted at the time and place identified in the Bidders and shall be enclosed in an opaque sealed envelope, marked with the Project Title (and, if applicable, the designated portion of the Project for which the Bid is submitted) and name and address of the Bidder and accompanied by the Bid Security and other required documents. If the Bid is sent through the mail or other delivery system, the sealed envelope shall be enclosed in a separate envelope with the notation of "BID ENCLOSED" on the face of it.

12.2 To be responsive CONTRACTOR must complete, sign and submit the following documents:

- a. Bid Proposal
- b. Bid Bond
- c. First Tier Disclosure

# **VELICLE 13 - MODIFICATION AND WITHDRAWAL OF BIDS**

- 13.1 Bids may be modified or withdrawn by an appropriate document duly executed (in the manner that a Bid must be executed) and delivered to the place where the Bids are to be submitted at any time prior to the opening of Bids.
- 13.2 If, within twenty-four hours after Bids are opened, any Bidder files a duly signed, written notice with OWNER that there was a material and substantial mistake in the preparation of OWNER that there was a material and substantial mistake in the returned. Thereafter, that Bidder may withdraw its Bid and the Bid Security will be Work to be provided under the Contract Documents.

#### **VELICIE 14 - OFENING OF BIDS**

 Bids will be opened and read aloud publicly. An abstract of the amounts of the base Bids and major alternates (if any) may be made available to Bidders after the opening of the Bids.

# **VELICLE 15 - BIDS TO REMAIN SUBJECT TO ACCEPTANCE**

15.1 All bids will remain subject to acceptance for thirty (30) days after the day of the Bid opening, but OWNER may, in its sole discretion, release any Bid and return the Bid Security prior to that date.

#### **ARTICLE 16 - AWARD OF CONTRACT**

- 16.1 OWNER reserves the right to reject any and all Bids, to waive any and all informalities not involving price, time, or changes in the Work and to negotiate contract terms with the Successful Bidder, and the right to disregard all nonconforming, non-responsive, unbalanced, or conditional Bids. Also, OWNER reserves the right to reject the Bid of any Bidder if OWNER determines the Bid to be non-responsive per ORS 279B.110. In the event of discrepancy between the written and numerical amounts the written prices will govern. Discrepancies in the multiplication of units of Work and unit price will be resolved in favor of the unit prices. Discrepancies between the indicated sum of any column of figures and the correct sum thereof will be resolved in favor of the correct sum.
- 16.2 In evaluating Bids, OWNER will consider the qualifications of the Bidders, whether or not the Bids comply with the prescribed requirements, and such alternates, unit prices, and other data, as may be requested in the Bid Form or prior to the Notice of Award.
- 16.3 OWNER may consider the qualifications and experience of Subcontractors, Suppliers, and other persons and organizations proposed for those portions of the Work as to which the identity of Subcontractors, Suppliers, and other persons and organizations must be submitted as provided in the Supplementary Conditions.

OWNER also may consider the operating costs, maintenance requirements, performance data, and guarantees of major items of materials and equipment proposed for incorporation in the Work when such data is required to be submitted prior to the Notice of Award.

- 16.4 OWNER may conduct such investigations as OWNER deems necessary to assist in the evaluation of any Bid and to establish the responsibility, qualifications and financial ability of Bidders, proposed Subcontractors, Suppliers, and other persons and organizations to perform and furnish the Work in accordance with the Contract Documents to OWNER'S satisfaction within the prescribed time.
- 16.5 If the contract is to be awarded, the award will be based upon the lowest responsive/responsible bid for the total of the Basic Bid and such alternatives that in the sole judgement of the OWNER will best serve its interests.
- 16.6 If the contract is to be awarded, OWNER will give the Successful Bidder a Notice of Award within thirty (30) days after the Bid opening.
- 16.7 More than one Bid for the same Work from an individual or entity under the same or different names will not be considered. Reasonable grounds for believing that any Bidder has an interest in more than one Bid for the Work may be cause for disqualification of that Bidder and the rejection of all Bids in which that Bidder has an interest.

#### **VELICLE 17 - CONTRACT SECURITY**

17.1 Prior to execution of the Contract, the Bidder shall furnish separate bonds covering the faithful performance of the Contract and the payment of all obligations arising thereunder, each in an amount equal to 100 percent of the Contract Sum. The cost of bonds shall be licensed to issue bonds in the Bid. The Surety issuing such bonds shall be licensed to issue bonds in the Gate of Oregon. Such bonds shall comply with the provisions of Article 6 of the General Conditions.

#### **VBLICTE 18 - SIGNING OF AGREEMENT**

18.1 When OWNER gives a Notice of Award to the Successful Bidder, it will be accompanied by three unsigned counterparts of the Agreement with all other written Contract Documents attached. Within ten (10) days thereafter, Contractor shall sign and deliver the counterparts of the Agreement and attached documents to OWNER with the required Bonds. Within twenty-five (25) days thereafter, OWNER shall deliver one fully signed counterpart to Contractor. Each counterpart shall include a complete set of Drawings attached by reference.

#### **VBLICTE 10 - CONFERENCES**

#### 16'I **BKE-BID CONFERENCE**

If a pre-bid conference is specified in the Notice to Contractors, it shall be non-mandatory. If so scheduled, representatives of OWNER and ENGINEER will be present to discuss the Project. Bidders are encouraged to attend and participate in the conference. ENGINEER will transmit to all prospective Bidders of record such Addenda as ENGINEER considers necessary in response to questions arising at the conference. Oral statements may not be the pre-bid meeting will be provided by the Engineer. the pre-bid meeting will be provided by the Engineer.

#### 19.7 **BKECONZLKNCLION CONFERENCE**

Either before or soon after the actual award of the Contract (but in any event prior to the start of Work at the site), the Contractor or his representative, and its subcontractors, shall attend a Preconstruction Conference with representatives of the OWNER, the funding agencies and the ENGINEER. The Conference will be held for review and acceptance of schedules referred to in paragraph 2.04 of the General Conditions, to establish procedures for handling shop drawings and other submittal and for processing applications for payment, and to acquaint the participants with the general plan of contract administration and requirements under which the construction operation is to proceed. The date, time, and place of the Conference will be furnished to the Contractor by the OWNER. The Contractor will notify his subcontractors of the Conference and require their attendance.

#### **ARTICLE 20 - IDENTIFICATION**

- 20.1 The word "OWNER" as used herein refers to the CITY OF CANBY, as identified in the proposal, with whom the CONTRACTOR will enter into an agreement for the work. The OWNER shall be responsible for actions requested by the ENGINEER including payment in accordance with the terms of the Contract.
- 20.2 The word "ENGINEER" as used throughout these documents refers to the firm of CURRAN-McLEOD, INC. Consulting Engineers. All correspondence, notifications and requests of the OWNER by the CONTRACTOR shall be through the ENGINEER.
- 20.3 The word "CONTRACTOR" refers to the corporation, partnership or sole proprietorship which enters into a contractual obligation with the OWNER to complete the work. Subcontractors shall not be recognized.

#### **ARTICLE 21 - FIRST - TIER SUBCONTRACTOR DISCLOSURE**

- 21.1 Within two working hours after the date and time of the deadline when bids are due to a contracting agency for a public improvement contract, a bidder shall submit to the contracting agency a disclosure of the first-tier subcontractors that:
  - (A) Will be furnishing labor or will be furnishing labor and materials in connection with the public improvement contract; and
  - (B) Will have a contract value that is equal to or greater than five percent of the total project bid, or \$15,000, whichever is greater, or \$350,000 regardless of the percentage of the total project bid.
- 21.2 The disclosure of first-tier subcontractors under this subsection must include the name of each subcontractor, the category of work that each subcontractor will perform and the dollar value of each subcontract. The information shall be disclosed in the form provided in this bid document.
  - 21.3 If the Bidder will not be using any Subcontractors that are subject to the above disclosure requirements, you are required to indicate "NONE" on the accompanying form.
  - 21.4 THE OWNER MUST REJECT A BID IF THE BIDDER FAILS TO SUBMIT THE DISCLOSURE FORM WITH THIS INFORMATION BY THE STATED DEADLINE, EVEN IF NO SUBCONTRACTORS ARE LARGE ENOUGH TO BE DISCLOSED PER OAR 125-249-0360.

And Page Intentionally Left Blank

-

لہ ۔

ۇند ،

أورز

.....

<u>ند</u>

**BID PROPOSAL** 

BID **BKOPOSAL** 

. ...

لتب

#### **REVISED BID PROPOSAL**

TO: Mr. Jerry Nelzen Public Works Operations Supervisor City of Canby 222 NE 2<sup>nd</sup> Avenue Canby, OR 97013

The undersigned is a (indicate one):

RESIDENT D NON RESIDENT

April 18, 2019

bidder as defined by ORS 279A.120, structured as (check one of the following and insert information requested):

a. A Corporation organized and existing under laws of the State of <u>Oregon</u>; or

□ b. A Partnership registered under the laws of the State of \_\_\_\_\_; or

□ c. An individual doing business under an assumed name registered under the laws of the State of \_\_\_\_\_.

The undersigned bidder declares that the only persons or parties interested in this proposal are those named herein, that this proposal is in all respects fair and without fraud and that it is made without collusion with any representatives of the OWNER.

The bidder further declares that: a) he has examined the plans, specifications, and other proposed contract documents; b) he has determined the extent, character, and location of the proposed Work, the nature and type of excavation to be done, the location and condition of existing streets and roadways giving access to the site of the Work, and topography of the site of the Work; and c) he has personally inspected the site of the Work and has satisfied himself as to the conditions of the Work and understands the listing of materials as included herein is brief and is intended only to associate the said quantities with detailed requirements of the contract documents.

The bidder does hereby propose to furnish all materials, tools, equipment, and appliances, and to perform all labor and work necessary to construct and complete the project entitled:

#### City of Canby 2019 Street Maintenance

and all specified work appurtenant thereto, and in connection with this project for the OWNER with the time limit specified, and in accordance with plans, specifications and change order documents prepared by the ENGINEER for the sums set forth in the following schedule of prices, it being understood that the unit prices are independent of the exact quantities involved and that they represent a true measure of the labor and materials required to perform the Work.

The successful bidder agrees that if this proposal is accepted, bidder will execute the required documents and supply the required submittal information as specified herein, within the time frames established herein.

Substantial Completion including punch list items. to Proceed and achieve final completion within 15 calendar days after issuance of the Certificate of The successful bidder agrees to be substantially complete with all work within 45 days of the Notice

number of days in default. Final Completion is achieved. Sundays and legal holidays shall be excluded in determining the Hundred Dollars (\$400.00) per calendar day until Substantial Completion is achieved, and/or until shall be paid to the OWNER or deducted from amounts due the Contractor, at the rate of Four project within the time limits specified or extended time limits as agreed upon, liquidated damages In the event the successful bidder fails to achieve Substantial Completion or Final Completion of the

budget. All adjusted quantities will be paid based on the bid unit prices. the right to adjust the basic bid quantities if the submitted low responsive bid exceeds the available The City has an estimated available budget not to exceed \$700,000 for this work. The City reserves

#### BASIC BID SCHEDULE - 2019 Street Resurfacing:

Dollars and Ninety-One installation, at the unit price of: 300 tons, including all labor, equipment, material and 12 VIII CONCRETE PAVEMENT PRELEVEL 7.A 000 201 \$ Zevo Cents ( \$ 103,000<sup>20</sup>) per lump sum. One Hundred Three Thousand Dollars and at the lump sum price of: items listed in G.1 thru G.10 of the Special Requirements, landscaping, clean up, all bonding and insurance costs notifications, final site restoration, incidental grading & mobilization and demobilization, traffic control, public start-up costs, equipment move-in, all associated costs for schedule and contract special requirements, miscellaneous material to complete the work as listed in the basic bid MOBILIZATION, including all labor, equipment, I.A

5 51300 00

omt- Ktris ioints, at the unit price of: equipment, material tack coat, placement and sand seal PAVEMENT OVERLAY, 4,500 tons, including all labor, 2" THICK, ONE LIFT, 1/3" ASPHALT CONCRETE £.A Zero Cents ( \$ 9/20 ) per ton.

Cents ( \$ 6200 ) per ton.

- 000622 \$

Dollars and
<u>3" THICK, TWO LIFTS, ½" ASPHALT CONCRETE</u> A.4 PAVEMENT OVERLAY, 3,700 tons, including all labor, equipment, material tack coat, placement and sand seal joints, at the unit price of:  $\frac{5ixt_y - 0ixe}{20ixt_y - 0ixe}$  Dollars and  $\frac{20ixt_y - 0ixe}{20ixt_y - 0ixe}$  Dollars and  $\frac{20ixt_y - 0ixe}{20ixt_y - 0ixe}$  Dollars and  $\frac{20ixt_y - 0ixe}{20ixt_y - 0ixe}$ \$ 225,700 -A.5 **GRIND EXISTING PAVEMENT**, (6' WIDE PANEL, 0"-2" MAX DEPTH), 26.000 lineal feet, including all labor, equipment, material, and disposal of excess pavement material at the unit price of:  $\frac{T\omega o}{Forty - Five} \text{ Cents } (\underline{\$ 245}) \text{ per lineal foot.}$ \$ 63,70000 A.6 GRIND EXISTING PAVEMENT NW & NE 2nd AVE, 2" MAX DEPTH, 2,640 square yards, including all labor, equipment, material, and disposal of excess pavement material at the unit price of:  $\frac{Four}{Four} \qquad Dollars and$   $\frac{Four}{fy-Four} \quad Cents \quad (\$ 4.75) \quad per square yard. \quad \$ 12,540^{\circ 0}$ A.7 EXISTING PAVEMENT REMOVAL/ EXCAVATION. NW & NE 2<sup>nd</sup> AVE, 2" MAX DEPTH, 710 square yards, including all labor, equipment, material, and disposal of excess pavement material at the unit price of:  $\frac{F_{our}}{5eventy-Five} \text{ Cents } (\$4^{75}) \text{ per square yard.} \$ 3372^{50}$ A.8 ASPHALT PAVEMENT REPAIR, 150 square yards, including all labor, equipment, material, AC sawcut, removal and disposal of existing pavement, fine grading, placement 4" thick of 1/2" Asphalt Concrete pavement (ACP), ACP will be paid under bid item A.3), sand seal joints, at the unit price of: Forty-FiveDollars andZeroCents ( $$450^{\circ}$ ) per square yard.\$6750^{\circ}

	) per lineal foot.	507 \$ )	Cents	26LQ
0 <sup>p</sup>	Dollars and		Kp-10-1	<u></u>
				:ło
əc	v curb, at the unit pric	ipact and pour nev	ed Rock, com	Grush
	ice 2" of 34"-0"	site, grade and pla	fto lanaterial off-	SSADXA
	all labor, equipment			
	<u>'C" at the back of the 'C'</u>			
4		550 <u>5</u> \$)	ents	
00 20 20 20 20	Dollars and		tot -	<del>,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,</del>
			•	:to
a	verte, at the unit price	pact and pour new	ED KOCK' COM	0
	$\frac{1}{1000} = \frac{1}{1000} = \frac{1}{1000} = \frac{1}{1000} = \frac{1}{10000} = \frac{1}{10000000000000000000000000000000000$	site, grade and pla	- <u>Tto Isinatem</u>	<u>excess</u>
	sting curb, dispose	ncrete, remove ex	al, sawcut cor	materi
	or, equipment,	t, including all lal	<u>750 lineal fee</u>	tamp,
	AUA of the ADA	3 TYPE "C" at the	SETE CUR	IS CONC
5, 9+16+ \$		50295)		
<b>U</b> *	Dollars and	And the bar	ix Flun	\$
		ution, at the unit p		
	all labor, equipment,			
	E WARNING CAST	ME DETECTABI	CATED DOI	II TRUN
5000'0L \$	) per lineal foot.	F1 (\$)	cents	64-10-1
( <i>a</i>	Dollars and	à	0 N	
			ant, at the uni	
	leaning and placemen			
	al feet, including all	AIR, 50,000 line	K SEAL REI	10 CEAC
505E'2+ \$	) bet square yard			
	Dollars and	not bo	-pung zu(	<u>.</u>
	init price of:	r oft is <u>ear joints</u> , at the r	bass ,(E.A n	<u>bid</u> iter
<u>.</u>	CP will be paid unde	vement (ACP), A	t Concrete pa	lenge
	$4^{11}$ , thick of $1/2^{11}$	Rock, placement	cted Crushed	COMDA
	nent and unsuitable bric and 12" of	of geotechnical fa		
	eldeturgan hae tage	adiana maitaina to	1	
		instent, materi		Ipnjour

4

أر

1

نہ ۔

3

لندب

۲ ۲ . لا تا

j.

ŝ

A.14	4" CONCRETE SIDEWALK, 450 squa	re vards, including	
	all labor, equipment, material, sawcut co	oncrete, remove	
	existing sidewalk, dispose excess materi	al off-site, grade	
	and place 2" of 34"-0" Crushed Rock, co	mpact and pour	
	new sidewalk, at the unit price of:		
	One Hundred Fifty	Dollars and	
	One Hundred Fifty Zero Cents (\$ 15000	) per square yard.	\$ 67,50000
A.15	4" WIDE WHITE OR YELLOW THER STRIPE, 11,100 lineal feet including all material and installation. All material to (installed complete including layout in th price of:	<u>MOPLASTIC</u> labor, equipment, be thermoplastic he field), at the unit	
	Zero	Dollars and	
	Zero Ninety Cents ( <u>\$ 090</u>	) per lineal foot.	\$ 9990 -
A.16		1 REFLECTORS, naterial and	
	Sixteen	Dollars and	
	Zero Cents ( <u>\$ 160</u>		\$ 1600
A.17	8" WIDE WHITE THERMOPLASTIC S lineal feet including all labor, equipment installation. All material to be thermoplas complete including layout in the field), a	<u>, material and</u> stic (installed	
	One	Dollars and	
7	Thirty Cents (\$ 130)	per lineal foot.	\$ 910000
	12" or 24" WIDE WHITE THERMOPLA AND CROSSWALK BARS, 2' WIDE x CONTINENTAL CROSSWALK (NON- 1,500 square feet, including all labor, equ and installation. All material to be thermo- complete including layout in the field), at	ASTIC STOP 9' LONG SKID FINISH) ipment, material, oplastic (installed	
	Field	Dollars and	
-	CITH	_ Donars and	•
F	Eight Pteen Cents ( <u>\$ 815</u> )	per square foot.	\$ 12,225 2

nal to one hundred percent	idersigned agrees, if awarded the Contract, to execute and del (15) days after receiving the Contract forms, an Agreement an mance and Construction Payment Bonds each in an amount eq ) of the Contract sum, using forms provided therefor by the OW ethe Construction Performance Bond will be:	00% Leen
	panying herewith is Bid Security which is equal to five percent of the bard.	seđ s
09 199'200'1\$	<b>LOTAL BASIC BID SCHEDULE:</b>	
- 0021 \$	Cents ( \$360 00 ) per each.	
	Three Mundred Dollars and	
	SIGNS POST AND V-LOC BASES, 4 each, including all labor, equipment, and material necessary to provide and plumb V-loc bases and posts, remove, salvage and reinstall existing signs on new posts to the size, kind and all associated accessories such as bolts, nuts, washers, etc. installed complete, at the unit price of:	17.
20021\$	Zero Cents ( \$ 300 00 ) per each.	
09	Three Mandred Dollars and	
	LEFT OR RIGHT TURN ARROWS, 4 each, including all labor, material, equipment and installation. All material to be thermoplastic (installed complete including layout in the field), at the unit price of:	02.
	Zero Cents ( \$ 290 - ) per each.	
	BIKE STENCILS, SLOW & HUMP, ADA LEGENDS, 17 each, including all labor, material, equipment and installation. All material to be thermoplastic (installed complete including layout in the field), at the unit price of:	61.

· --

لند بنا

The undersigned hereby authorizes said Surety to disclose any information to the OWNER concerning the undersigned's ability to supply a Construction Performance Bond in the amount of the Contract.

(Name of Surety Company)

Ancher Ins. 45 weby Co. Inc. Jeed Dictzman 503-224-2500

The undersigned certifies that (1) this Bid has been arrived at independently and is being submitted without collusion with, and without any agreement, understanding, or planned common course of action with, any other vendor of materials, supplies, equipment, or services described in the Advertisement for Bids designed to limit independent bidding or competition, (2) the contents of this Bid have not been communicated by the undersigned or its employees or agents to any person not an employee or agent of the undersigned or its surety on any bond furnished with the Bid and will not be communicated to such person prior to the official opening of the bids, and (3) if awarded the Contract, he will commence work within ten (10) calendar days after the date of Notice to Proceed and that he will complete the Work within the specified number of days set forth in the Agreement.

The undersigned certifies that he has received and duly considered the following Addenda to the specifications.

Addenda: No. / to No. \_\_\_\_\_ inclusive.

The undersigned agrees if awarded the Contract, that he will comply with the provisions of the Contract Documents and will comply with the provisions of ORS 279C.800 and Oregon Prevailing Wage Rates.

Name of Firm _ Eagle - Elsner Inc
Signature Richard Tayle Tres
Name Richard Eagle, Prosident
Address POBox 23294 Tigened Or 97281
Telephone No. 503 - 628-1137
Construction Contractors Board No. # 27112
If Corporation, MARY MULTIN
Secretary of Corporation

\*If bid is by a partnership, then one of the partners must sign the bid.

AnalB the Lorisonally Left Blank

E.



1 3 3 ه. ت 3 لعدنا د ت

أنف

أنفر ...



Contraction of the local division of the loc

## **BID BOND**

Any singular reference to Bidder, Surety, Owner or other party shall be considered plural where applicable.

BIDDER (Name and Address):	
Eagle Elsner, Inc.	
P. O. Box 23294	
Tigard, OR 97281 SURETY (Name, and Address of Principal Place of Business):	
Western Surety Company	
12909 SW 68th Parkway, Suite 200	
Portland, OR 97223 OWNER (Name and Address):	
City of Canby	
170 NW 2nd Ave.	
Canby, OR 97013	
BID	
Bid Due Date: 4/18/2019	
Description (Project Name— Include Location):	
2019 Street Maintenance, Various Streets, Canby, OR	
BOND	
Bond Number: <sub>N/A</sub>	
Date: 4/18/2019	
Penal sum Five Percent of the Total Amount Bid	\$ 5%
(Words)	(Figures)
Surety and Bidder, intending to be legally bound hereby, subje	ct to the terms set forth below, do each cause
this Bid Bond to be duly executed by an authorized officer, age	nt, or representative.
BIDDER SURETY	
Eagle Elsner, Inc. (Seal) Western Su	
Bidder's Name and Corporate Seal Surety's	s Name and Corporate Seal
By: Stuckard Eagle, Pier By:	moninsunning
Signature	Signature (Attach Power of Attorney)
Richard Frank	
Print Name	Gloria Bruning Print Name
*,	- Intervenice
President	Attorney-in-Fact
Tite	Title
() MABIN () MABON	
Attest: Attest: Attest:	
Signature ()	Signature
	Witness for Surety
Title SECKETARY	Title
Note: Addresses are to be used for giving any required notice.	
Provide execution by any additional parties, such as joint ventu	Irers. if necessary.
· · · · · · · · · · · · · · · · · · ·	

EJCDC <sup>®</sup> C-430, Bid Bond (Penal Sum Form). Published 2013.
Prepared by the Engineers Joint Contract Documents Committee.
Page 1 of 2

ت ت



1. Bidder and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and penal sum is the extent of Bidder the penal sum set forth on the face of this Bond. Payment of the penal sum is the extent of Bidder's and Surety's liability. Recovery of such penal sum under the terms of this Bond is the Owner's sole and exclusive remedy upon default of Bidder.

2. Default of Bidder shall occur upon the failure of Bidder to deliver within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed Agreement required by the Bidding Documents and any performance and payment bonds required by the Bidding Documents.

- 3. This obligation shall be null and void if:
- 3.1 Owner accepts Bidder's Bid and Bidder delivers within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed by the Bidding Documents, or Documents and any performance and payment bonds required by the Bidding Documents, or
- 3.2 All Bids are rejected by Owner, or
- 3.3 Owner fails to issue a Notice of Award to Bidder within the time specified in the Bidding Documents (or any extension thereof agreed to in writing by Bidder and, if applicable, consented to by Surety when required by Paragraph 5 hereof).

4. Payment under this Bond will be due and payable upon default of Bidder and within 30 calendar days after receipt by Bidder and Surety of written notice of default from Owner, which notice will be given with reasonable promptness, identifying this Bond and the Project and including a statement of the amount due.

5. Surety waives notice of any and all defenses based on or arising out of any time extension to issue Notice of Award agreed to in writing by Owner and Bidder, provided that the total time for issuing Notice of Award including extensions shall not in the aggregate exceed 120 days from the Bid due date without Surety's written consent.

6. No suit or action shall be commenced under this Bond prior to 30 calendar days after the notice of default required in Paragraph 4 above is received by Bidder and Surety and in no case later than one year after the Bid due date.

 Any suit or action under this Bond shall be commenced only in a court of competent jurisdiction located in the state in which the Project is located.

8. Notices required hereunder shall be in writing and sent to Bidder and Surety at their respective addresses shown on the face of this Bond. Such notices may be sent by personal delivery, commercial courier, or by United States Registered or Certified Mail, return receipt requested, postage pre-paid, and shall be deemed to be effective upon receipt by the party concerned.

9. Surety shall cause to be attached to this Bond a current and effective Power of Attorney evidencing the authority of the officer, agent, or representative who executed this Bond on behalf of Surety to execute, seal, and deliver such Bond and and and and the Current to execute the such the event of the such and the such the such and the such and the such a such and and and and and the such and the such an executed this Bond on behalf of Surety to execute, seal, and deliver such a such and and benet with the such a such and and behalf of Surety thereby.

10. This Bond is intended to conform to all applicable statutory requirements. Any applicable requirement of any applicable statute that has been omitted from this Bond shall be deemed to be included herein as if set forth at length. If any provision of this Bond conflicts with any applicable statute, then the provision of said statute shall govern and the remainder of this Bond that is not in conflict therewith shall continue in full force and effect.

11. The term "Bid" as used herein includes a Bid, offer, or proposal as applicable.

2 to 2 9869
Prepared by the Engineers Joint Contract Documents Committee.
EICDC® C-430, Bid Bond (Penal Sum Form). Published 2013.

# Western Surety Company

#### POWER OF ATTORNEY APPOINTING INDIVIDUAL ATTORNEY-IN-FACT

Know All Men By These Presents, That WESTERN SURETY COMPANY, a South Dakota corporation, is a duly organized and existing corporation having its principal office in the City of Sioux Falls, and State of South Dakota, and that it does by virtue of the signature and seal herein affixed hereby make, constitute and appoint

## Philip O Forker, Ray M Paiement, Vicki Mather, Gloria Bruning, Richard W Kowalski, Brent Olson, J Patrick Dooney, Karen A Pierce, Joel Dietzman, Christopher A Reburn, Gail A Price, Tami Jones, Individually

of Portland, OR, its true and lawful Attorney(s)-in-Fact with full power and authority hereby conferred to sign, seal and execute for and on its behalf bonds, undertakings and other obligatory instruments of similar nature

## - In Unlimited Amounts -

and to bind it thereby as fully and to the same extent as if such instruments were signed by a duly authorized officer of the corporation and all the acts of said Attorney, pursuant to the authority hereby given, are hereby ratified and confirmed.

This Power of Attorney is made and executed pursuant to and by authority of the By-Law printed on the reverse hereof, duly adopted, as indicated, by the shareholders of the corporation.

In Witness Whereof, WESTERN SURETY COMPANY has caused these presents to be signed by its Vice President and its corporate seal to be hereto affixed on this 29th day of November, 2018.



WESTERN SURETY COMPANY

Paul T. Bruflat, Vice President

State of South Dakota County of Minnehaha } ss

On this 29th day of November, 2018, before me personally came Paul T. Bruflat, to me known, who, being by me duly sworn, did depose and say: that he resides in the City of Sioux Falls, State of South Dakota; that he is the Vice President of WESTERN SURETY COMPANY described in and which executed the above instrument; that he knows the seal of said corporation; that the seal affixed to the said instrument is such corporate seal; that it was so affixed pursuant to authority given by the Board of Directors of said corporation and that he signed his name thereto pursuant to like authority, and acknowledges same to be the act and deed of said corporation.

My commission expires

June 23, 2021

<b> </b>
🕻 J. MOHR 🕴
SOUTH DAKOTA
SOUTH DAKOTA CERES
÷

Voh

J. Mohr, Notary Public

#### CERTIFICATE

I, L. Nelson, Assistant Secretary of WESTERN SURETY COMPANY do hereby certify that the Power of Attorney hereinabove set forth is still in force, and further certify that the By-Law of the corporation printed on the reverse hereof is still in force. In testimony whereof I have hereunto subscribed my name and affixed the seal of the said corporation this \_\_\_\_\_\_ day of \_\_\_\_\_\_,  $\frac{20/9}{20.19}$ .



WESTERN SURETY COMPANY

Form F4280-7-2012

Go to www.cnasurety.com > Owner / Obligee Services > Validate Bond Coverage, if you want to verify bond authenticity.

#### Wed-ya gnizinothuA

## VDOPTED BY THE SHAREHOLDERS OF WESTERN SURETY COMPANY

of the Company.

Section 7. All bonds, policies, undertakings, Powers of Attorney, or other obligations of the corporation shall be executed in the corporate name of the Company by the President, Secretary, and Assistant Secretary, Treasurer, or any Vice President, or by such other may appoint Attorneys in Fact or agents who shall have authority to issue bonds, policies, or undertakings in the name of the Company. The corporate seal is not necessary for the validity of any bonds, policies, undertakings, Powers of Attorney or other obligations of the Company. The corporate seal is not necessary for the validity of any bonds, policies, undertakings, Powers of Attorney or other obligations of the Company. The signature of any such officer and the corporate seal may be printed by facting.

FIRST-TIER DISCLOSURE STATEMENT

## **EIBST-TIER DISCLOSURE STATEMENT**

2

<u>لا الم</u>

1

3

ر ر

4

است با

ند له

್ರಭ

4

ARCHITER OF FIRST-TIER SUBCONTRACTOR DISCLOSURE

den datasys.

ALC: N

Sector Sector

PROJECT NAME: City of Canby

2019 Street Maintenance BID #

BID CLOSING: Date: <u>つサー18-19</u> Time: <u>2.'vo</u> アハ

This form must be submitted at the location specified in the Invitation to Bid on the advertised bid closing date and within two working hours after the advertised bid

List below the name of each subcontractor that will be furnishing labor or will be furnishing labor and materials and that is required to be disclosed, the category of work that the subcontractor will be performing and the dollar value of the subcontract. Enter "NONE" if there are no subcontractors that need to be disclosed

	CALEGORY OF WORK	Concrete									result in a non-responsive bid. A non-responsive bid will not be considered for award.
DOLLAR VALUE	S 110 900		÷ €	<del>₽</del>	æ	<del>с</del>	ଚ	6	<del>8</del>	8	dline will result in a non-responsive bid. A non-re
NAME	(1) Dd. () Concrete & 1141, tipe Tinc	(2)	(3)	(4)	(1) (5)			(1)		Failure to submit this form how the difference	

Form submitted by (bidder name):

Easle - Elsner

Eagle Vacob Contact name:

ORS 279C.370 First-tier subcontractor disclósure. (1)(a) Within two working hours after the date and time of the deadline when bids are due to a contracting agency for a public improvement contract, Will be furnishing labor or will be furnishing labor and materials in connection with the public improvement contract; and a bidder shall submit to the contracting agency a disclosure of the first-tier subcontractors that:

- ₹@
- Will have a contract value that is equal to or greater than five percent of the total project bid or \$15,000, whichever is greater, or \$350,000 regardless of the percentage of the total project bid. (b) For each contract to which this subsection applies, the contracting agency shall designate a deadline for submission of bids that has a date on a Tuesday, Wednesday or Thursday and a time between 2 p.m. and 5 p.m., except that this paragraph does not apply to public contracts for maintenance or construction of highways, bridges or other transportation facilities. This subsection applies only to public improvement contracts ("projects") with a value, estimated by the contracting agency, of more than \$100,000.
  - - ମ
  - (c) This subsection applies only to public improvement contracts ("projects") with a value, estimated by the contracting agency, of more than \$100,000.
     (d) This subsection does not apply to public improvement contracts that have been exempted from competitive bidding requirements under ORS 279C.335 (2).
     The disclosure of first-tier subcontractors under subsection (1) of this section must include the name of each subcontractor, the category of work that each subcontractor will perform and the dollar ල
- A contracting agency to be a non-responsive bid and may not award the contractor. A contracting agency is not required to determine the accuracy of the completeness of the subcontract of the subcontract to the contractor. A contracting agency is not required to determine the accuracy of the completeness of the subcontractor disclosure to the contracting agency to be a non-responsive bid and may not award the contractor. A contracting agency is not required to determine the accuracy of the completeness of the subcontractor disclosure to the contractor.
  - After the bids are opened, the subcontractor disclosures must be made available for public inspection.
  - <u>400</u>
- A contractor may substitute a first-tiler subcontractor under the provisions of ORS 279C.585. A subcontractor may file a complaint under ORS 279C.590 based on the disclosure requirements of subsection (1) of this section. WH-179 (08-10-10)

hapla iləl yilanoiinəini əzaq zidi

**CONTRACT FOR CONSTRUCTION** 

## CONTRACT FOR CONSTRUCTION

\_\_\_\_\_;

25.0

## **CONTRACT FOR CONSTRUCTION**

THIS AGREEMENT is dated as of the  $15^{+1}$  day of May in the year 2019 by and between

City of Canby

(hereinafter called OWNER) and

Eagle-Elsner, Inc.

(hereinafter called CONTRACTOR)

OWNER and CONTRACTOR, in consideration of the mutual covenants hereinafter set forth, agree as follows:

#### **ARTICLE 1 - WORK**

CONTRACTOR shall complete all Work as specified or indicated in the Contract Documents:

### City of Canby 2019 Street Maintenance

The City has an estimated available budget not to exceed \$700,000 for this work. The City reserves the right to adjust the basic bid quantities if the submitted low responsive bid exceeds the available budget. All adjusted quantities will be paid based on the bid unit prices.

The scope of work generally consists of the following:

- Approximately 8,000 tons of asphaltic concrete pavement at 2" or 3" thickness.
- Approximately 26,000 lineal feet of 6-foot wide panel grinding at 0"-2" depth and approximately 24,000 SF area at 2" depth.
- Retrofit an estimated 42 ADA ramps, including curbs, sidewalks, and 3 pedestrian midblocks.
- Restore approximately 17,000 lineal feet of 4" and 8" wide stripes, including legends such as arrows, continental crosswalks, stop bars, bike stencils, etc.
- Repair an approximate 50,000 lineal feet of joint crack sealing.

### **ARTICLE 2 - ENGINEER**

The Project has been designed by CURRAN-McLEOD, INC., Consulting Engineers, who is hereinafter called ENGINEER and who will assume all duties and responsibilities and will have the rights and authority assigned to ENGINEER in the Contract Documents in connection with completion of the Work in accordance with the Contract Documents.

## **VBTICLE 3 - CONTRACT TIME**

3.2

I.E

The Work will be substantially completed within 45 calendar days after the date when the Contract Time commences to run as provided in paragraph 4.01 of the General Conditions, and completed and ready for final payment in accordance with Paragraph 15.06 of the General Conditions within 15 days after the date when the issuance of the Certificate of Substantial Completion including punch list items.

Liquidated Damages: OWNER and CONTRACTOR recognize that time is of the essence of this Agreement and that OWNER will suffer financial loss if the Work is not substantially complete within the time specified in paragraph 3.1 above, plus any extensions thereof allowed in accordance with Article 12 of the General Conditions. They also recognize the delays, expense and difficulties involved in proving in a legal proceeding the actual loss suffered by OWNER if the Work is not substantially complete on time.

Accordingly, instead of requiring any such proof, OWNER and CONTRACTOR agree that as liquidated damages for delay (but not as a penalty) CONTRACTOR Four Hundred Dollars or the OWNER may withhold from amounts due the CONTRACTOR Four Hundred Dollars (\$400.00) for each day that expires after the time specified in paragraph 3.1. for Substantial Completion until the Work is substantially complete AND/OR for each day of delay beyond the deadline for Final Completion.

## **VBLICLE 4 - CONTRACT PRICE**

4.1 OWNER shall pay CONTRACTOR for performance of the Work in accordance with the Contract Documents in current funds by check, an amount totaling

3

لو ب

لحرب

One Million Forty-Two Thousand Six Hundred Fifty-Seven and 50/100 Dollars

(<u>\$1,042,657.50</u>) as shown in the attached Bid Proposal.

## **VBLICTE 2 - BVAMENT PROCEDURES**

CONTRACTOR shall submit Applications for Payment in accordance with Article 15 of the General Conditions. Applications for Payment will be processed by ENGINEER as provided in the General Conditions.

5.1 Progress Payments: OWNER shall make progress payments on account of the Contract Price on the basis of CONTRACTOR'S Applications for Payment as recommended by ENGINEER, on or about the 25th day of each month during progress of the Work measured by the schedule of values provided for in paragraph 2.03 of the General Conditions.

- 5.1.1 Prior to Substantial Completion progress payments will be in an amount equal to:
  - (a) 95 % of the Work completed; and
  - (b) 95 % of materials and equipment not incorporated in the Work but delivered and suitably stored, less in each case the aggregate of payments previously made.
- 5.1.2 Upon Substantial Completion, OWNER shall pay an amount sufficient to increase total payments to CONTRACTOR to 95% of the value of the Contract Work completed, less such amounts as ENGINEER shall determine in accordance with paragraph 15.01 of the General Conditions.
- 5.2 Final Payment: Upon final completion and acceptance of the Work in accordance with paragraph 15.06 of the General Conditions, OWNER shall pay the remainder of the value of the Contract Work completed, as recommended by ENGINEER as provided in said paragraph 15.06.

### **ARTICLE 6 - INTEREST**

All monies not paid when due hereunder shall bear interest at the maximum rate allowed by law at the place of the Project, when requested in accordance with ORS 279C.570

### **ARTICLE 7 - CONTRACTOR'S REPRESENTATIONS**

In order to induce OWNER to enter into this Agreement CONTRACTOR makes the following representations:

- 7.1 CONTRACTOR has familiarized himself with the nature and extent of the Contract Documents, Work, locality, and with all local conditions and federal, state and local laws, ordinances, rules and regulations that in any manner may affect cost, progress or performance of the Work.
- 7.2 CONTRACTOR has studied carefully all reports of investigations and tests of subsurface and latent physical conditions at the site or otherwise affecting cost, progress or performance of the Work which were relied upon by ENGINEER in the preparation of the Drawings and Specifications and which have been identified in the Supplementary Conditions.
- 7.3 CONTRACTOR has made or caused to be made examinations, investigations and tests and studies of such reports and related data in addition to those referred to in paragraph 7.2 as he deems necessary for the performance of the Work at the Contract Price, within the Contract Time and in accordance with the other terms and conditions of the Contract Documents; and no additional examinations, investigations, tests, reports or similar data are or will be required by CONTRACTOR for such purposes.

- 7.4 CONTRACTOR has correlated the results of all such observations, examinations, investigations, tests, reports and data with the terms and conditions of the Contract Documents.
- 7.5 CONTRACTOR has given ENGINEER written notice of all conflicts, errors or discrepancies that he has discovered in the Contract Documents and the written resolution thereof by ENGINEER is acceptable to CONTRACTOR.

#### **VBLICLE 8 - CONTRACT DOCUMENTS**

- 1.8 Agreement
- .2 Exhibits to this Agreement.
- 8.3 Performance and other Bonds
- .4. Notice of Award.
- 8.5 General Conditions of the Construction Contract
- 8.6 Supplementary Conditions
- 8.7 Technical Specifications as listed in the Table of Contents.
- 8.8 Drawings & Specifications bearing the following general title:
   2019 Street Maintenance
- 8.9 Addenda numbers 1-3.
- 8.10 CONTRACTOR'S Bid
- Any Modification, including Change Orders, duly delivered after execution of Agreement.

There are no Contract Documents other than those listed above in this ARTICLE 8. The Contract Documents may only be altered, amended or repealed by a Modification (as defined in Article 1 of the General Conditions).

#### **VBLICTE 9 - WISCELLANEOUS**

9.1 Terms used in this Agreement which are defined in Article 1 of the General Conditions.

- 9.2 No assignment by a party hereto of any rights under or interests in the Contract Documents will be binding on another party hereto without the written consent of the party sought to be bound; and specifically by without limitation, moneys that may become due and moneys that are due may not be assigned without such consent (except to the extent that the effect of this restriction may be limited by law), and unless specifically stated to the contrary in any written consent to an assignment no assignment will release or discharge the assignor from any duty or responsibility under the Contract Documents.
- 9.3 OWNER and CONTRACTOR each binds himself, his partners, successors, assigns and legal representatives to the other party hereto, his partners, successors, assigns and legal representatives in respect to all covenants, agreements and obligations contained in the Contract Documents.
- 9.4 In the event a suit, arbitration or other legal action is required by either the OWNER or the CONTRACTOR to enforce any provisions of this Agreement, the prevailing parties shall be entitled to all reasonable costs and reasonable attorney's fees upon trial or subsequent appeal.

IN WITNESS WHEREOF, the parties hereto have signed three counterparts of this Agreement.

This Agreement will be effective on Mau , 2019. **OWNER: CONTRACTOR: City of Canby Eagle-Elsner**, Inc. P.O. Box 930 17400 SW Hillsboro hwy Canby, OR 97013 Sherwood, OR 97140 Name/Title: Brian Hod Name/Title: Nichar Name/Title:\_

Address for giving notices:

AnolB the Language Interview  $L_{gamma}$  and  $L_{gamma}$ 



## CONSTRUCTION PERFORMANCE BOND

 $\gamma$ 

j\$

ل ي

5. Lá

لأذار

<b>EJCDC</b>
ENGINEERS JOINT CONTRACT
DOCUMENTS COMMITTEE

## PERFORMANCE BOND

CONTRA	CTOR	Iname	and	address	):

Eagle Elsner, Inc. PO Box 23294 Tigard, OR 97281 SURETY (name and address of principal place of business):

Western Surety Company 12909 SW 68th Parkway, Suite 200 Portland, OR 97223

OWNER (name and address):
City of Canby
222 NE 2nd Ave.
Canby, OR 97013
CONSTRUCTION CONTRACT
Effective Date of the Agreement: May $15$ , 2019
Amount: One Million Forty Two Thousand Six Hundred Fifty Seven & 50/100(\$1,042,657.50) Dollars
Description (name and location): City of Canby
2019 Street Maintenance
BOND
Bond Number: 30062437
Date (not earlier than the Effective Date of the Agreement of the Construction Contract): May 15, 2019
Amount: One Million Forty Two Thousand Six Hundred Fifty Seven & 50/100(\$1,042,657.50) Dollars
Modifications to this Bond Form: 🔀 None 🗌 See Paragraph 16

Surety and Contractor, intending to be legally bound hereby, subject to the terms set forth below, do each cause this Performance Bond to be duly executed by an authorized officer, agent, or representative.

#### CONTRACTOR AS PRINCIPAL

#### SURETY

Eagle Elsner, Inc.	(seal)	Western Surety Company	(seal)
Contractor's Name and Corporate Seal	_ ( )	Surety's Name and Corporate Seal	
By: Fichand Eagle, Tras		By:	
Richard Eagle		Gloria Bruning	
Print Name /		Print Name	
President		Attorney-in-fact	
Title Attest: Mary Merron	)	Title Attest:	
Signature		Signature	
GECKËTARY		Agent-Witness for Surety	
Title		Title	

Notes: (1) Provide supplemental execution by any additional parties, such as joint venturers. (2) Any singular reference to Contractor, Surety, Owner, or other party shall be considered plural where applicable.

EJCDC <sup>®</sup> C-610, Performance Bond	
Copyright © 2013 National Society of Professional Engineers, American Council of	Engineering Companies,
and American Society of Civil Engineers. All rights reserved.	1 of 3

1. The Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to the Owner for the performance of the Construction Contract, which is incorporated herein by reference.

2. If the Contractor performs the Construction Contract, the Surety and the Contractor shall have no obligation under this Bond, except when applicable to participate in a conference as provided in Paragraph 3.

3. If there is no Owner Default under the Construction Contract, the Surety's obligation under this Bond shall arise after:

3.1 The Owner first provides notice to the Contractor and the Surety that the Owner is considering declaring a Contractor Default. Such notice shall indicate whether the Owner is requesting a conference among the Owner, Contractor, and Surety to discuss the Contractor's performance. If the Owner does not request a conference, the Surety may, within five (5) business days after receipt of the Owner's notice, request such a conference. If the Surety timely requests a conference, the Owner shall attend. Unless the Owner agrees otherwise, any conference requested under this Paragraph 3.1 shall be held within ten (10) business days of the Surety's receipt of the Owner's notice. If the Owner, the Contractor, and the Surety agree, the Contractor shall be allowed a reasonable time to perform the Construction Contract, but such an agreement shall not waive the Owner's right, if any, subsequently to declare a Contractor Default:

3.2 The Owner declares a Contractor Default, terminates the Construction Contract and notifies the Surety; and

3.3 The Owner has agreed to pay the Balance of the Contract Price in accordance with the terms of the Construction Contract to the Surety or to a contractor selected to perform the Construction Contract.

4. Failure on the part of the Owner to comply with the notice requirement in Paragraph 3.1 shall not constitute a failure to comply with a condition precedent to the Surety's obligations, or release the Surety from its obligations, except to the extent the Surety demonstrates actual prejudice.

5. When the Owner has satisfied the conditions of Paragraph 3, the Surety shall promptly and at the Surety's expense take one of the following actions:

5.1 Arrange for the Contractor, with the consent of the Owner, to perform and complete the Construction Contract;

5.2 Undertake to perform and complete the Construction Contract itself, through its agents or independent contractors;

5.3 Obtain bids or negotiated proposals from qualified contractors acceptable to the Owner for a contract for performance and completion of the Construction Contract, arrange for a contract to be prepared for execution by the

Owner and a contractor selected with the Owners concurrence, to be secured with performance and payment bonds executed by a qualified surety equivalent to the bonds issued on the Construction Contract, and pay to the Owner the amount of damages as described in Paragraph 7 in excess of the Balance of the Contract Price incurred by the Owner as a result of the Contractor Default; or

5.4 Waive its right to perform and complete, arrange for completion, or obtain a new contractor, and with reasonable promptness under the circumstances:

5.4.1 After investigation, determine the amount for which it may be liable to the Owner and, as soon as practicable after the amount is determined, make payment to the Owner; or

5.4.2 Deny liability in whole or in part and notify the Owner, citing the reasons for denial.

6. If the Surety does not proceed as provided in Paragraph 5 with reasonable promptness, the Surety shall be deemed to be in default on this Bond seven days after receipt of an additional written notice from the Owner to the Surety demanding that the Surety perform its obligations under this Bond, and the Owner shall be entitled to enforce any remedy available to the Owner. If the Surety proceeds as provided in Paragraph 5.4, and the Owner refuses the payment or the Surety has denied liability, in whole or in part, without further notice the Owner shall be entitled to enforce any remedy available to the Owner and the Owner shall be entitled to enforce any remedy available to the Owner refuses the payment or the Surety has denied liability.

7. If the Surety elects to act under Paragraph 5.1, 5.2, or 5.3, then the responsibilities of the Surety to the Owner shall not be greater than those of the Contractor under the Construction Contract, and the responsibilities of the Owner to the Surety shall not be greater than those of the Owner under the Construction Contract. Subject to the commitment by the Owner to pay the Balance of the Contract Price, the Surety is obligated, without duplication for:

7.1 the responsibilities of the Contractor for correction of defective work and completion of the Construction Contract;

7.2 additional legal, design professional, and delay costs resulting from the Contractor's Default, and resulting from the actions or failure to act of the Surety under Paragraph 5; and

7.3 liquidated damages, or if no liquidated damages are specified in the Construction Contract, actual damages caused by delayed performance or non-performance of the Contractor.

8. If the Surety elects to act under Paragraph 5.1, 5.3, or 5.4, the Surety's liability is limited to the amount of this Bond.

9. The Surety shall not be liable to the Owner or others for obligations of the Contractor that are unrelated to the Construction Contract, and the Balance of the Contract Price shall not be reduced or set off on account of any such unrelated obligations. No right of action shall accrue on this Bond to any person or entity other than

EJCDC® C-610, Performance Bond Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. 2 of 3 ĺ , 

ĺ



**CONSTRUCTION PAYMENT BOND** 

## CONSTRUCTION PAYMENT BOND



## **PAYMENT BOND**

CONTRACTOR (name and address):		SURETY (name and address of principal place of business):	
Eagle Elsner, Inc. PO Box 23294 Tigard, OR 97281		Western Surety Company 12909 SW 68th Parkway, Suite 200 Portland, OR 97223	
OWNER (name and address): City of Canby, 222 NE 2nd Ave., Canby	v, OR 97013		
CONSTRUCTION CONTRACT			
Effective Date of the Agreemen Amount: One Million Forty Two Tho		& 50/100(\$1,042,657.50) Dollars	
Description (name and location):	City of Canby 2019 Street Maintenance		
BOND			
Bond Number: 30062437			
Date (not earlier than the Effective De	ate of the Agreement of the Co	nstruction Contract): May 15 , 2019	
Amount: One Million Forty Two Thousand Six Hundred Fifty Seven & 50/100(\$1,042,657.50) Dollars			
Modifications to this Bond Forn	m: 🔯 None 🛄 See	e Paragraph 18	

Surety and Contractor, intending to be legally bound hereby, subject to the terms set forth below, do each cause this Payment Bond to be duly executed by an authorized officer, agent, or representative.

#### CONTRACTOR AS PRINCIPAL

#### SURETY

Eagle Elsner, Inc. (se	eal) Western Surety Company	(seal)
Contractor's Name and Corporate Seal	Surety's Name and Corporate Seal	
By: Stichard Eagh, Firs Signature	By:	
Richard Eagle	Gloria Bruning	
Print Name	Print Name	
President	Attorney-in-fact	
Title Marken	Title	
Attest:	Attest:	
Signature	Signature	
SECKETARY	Agent-Witness for Surety	
Title	Title	

Notes: (1) Provide supplemental execution by any additional parties, such as joint venturers. (2) Any singular reference to Contractor, Surety, Owner, or other party shall be considered plural where applicable.

- 1. The Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to the Owner to pay for labor, materials, and equipment furnished for use in the performance of the Construction Contract, which is incorporated herein by reference, subject to the following terms.
- 2. If the Contractor promptly makes payment of all sums due to Claimants, and defends, indemnifies, and holds harmless the Owner from claims, demands, liens, or suits by any person or entity seeking payment for labor, materials, or equipment furnished for use in the performance of the Construction Contract, then the Surety and the Contractor shall have no obligation under this Bond.
- 3. If there is no Owner Default under the Construction Contract, the Surety's obligation to the Owner under this Bond shall arise after the Owner has promptly notified the Contractor and the Surety (at the address described in Paragraph 13) of claims, demands, liens, or suits against the Owner or the Owner's property by any person or entity seeking payment for labor, materials, or equipment furnished for use in the performance of the Construction Contract, and tendered defense of such claims, demands, liens, or suits to the Contractor and the Surety.
- 4. When the Owner has satisfied the conditions in Paragraph 3, the Surety shall promptly and at the Surety's expense defend, indemnify, and hold harmless the Owner against a duly tendered claim, demand, lien, or suit.
- 5. The Surety's obligations to a Claimant under this Bond shall arise after the following:
  - 5.1 Claimants who do not have a direct contract with the Contractor,
    - 5.1.1 have furnished a written notice of nonpayment to the Contractor, stating with substantial accuracy the amount claimed and the name of the party to whom the materials were, or equipment was, furnished or supplied or for whom the labor was done or performed, within ninety (90) days after having last performed labor or last furnished materials or equipment included in the Claim; and
    - 5.1.2 have sent a Claim to the Surety (at the address described in Paragraph 13).
  - 5.2 Claimants who are employed by or have a direct contract with the Contractor have sent a Claim to the Surety (at the address described in Paragraph 13).

- 6. If a notice of non-payment required by Paragraph 5.1.1 is given by the Owner to the Contractor, that is sufficient to satisfy a Claimant's obligation to furnish a written notice of non-payment under Paragraph 5.1.1.
- 7. When a Claimant has satisfied the conditions of Paragraph 5.1 or 5.2, whichever is applicable, the Surety shall promptly and at the Surety's expense take the following actions:
  - 7.1 Send an answer to the Claimant, with a copy to the Owner, within sixty (60) days after receipt of the Claim, stating the amounts that are undisputed and the basis for challenging any amounts that are disputed; and
  - 7.2 Pay or arrange for payment of any undisputed amounts.
  - 7.3 The Surety's failure to discharge its obligations under Paragraph 7.1 or 7.2 shall not be deemed to constitute a waiver of defenses the Surety or Contractor may have or acquire as to a Claim, except as to undisputed amounts for which the Surety and Claimant have reached agreement. If, however, the Surety fails to discharge its obligations under Paragraph 7.1 or 7.2, the Surety shall indemnify the Claimant for the reasonable attorney's fees the Claimant incurs thereafter to recover any sums found to be due and owing to the Claimant.
- 8. The Surety's total obligation shall not exceed the amount of this Bond, plus the amount of reasonable attorney's fees provided under Paragraph 7.3, and the amount of this Bond shall be credited for any payments made in good faith by the Surety.
- 9. Amounts owed by the Owner to the Contractor under the Construction Contract shall be used for the performance of the Construction Contract and to satisfy claims, if any, under any construction performance bond. By the Contractor furnishing and the Owner accepting this Bond, they agree that all funds earned by the Contractor in the performance of the Construction Contract are dedicated to satisfy obligations of the Contractor and Surety under this Bond, subject to the Owner's priority to use the funds for the completion of the work.
- 10. The Surety shall not be liable to the Owner, Claimants, or others for obligations of the Contractor that are unrelated to the Construction Contract. The Owner shall not be liable for the payment of any costs or expenses of any Claimant under this Bond, and shall have under this Bond no obligation to make payments to or give notice on behalf of Claimants, or otherwise have any obligations to Claimants under this Bond.
- 11. The Surety hereby waives notice of any change, including changes of time, to the Construction Contract or to related subcontracts, purchase orders, and other obligations.

EJCDC<sup>®</sup> C-615, Payment Bond Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. 2 of 3

- 12. No suit or action shall be commenced by a Claimant under this Bond other than in a court of competent jurisdiction in the state in which the project that is the subject of the Construction Contract is located or after the expiration of one year from the date (1) on which the Claimant sent a Claim to the Surety pursuant to Paragraph 5.1.2 or 5.2, or (2) on which the last labor or service was performed by anyone or the last materials or equipment were furnished by anyone under the Construction Contract, whichever of (1) or (2) first occurs. If the provisions of this paragraph are void or prohibited by law, the minimum period of limitation available to sureties as a defense in the jurisdiction of the suit shall be applicable.
- 13. Notice and Claims to the Surety, the Owner, or the Contractor shall be mailed or delivered to the address shown on the page on which their signature appears. Actual receipt of notice or Claims, however accomplished, shall be sufficient compliance as of the date received.
- 14. When this Bond has been furnished to comply with a statutory or other legal requirement in the location where the construction was to be performed, any provision in this Bond conflicting with said statutory or legal requirement shall be deemed deleted herefrom and provisions conforming to such statutory or other legal requirement shall be deemed incorporated herein. When so furnished, the intent is that this Bond shall be construed as a statutory bond and not as a common law bond.
- 15. Upon requests by any person or entity appearing to be a potential beneficiary of this Bond, the Contractor and Owner shall promptly furnish a copy of this Bond or shall permit a copy to be made.

#### 16. Definitions

- 16.1 **Claim:** A written statement by the Claimant including at a minimum:
  - 1. The name of the Claimant;
  - The name of the person for whom the labor was done, or materials or equipment furnished;
  - 3. A copy of the agreement or purchase order pursuant to which labor, materials, or equipment was furnished for use in the performance of the Construction Contract;
  - 4. A brief description of the labor, materials, or equipment furnished;
  - The date on which the Claimant last performed labor or last furnished materials or equipment for use in the performance of the Construction Contract;
  - The total amount earned by the Claimant for labor, materials, or equipment furnished as of the date of the Claim;
  - 7. The total amount of previous payments received by the Claimant; and

- 8. The total amount due and unpaid to the Claimant for labor, materials, or equipment furnished as of the date of the Claim.
- 16.2 Claimant: An individual or entity having a direct contract with the Contractor or with a subcontractor of the Contractor to furnish labor, materials, or equipment for use in the performance of the Construction Contract. The term Claimant also includes any individual or entity that has rightfully asserted a claim under an applicable mechanic's lien or similar statute against the real property upon which the Project is located. The intent of this Bond shall be to include without limitation in the terms of "labor, materials, or equipment" that part of the water, gas, power, light, heat, oil, gasoline, telephone service, or rental equipment used in the Construction Contract, architectural and engineering services required for performance of the work of the Contractor and the Contractor's subcontractors, and all other items for which a mechanic's lien may be asserted in the jurisdiction where the labor, materials, or equipment were furnished.
- 16.3 **Construction Contract:** The agreement between the Owner and Contractor identified on the cover page, including all Contract Documents and all changes made to the agreement and the Contract Documents.
- 16.4 **Owner Default**: Failure of the Owner, which has not been remedied or waived, to pay the Contractor as required under the Construction Contract or to perform and complete or comply with the other material terms of the Construction Contract.
- 16.5 **Contract Documents:** All the documents that comprise the agreement between the Owner and Contractor.
- 17. If this Bond is issued for an agreement between a contractor and subcontractor, the term Contractor in this Bond shall be deemed to be Subcontractor and the term Owner shall be deemed to be Contractor.
- 18. Modifications to this Bond are as follows: NONE

Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. 3 of 3

# Western Surety Company

### POWER OF ATTORNEY APPOINTING INDIVIDUAL ATTORNEY-IN-FACT

Know All Men By These Presents, That WESTERN SURETY COMPANY, a South Dakota corporation, is a duly organized and existing corporation having its principal office in the City of Sioux Falls, and State of South Dakota, and that it does by virtue of the signature and seal herein affixed hereby make, constitute and appoint

## Philip O Forker, Ray M Paiement, Vicki Mather, Gloria Bruning, Richard W Kowalski, Brent Olson, J Patrick Dooney, Karen A Pierce, Joel Dietzman, Christopher A Reburn, Gail A Price, Tami Jones, Individually

of Portland, OR, its true and lawful Attorney(s)-in-Fact with full power and authority hereby conferred to sign, seal and execute for and on its behalf bonds, undertakings and other obligatory instruments of similar nature

#### - In Unlimited Amounts -

and to bind it thereby as fully and to the same extent as if such instruments were signed by a duly authorized officer of the corporation and all the acts of said Attorney, pursuant to the authority hereby given, are hereby ratified and confirmed.

This Power of Attorney is made and executed pursuant to and by authority of the By-Law printed on the reverse hereof, duly adopted, as indicated, by the shareholders of the corporation.

In Witness Whereof, WESTERN SURETY COMPANY has caused these presents to be signed by its Vice President and its corporate seal to be hereto affixed on this 29th day of November, 2018.

State of South Dakota County of Minnehaha

} 'ss

On this 29th day of November, 2018, before me personally came Paul T. Bruflat, to me known, who, being by me duly sworn, did depose and say: that he resides in the City of Sioux Falls, State of South Dakota; that he is the Vice President of WESTERN SURETY COMPANY described in and which executed the above instrument; that he knows the seal of said corporation; that the seal affixed to the said instrument is such corporate seal; that it was so affixed pursuant to authority given by the Board of Directors of said corporation and that he signed his name thereto pursuant to like authority, and acknowledges same to be the act and deed of said corporation.

My commission expires

June 23, 2021

J. MOHR
SEAL NOTARY PUBLIC SEAL SOUTH DAKOTA
*************************************

lon.

WESTERN SURETY COMPANY

#### CERTIFICATE

J. Mohr, Notary Public

Bruflat, Vice President

I, L. Nelson, Assistant Secretary of WESTERN SURETY COMPANY do hereby certify that the Power of Attorney hereinabove set forth is still in force, and further certify that the By-Law of the corporation printed on the reverse hereof is still in force. In testimony whereof I have hereunto subscribed my name and affixed the seal of the said corporation this \_\_\_\_\_\_ day of \_\_\_\_\_\_ May of \_\_\_\_\_\_\_.

SEAL ANDROT

WESTERN SURETY COMPANY

Relson Nelson, Assistant Secretary

Form F4280-7-2012

Go to www.cnasurety.com > Owner / Obligee Services > Validate Bond Coverage, if you want to verify bond authenticity.



#### wel-ya gnizirohtuA

## VDOPTED BY THE SHAREHOLDERS OF WESTERN SURETY COMPANY

This Power of Attorney is made and executed pursuant to and by authority of the following By-Law duly adopted by the shareholders of the Company.

Section 7. All bonds, policies, undertakings, Powers of Attorney, or other obligations of the corporation shall be executed in the corporate name of the Company by the President, Secretary, and Assistant Secretary, Treasurer, or any Vice President, or by such other may appoint Attorneys in Fact or agents who shall have authority to issue bonds, policies, or undertakings, Powers of Attorney or the Company. The corporate scal is not necessary for the validity of any bonds, policies, undertakings, Powers of Attorney or other obligations of the Company. The signature of any such officer and the corporate scal may be printed by factings.


# EAGLELS-01

DATE (MM/DD/YYYY)

								5	/2/2019
THIS CERTIFICATE IS ISSUED AS A CERTIFICATE DOES NOT AFFIRMAT BELOW. THIS CERTIFICATE OF IN REPRESENTATIVE OR PRODUCER, A	SUR ND 1		R NEGATIVELY AMEND E DOES NOT CONSTITU ERTIFICATE HOLDER.	), EXTE JTE A	ND OR ALI CONTRACT	FER THE CO BETWEEN	OVERAGE AFFORDED THE ISSUING INSUREF	BY TH R(S), AL	IE POLICIES JTHORIZED
IMPORTANT: If the certificate holde If SUBROGATION IS WAIVED, subje this certificate does not confer rights	ct to	o the	terms and conditions of	f the ne	licy cortain	nolicios mov	NAL INSURED provisio	nsorb nt.As	e endorsed. tatement on
PRODUCER				CONTA NAME:	CT Kimberly	/ Lee			
Anchor Insurance & Surety, Inc. 1201 SW 12th Ave. Suite 500				PHONE (A/C, No	, Ext): (503) 2	224-2500	FAX (A/C, No)	(503)	224-9830
Portland, OR 97205				E-MAIL	<sub>ss:</sub> klee@an	chorias.co	m		
					INS	SURER(S) AFFO	RDING COVERAGE		NAIC #
				INSURE	RA: Charter	Oak Fire I	ns. Co.		25615
INSURED				INSURE	<u>к в :</u> T <b>ravele</b>	rs Property	/ Casualty Co. of Am	erica	25674
Eagle Elsner, Inc.				INSURE	R c : T <b>ravele</b>	rs Casualt	y Insurance Co. of Ar	nerica	
P. O. Box 23294 Tigard, OR 97281				INSURE	RD:SAIF C	orporation			36196
				INSURE	RE:				
				INSURE	RF:				
			E NUMBER:				<b>REVISION NUMBER:</b>		
THIS IS TO CERTIFY THAT THE POLICI INDICATED. NOTWITHSTANDING ANY F CERTIFICATE MAY BE ISSUED OR MAY EXCLUSIONS AND CONDITIONS OF SUCH	PEF POLI	IREMI RTAIN, ICIES.	ENT, TERM OR CONDITIO THE INSURANCE AFFOR LIMITS SHOWN MAY HAVE	N OF A	NY CONTRA THE POLIC EDUCED BY	CT OR OTHEF IES DESCRIE PAID CLAIMS	R DOCUMENT WITH RESP BED HEREIN IS SUBJECT	FOT TO	MALIOU TURO
INSR TYPE OF INSURANCE		SUBR	POLICY NUMBER		POLICY EFF (MM/DD/YYYY)	POLICY EXP (MM/DD/YYYY)	LIM	TS	
A X COMMERCIAL GENERAL LIABILITY							EACH OCCURRENCE	\$	1,000,000
	X		CO-1019R236-COF-18		6/1/2018	6/1/2019	DAMAGE TO RENTED PREMISES (Ea occurrence)	\$	300,000
X XC&U COVERAGE							MED EXP (Any one person)	\$	10,000
X CONTRACTUAL LIAB							PERSONAL & ADV INJURY	\$	1,000,000
GEN'L AGGREGATE LIMIT APPLIES PER:							GENERAL AGGREGATE	\$	2,000,000
POLICY X PRO- JECT LOC							PRODUCTS - COMP/OP AGG	\$	2,000,000
OTHER: WA STOP GAP							POLLUTION	\$	1,000,000
B AUTOMOBILE LIABILITY							COMBINED SINGLE LIMIT (Ea accident)	\$	1,000,000
	X		810-1019R236-TIL-18	6/1/20	6/1/2018	6/1/2019	BODILY INJURY (Per person)	\$	
OWNED AUTOS ONLY SCHEDULED AUTOS	c						BODILY INJURY (Per accident)	\$	
HIRED AUTOS ONLY AUTOS ONLY					PROPERTY DAMAGE (Per accident)	\$			
	ļ						AUTO POLLUTION	\$	1,000,000
C X UMBRELLA LIAB X OCCUR						6/1/2010	EACH OCCURRENCE	\$	1,000,000
EXCESS LIAB CLAIMS-MADE	1		CUP-5J064957-18-26		6/1/2018	6/1/2019	AGGREGATE	\$	1,000,000
DED X RETENTION \$ 10,000								\$	
AND EMPLOYERS' LIABILITY			810540	10/1/2018	10/1/2019	PER OTH- STATUTE ER			
ANY PROPRIETOR/PARTNER/EXECUTIVE OFFICER/MEMBER EXCLUDED? (Mandatory in NH)	N / A		010040			E.L. EACH ACCIDENT	\$	500,000	
If yes, describe under						E.L. DISEASE - EA EMPLOYEI	\$	500,000	
DESCRIPTION OF OPERATIONS below	DESCRIPTION OF OPERATIONS below					· · ·	E.L. DISEASE - POLICY LIMIT	\$	500,000
DESCRIPTION OF OPERATIONS / LOCATIONS / VEHIC	FS U	ACORD	101 Additional Remarks Schedu	le may be					
Certificate holder and all other entities are a waiver of subrogation when required by wri 46 08 05, CG D3 16 11 11, CG D2 11 01 04, C Umbrella Excess Liability goes over Genera	dditi tten A T3	onal i contra 53 02	insureds when specified b act. All subject to the term 2 15. WC000313.	y writte Is, cond	n contract. C	overane is n	rimany & non-contributor	y and in its attac	ncludes hed: CG D2
CERTIFICATE HOLDER					ELLATION				
City of Canby 222 NE 2nd Avenue Canby, OR 97013				ACC	EXPIRATION DRDANCE WI	N DATE TH	ESCRIBED POLICIES BE C IEREOF, NOTICE WILL Y PROVISIONS.	ANCELI BE DE	_ED BEFORE LIVERED IN
				AUTHOR	joel of	0			

ACORD	25 (	(201	6/0	(3)

© 1988-2015 ACORD CORPORATION. All rights reserved.

• • j j

### THIS ENDORSEMENT CHANGES THE POLICY. PLEASE READ IT CAREFULLY

## BLANKET ADDITIONAL INSURED (CONTRACTORS)

This endorsement modifies insurance provided under the following: COMMERCIAL GENERAL LIABILITY COVERAGE PART

- WHO IS AN INSURED (Section II) is amended to include any person or organization that you agree in a "written contract requiring insurance" to include as an additional insured on this Coverage Part, but:
  - a) Only with respect to liability for "bodily injury", "property damage" or "personal injury"; and
  - b) If, and only to the extent that, the injury or damage is caused by acts or omissions of you or your subcontractor in the performance of "your work" to which the "written contract requiring insurance" applies. The person or organization does not qualify as an additional insured with respect to the independent acts or omissions of such person or organization.
- 2. The insurance provided to the additional insured by this endorsement is limited as follows:
  - a) In the event that the Limits of Insurance of this Coverage Part shown in the Declarations exceed the limits of liability required by the "written contract requiring insurance", the insurance provided to the additional insured shall be limited to the limits of liability required by that "written contract requiring insurance". This endorsement shall not increase the limits of insurance described in Section III – Limits Of Insurance.
  - b) The insurance provided to the additional insured does not apply to "bodily injury", "property damage" or "personal injury" arising out of the rendering of, or failure to render, any professional architectural, engineering or surveying services, including:
    - i. The preparing, approving, or failing to prepare or approve, maps, shop drawings, opinions, reports, surveys, field orders or change orders, or the preparing, approving, or failing to prepare or approve, drawings and specifications; and
    - ii. Supervisory, inspection, architectural or engineering activities.

- c) The insurance provided to the additional insured does not apply to "bodily injury" or "property damage" caused by "your work" and included in the "products-completed operations hazard" unless the "written contract requiring insurance" specifically requires you to provide such coverage for that additional insured, and then the insurance provided to the additional insured applies only to such "bodily injury" or "property damage" that occurs before the end of the period of time for which the "written contract requiring insurance" requires you to provide such coverage or the end of the policy period, whichever is earlier.
- 3. The insurance provided to the additional insured by this endorsement is excess over any valid and collectible "other insurance", whether primary, excess, contingent or on any other basis, that is available to the additional insured for a loss we cover under this endorsement. However, if the "written contract requiring insurance" specifically requires that this insurance apply on a primary basis or a primary and non-contributory basis. this insurance is primary to "other insurance" available to the additional insured which covers that person or organization as a named insured for such loss, and we will not share with that "other insurance". But the insurance provided to the additional insured by this endorsement still is excess over any valid and collectible "other insurance", whether primary, excess, contingent or on any other basis, that is available to the additional insured when that person or organization is an additional insured under such "other insurance".
- 4. As a condition of coverage provided to the additional insured by this endorsement:
  - a) The additional insured must give us written notice as soon as practicable of an "occurrence" or an offense which may result in a claim. To the extent possible, such notice should include:

### COMMERCIAL GENERAL LIABILITY

- i. How, when and where the "occurrence" or offense took place;
- II. The names and addresses of any injured persons and witnesses; and
- iii. The nature and location of any injury or damage arising out of the "occurrence" or offense.
- b) If a claim is made or "suit" is brought against the additional insured, the additional insured must:
  - Immediately record the specifics of the claim or "suit" and the date received; and
  - II. Notify us as soon as practicable.

The additional insured must see to it that we receive written notice of the claim or "suit" as soon as practicable.

- c) The additional insured must immediately send us copies of all legal papers received in connection with the claim or "suit", cooperate with us in the investigation or settlement of the claim or defense against the "suit", and otherwise comply with all policy conditions.
- d) The additional insured must tender the defense and indemnity of any claim or "suit" to

any provider of "other insurance" which would cover the additional insured for a loss we cover under this endorsement. However, this condition does not affect whether the insurance provided to the additional insured by this endorsement is primary to "other insurance" available to the additional insured which covers that person or organization as a named insured as described in paragraph **3**. above.

 The following definition is added to SECTION V. – DEFINITIONS:

> "Written contract requiring insurance" means that part of any written contract or agreement under which you are required to include a person or organization as an additional insured on this Coverage Part, provided that the "bodily injury" and "property damage" occurs and the "personal injury" is caused by an offense committed:

- After the signing and execution of the contract or agreement by you;
- **b.** While that part of the contract or agreement is in effect; and
- c. Before the end of the policy period.

### THIS ENDORSEMENT CHANGES THE POLICY. PLEASE READ IT CAREFULLY.

## CONTRACTORS XTEND ENDORSEMENT

This endorsement modifies insurance provided under the following:

COMMERCIAL GENERAL LIABILITY COVERAGE PART

**GENERAL DESCRIPTION OF COVERAGE** – This endorsement broadens coverage. However, coverage for any injury, damage or medical expenses described in any of the provisions of this endorsement may be excluded or limited by another endorsement to this Coverage Part, and these coverage broadening provisions do not apply to the extent that coverage is excluded or limited by such an endorsement. The following listing is a general coverage description only. Limitations and exclusions may apply to these coverages. Read all the provisions of this endorsement and the rest of your policy carefully to determine rights, duties, and what is and is not covered.

- A. Aircraft Chartered With Pilot
- B. Damage To Premises Rented To You
- C. Increased Supplementary Payments
- D. Incidental Medical Malpractice
- E. Who Is An Insured Newly Acquired Or Formed Organizations
- F. Who Is An Insured Broadened Named Insured Unnamed Subsidiaries
- **G.** Blanket Additional Insured Owners, Managers Or Lessors Of Premises

### PROVISIONS

### A. AIRCRAFT CHARTERED WITH PILOT

The following is added to Exclusion g., Aircraft, Auto Or Watercraft, in Paragraph 2. of SECTION I – COVERAGES – COVERAGE A BODILY IN-JURY AND PROPERTY DAMAGE LIABILITY:

This exclusion does not apply to an aircraft that is:

- (a) Chartered with a pilot to any insured;
- (b) Not owned by any insured; and
- (c) Not being used to carry any person or property for a charge.

### B. DAMAGE TO PREMISES RENTED TO YOU

- The first paragraph of the exceptions in Exclusion j., Damage To Property, in Paragraph 2. of SECTION I COVERAGES COVERAGE A BODILY INJURY AND PROPERTY DAMAGE LIABILITY is deleted.
- The following replaces the last paragraph of Paragraph 2., Exclusions, of SECTION I – COVERAGES – COVERAGE A. BODILY

- H. Blanket Additional Insured Lessors Of Leased Equipment
- I. Blanket Additional Insured States Or Political Subdivisions – Permits
- J. Knowledge And Notice Of Occurrence Or Offense
- K. Unintentional Omission
- L. Blanket Waiver Of Subrogation
- M. Amended Bodily Injury Definition
- N. Contractual Liability Railroads

### INJURY AND PROPERTY DAMAGE LI-ABILITY:

Exclusions **c.** and **g.** through **n.** do not apply to "premises damage". Exclusion **f.(1)(a)** does not apply to "premises damage" caused by:

- a. Fire;
- b. Explosion;
- c. Lightning;
- **d.** Smoke resulting from such fire, explosion, or lightning; or
- e. Water;

unless Exclusion **f.** of Section **I** – Coverage **A** – Bodily Injury And Property Damage Liability is replaced by another endorsement to this Coverage Part that has Exclusion – All Pollution Injury Or Damage or Total Pollution Exclusion in its title.

A separate limit of insurance applies to "premises damage" as described in Paragraph 6. of SECTION III – LIMITS OF IN-SURANCE. 3. The following replaces Paragraph 6. of SEC-TION III – LIMITS OF INSURANCE:

Subject to **5.** above, the Damage To Premises Rented To You Limit is the most we will pay under Coverage **A** for damages because of "premises damage" to any one premises. The Damage To Premises Rented To You Limit will apply to all "property damage" proximately caused by the same "occurrence", whether such damage results from: fire; explosion; lightning; smoke resulting from such fire, explosion, or lightning; or water; or any combination of any of these causes.

The Damage To Premises Rented To You Limit will be:

- **a.** The amount shown for the Damage To Premises Rented To You Limit on the Declarations of this Coverage Part; or
- **b.** \$300,000 if no amount is shown for the Damage To Premises Rented To You Limit on the Declarations of this Coverage Part.
- The following replaces Paragraph a. of the definition of "insured contract" in the DEFINI-TIONS Section:
  - a. A contract for a lease of premises. However, that portion of the contract for a lease of premises that indemnifies any person or organization for "premises damage" is not an "insured contract";
- 5. The following is added to the **DEFINITIONS** Section:

"Premises damage" means "property damage" to:

- a. Any premises while rented to you or temporarily occupied by you with permission of the owner; or
- **b.** The contents of any premises while such premises is rented to you, if you rent such premises for a period of seven or fewer consecutive days.
- 6. The following replaces Paragraph 4.b.(1)(b) of SECTION IV COMMERCIAL GENERAL LIABILITY CONDITIONS:
  - (b) That is insurance for "premises damage"; or
- Paragraph 4.b.(1)(c) of SECTION IV COMMERCIAL GENERAL LIABILITY CON-DITIONS is deleted.

### C. INCREASED SUPPLEMENTARY PAYMENTS

- 1. The following replaces Paragraph 1.b. of SUPPLEMENTARY PAYMENTS – COVER-AGES A AND B of SECTION I – COVER-AGE:
  - b. Up to \$2,500 for the cost of bail bonds required because of accidents or traffic law violations arising out of the use of any vehicle to which the Bodily Injury Liability Coverage applies. We do not have to furnish these bonds.
- 2. The following replaces Paragraph 1.d. of SUPPLEMENTARY PAYMENTS – COVER-AGES A AND B of SECTION I – COVER-AGES:
  - **d.** All reasonable expenses incurred by the insured at our request to assist us in the investigation or defense of the claim or "suit", including actual loss of earnings up to \$500 a day because of time off from work.

### D. INCIDENTAL MEDICAL MALPRACTICE

1. The following is added to the definition of "occurrence" in the **DEFINITIONS** Section:

"Occurrence" also means an act or omission committed in providing or failing to provide "incidental medical services", first aid or "Good Samaritan services" to a person.

2. The following is added to Paragraph 2.a.(1) of SECTION II – WHO IS AN INSURED:

Paragraph (1)(d) above does not apply to "bodily injury" arising out of providing or failing to provide:

- (i) "Incidental medical services" by any of your "employees" who is a nurse practitioner, registered nurse, licensed practical nurse, nurse assistant, emergency medical technician or paramedic; or
- (ii) First aid or "Good Samaritan services" by any of your "employees" or "volunteer workers", other than an employed or volunteer doctor. Any such "employees" or "volunteer workers" providing or failing to provide first aid or "Good Samaritan services" during their work hours for you will be deemed to be acting within the scope of their employment by you or performing duties related to the conduct of your business.

3. The following is added to Paragraph 5. of SECTION III – LIMITS OF INSURANCE:

For the purposes of determining the applicable Each Occurrence Limit, all related acts or omissions committed in providing or failing to provide "incidental medical services", first aid or "Good Samaritan services" to any one person will be deemed to be one "occurrence".

 The following exclusion is added to Paragraph 2., Exclusions, of SECTION I – COV-ERAGES – COVERAGE A BODILY INJURY AND PROPERTY DAMAGE LIABILITY:

### Sale Of Pharmaceuticals

"Bodily injury" or "property damage" arising out of the willful violation of a penal statute or ordinance relating to the sale of pharmaceuticals committed by, or with the knowledge or consent of, the insured.

5. The following is added to the **DEFINITIONS** Section:

"Incidental medical services" means:

- Medical, surgical, dental, laboratory, x-ray or nursing service or treatment, advice or instruction, or the related furnishing of food or beverages; or
- **b.** The furnishing or dispensing of drugs or medical, dental, or surgical supplies or appliances.

"Good Samaritan services" means any emergency medical services for which no compensation is demanded or received.

6. The following is added to Paragraph 4.b., Excess Insurance, of SECTION IV – COM-MERCIAL GENERAL LIABILITY CONDI-TIONS:

The insurance is excess over any valid and collectible other insurance available to the insured, whether primary, excess, contingent or on any other basis, that is available to any of your "employees" or "volunteer workers" for "bodily injury" that arises out of providing or failing to provide "incidental medical services", first aid or "Good Samaritan services" to any person to the extent not subject to Paragraph **2.a.(1)** of Section **II** – Who Is An Insured.

E. WHO IS AN INSURED – NEWLY ACQUIRED OR FORMED ORGANIZATIONS

The following replaces Paragraph **4.** of **SECTION II – WHO IS AN INSURED**:

- 4. Any organization you newly acquire or form, other than a partnership, joint venture or limited liability company, of which you are the sole owner or in which you maintain the majority ownership interest, will qualify as a Named Insured if there is no other insurance which provides similar coverage to that organization. However:
- **a.** Coverage under this provision is afforded only:
  - (1) Until the 180th day after you acquire or form the organization or the end of the policy period, whichever is earlier, if you do not report such organization in writing to us within 180 days after you acquire or form it; or
  - (2) Until the end of the policy period, when that date is later than 180 days after you acquire or form such organization, if you report such organization in writing to us within 180 days after you acquire or form it, and we agree in writing that it will continue to be a Named Insured until the end of the policy period;
- b. Coverage A does not apply to "bodily injury" or "property damage" that occurred before you acquired or formed the organization; and
- **c.** Coverage **B** does not apply to "personal injury" or "advertising injury" arising out of an offense committed before you acquired or formed the organization.

### F. WHO IS AN INSURED – BROADENED NAMED INSURED – UNNAMED SUBSIDIARIES

The following is added to **SECTION II – WHO IS AN INSURED**:

Any of your subsidiaries, other than a partnership, joint venture or limited liability company, that is not shown as a Named Insured in the Declarations is a Named Insured if you maintain an ownership interest of more than 50% in such subsidiary on the first day of the policy period.

No such subsidiary is an insured for "bodily injury" or "property damage" that occurred, or "personal injury" or "advertising injury" caused by an offense committed after the date, if any, during the policy period, that you no longer maintain an ownership interest of more than 50% in such subsidiary.

### G. BLANKET ADDITIONAL INSURED – OWNERS, MANAGERS OR LESSORS OF PREMISES

# The following is added to **SECTION II – WHO IS AN INSURED**:

Any person or organization that is a premises owner, manager or lessor and that you have agreed in a written contract or agreement to include as an additional insured on this Coverage Part is an insured, but only with respect to liability for "bodily injury", "property damage", "personal injury" or "advertising injury" that:

- a. Is "bodily injury" or "property damage" that occurs, or is "personal injury" or "advertising injury" caused by an offense that is committed, subsequent to the execution of that contract or agreement; and
- **b.** Arises out of the ownership, maintenance or use of that part of any premises leased to you.

The insurance provided to such premises owner, manager or lessor is subject to the following provisions:

- a. The limits of insurance provided to such premises owner, manager or lessor will be the minimum limits which you agreed to provide in the written contract or agreement, or the limits shown on the Declarations, whichever are less.
- **b.** The insurance provided to such premises owner, manager or lessor does not apply to:
  - (1) Any "bodily injury" or "property damage" that occurs, or "personal injury" or "advertising injury" caused by an offense that is committed, after you cease to be a tenant in that premises; or
  - (2) Structural alterations, new construction or demolition operations performed by or on behalf of such premises owner, lessor or manager.
- c. The insurance provided to such premises owner, manager or lessor is excess over any valid and collectible other insurance available to such premises owner, manager or lessor, whether primary, excess, contingent or on any other basis, unless you have agreed in the written contract or agreement that this insurance must be primary to, or noncontributory with, such other insurance, in which case this insurance will be primary to, and non-contributory with, such other insurance.

### H. BLANKET ADDITIONAL INSURED – LESSORS OF LEASED EQUIPMENT

# The following is added to **SECTION II – WHO IS AN INSURED**:

Any person or organization that is an equipment lessor and that you have agreed in a written contract or agreement to include as an insured on this Coverage Part is an insured, but only with respect to liability for "bodily injury", "property damage", "personal injury" or "advertising injury" that:

- a. Is "bodily injury" or "property damage" that occurs, or is "personal injury" or "advertising injury" caused by an offense that is committed, subsequent to the execution of that contract or agreement; and
- b. Is caused, in whole or in part, by your acts or omissions in the maintenance, operation or use of equipment leased to you by such equipment lessor.

The insurance provided to such equipment lessor is subject to the following provisions:

- a. The limits of insurance provided to such equipment lessor will be the minimum limits which you agreed to provide in the written contract or agreement, or the limits shown on the Declarations, whichever are less.
- b. The insurance provided to such equipment lessor does not apply to any "bodily injury" or "property damage" that occurs, or "personal injury" or "advertising injury" caused by an offense that is committed, after the equipment lease expires.
- c. The insurance provided to such equipment lessor is excess over any valid and collectible other insurance available to such equipment lessor, whether primary, excess, contingent or on any other basis, unless you have agreed in the written contract or agreement that this insurance must be primary to, or non-contributory with, such other insurance, in which case this insurance will be primary to, and non-contributory with, such other insurance.

### I. BLANKET ADDITIONAL INSURED – STATES OR POLITICAL SUBDIVISIONS – PERMITS

# The following is added to **SECTION II – WHO IS AN INSURED**:

Any state or political subdivision that has issued a permit in connection with operations performed by you or on your behalf and that you are required by any ordinance, law or building code to include as an additional insured on this Coverage Part is an insured, but only with respect to liability for "bodily injury", "property damage", "personal injury" or "advertising injury" arising out of such operations.

The insurance provided to such state or political subdivision does not apply to:

- a. Any "bodily injury," "property damage," "personal injury" or "advertising injury" arising out of operations performed for that state or political subdivision; or
- **b.** Any "bodily injury" or "property damage" included in the "products-completed operations hazard".

### J. KNOWLEDGE AND NOTICE OF OCCUR-RENCE OR OFFENSE

The following is added to Paragraph 2., Duties In The Event of Occurrence, Offense, Claim or Suit, of SECTION IV – COMMERCIAL GEN-ERAL LIABILITY CONDITIONS:

- The following provisions apply to Paragraph
   a. above, but only for the purposes of the insurance provided under this Coverage Part to you or any insured listed in Paragraph 1. or 2. of Section II Who Is An Insured:
  - (1) Notice to us of such "occurrence" or offense must be given as soon as practicable only after the "occurrence" or offense is known by you (if you are an individual), any of your partners or members who is an individual (if you are a partnership or joint venture), any of your managers who is an individual (if you are a limited liability company), any of your "executive officers" or directors (if you are an organization other than a partnership, joint venture or limited liability company) or any "employee" authorized by you to give notice of an "occurrence" or offense.
  - (2) If you are a partnership, joint venture or limited liability company, and none of your partners, joint venture members or managers are individuals, notice to us of such "occurrence" or offense must be given as soon as practicable only after the "occurrence" or offense is known by:
    - (a) Any individual who is:
      - (i) A partner or member of any partnership or joint venture;

- (ii) A manager of any limited liability company; or
- (iii) An executive officer or director of any other organization;

that is your partner, joint venture member or manager; or

- (b) Any "employee" authorized by such partnership, joint venture, limited liability company or other organization to give notice of an "occurrence" or offense.
- (3) Notice to us of such "occurrence" or of an offense will be deemed to be given as soon as practicable if it is given in good faith as soon as practicable to your workers' compensation insurer. This applies only if you subsequently give notice to us of the "occurrence" or offense as soon as practicable after any of the persons described in Paragraphs e. (1) or (2) above discovers that the "occurrence" or offense may result in sums to which the insurance provided under this Coverage Part may apply.

However, if this Coverage Part includes an endorsement that provides limited coverage for "bodily injury" or "property damage" or pollution costs arising out of a discharge, release or escape of "pollutants" which contains a requirement that the discharge, release or escape of "pollutants" must be reported to us within a specific number of days after its abrupt commencement, this Paragraph **e.** does not affect that requirement.

### K. UNINTENTIONAL OMISSION

The following is added to Paragraph 6., Representations, of SECTION IV – COMMERCIAL GENERAL LIABILITY CONDITIONS:

The unintentional omission of, or unintentional error in, any information provided by you which we relied upon in issuing this policy will not prejudice your rights under this insurance. However, this provision does not affect our right to collect additional premium or to exercise our rights of cancellation or nonrenewal in accordance with applicable insurance laws or regulations.

### L. BLANKET WAIVER OF SUBROGATION

The following is added to Paragraph 8., Transfer Of Rights Of Recovery Against Others To Us, of SECTION IV – COMMERCIAL GENERAL LI-ABILITY CONDITIONS: If the insured has agreed in a contract or agreement to waive that insured's right of recovery against any person or organization, we waive our right of recovery against such person or organization, but only for payments we make because of:

- a. "Bodily injury" or "property damage" that occurs; or
- **b.** "Personal injury" or "advertising injury" caused by an offense that is committed;

subsequent to the execution of that contract or agreement.

### M. AMENDED BODILY INJURY DEFINITION

The following replaces the definition of "bodily injury" in the **DEFINITIONS** Section:

**3.** "Bodily injury" means bodily injury, mental anguish, mental injury, shock, fright, disability, humiliation, sickness or disease sustained by a person, including death resulting from any of these at any time.

### N. CONTRACTUAL LIABILITY - RAILROADS

- The following replaces Paragraph c. of the definition of "insured contract" in the DEFINI-TIONS Section:
  - c. Any easement or license agreement;
- 2. Paragraph f.(1) of the definition of "insured contract" in the **DEFINITIONS** Section is deleted.

COMMERCIAL GENERAL LIABILITY ISSUE DATE: 6/1/2018

THIS ENDORSEMENT CHANGES THE POLICY. PLEASE READ IT CAREFULLY

## DESIGNATED PROJECT(S) GENERAL AGGREGATE LIMIT

This endorsement modifies insurance provided under the following: COMMERCIAL GENERAL LIABILITY COVERAGE PART

### SCHEDULE

### **Designated Project(s):**

EACH "PROJECT" FOR WHICH YOU HAVE AGREED, IN A WRITTEN CONTRACT WHICH IS IN EFFECT DURING THIS POLICY PERIOD, TO PROVIDE A SEPARATE GENERAL AGGREGATE LIMIT, PROVIDED THAT THE CONTRACT IS SIGNED AND EXECUTED BY YOU BEFORE THE "BODILY INJURY" OR "PROPERTY DAMAGE" OCCURS.

- A. For all sums which the insured becomes legally obligated to pay as damages caused by "occurrences" under COVERAGE A. (SECTION I), and for all medical expenses caused by accidents under COVERAGE C (SECTION I), which can be attributed only to operations at a single designated "project" shown in the Schedule above:
  - A separate Designated Project General Aggregate Limit applies to each designated "project", and that limit is equal to the amount of the General Aggregate Limit shown in the Declarations, unless separate Designated Project General Aggregate(s) are scheduled above.
  - The Designated Project General Aggregate Limit is the most we will pay for the sum of all damages under COVERAGE A., except damages because of "bodily injury" or "property damage" included in the "productscompleted operations hazard", and for medical expenses under COVERAGE C, regardless of the number of:
    - a. Insureds;
    - **b.** Claims made or "suits" brought; or
    - c. Persons or organizations making claims or bringing "suits".

Designated Project General Aggregate(s):

GENERAL AGGREGATE LIMIT SHOWN ON THE DECLARATIONS.

- 3. Any payments made under COVERAGE A. for damages or under COVERAGE C. for medical expenses shall reduce the Designated Project General Aggregate Limit for that designated "project". Such payments shall not reduce the General Aggregate Limit shown in the Declarations nor shall they reduce any other Designated Project General Aggregate Limit for any other designated "project" shown in the Schedule above.
- 4. The limits shown in the Declarations for Each Occurrence, Damage To Premises Rented To You and Medical Expense continue to apply. However, instead of being subject to the General Aggregate Limit shown in the Declarations, such limits will be subject to the applicable Designated Project General Aggregate Limit.
- B. For all sums which the insured becomes legally obligated to pay as damages caused by "occurrences" under COVERAGE A. (SECTION I), and for all medical expenses caused by accidents under COVERAGE C. (SECTION I), which cannot be attributed only to operations at a single designated "project" shown in the Schedule above:

### COMMERCIAL GENERAL LIABILITY

- Any payments made under COVERAGE A. for damages or under COVERAGE C. for medical expenses shall reduce the amount available under the General Aggregate Limit or the Products-Completed Operations Aggregate Limit, whichever is applicable; and
- 2. Such payments shall not reduce any Designated Project General Aggregate Limit.
- C. Part 2. of SECTION III LIMITS OF INSURANCE is deleted and replaced by the following:
  - 2. The General Aggregate Limit is the most we will pay for the sum of:
    - a. Damages under Coverage B; and
    - b. Damages from "occurrences" under COVERAGE A (SECTION I) and for all medical expenses caused by accidents under COVERAGE C (SECTION I) which cannot be attributed only to operations at a single designated "project" shown in the SCHEDULE above.
- D. When coverage for liability arising out of the "products-completed operations hazard" is pro-

vided, any payments for damages because of "bodily injury" or "property damage" included in the "products-completed operations hazard" will reduce the Products-Completed Operations Aggregate Limit, and not reduce the General Aggregate Limit nor the Designated Project General Aggregate Limit.

E. For the purposes of this endorsement the **Defini**tions Section is amended by the addition of the following definition:

"Project" means an area away from premises owned by or rented to you at which you are performing operations pursuant to a contract or agreement. For the purposes of determining the applicable aggregate limit of insurance, each "project" that includes premises involving the same or connecting lots, or premises whose connection is interrupted only by a street, roadway, waterway or right-of-way of a railroad shall be considered a single "project".

F. The provisions of SECTION III – LIMITS OF INSURANCE not otherwise modified by this endorsement shall continue to apply as stipulated.

## THIS ENDORSEMENT CHANGES THE POLICY. PLEASE READ IT CAREFULLY.

# **BUSINESS AUTO EXTENSION ENDORSEMENT**

This endorsement modifies insurance provided under the following:

### **BUSINESS AUTO COVERAGE FORM**

GENERAL DESCRIPTION OF COVERAGE – This endorsement broadens coverage. However, coverage for any injury, damage or medical expenses described in any of the provisions of this endorsement may be excluded or limited by another endorsement to the Coverage Part, and these coverage broadening provisions do not apply to the extent that coverage is excluded or limited by such an endorsement. The following listing is a general coverage description only. Limitations and exclusions may apply to these coverages. Read all the provisions of this endorsement and the rest of your policy carefully to determine rights, duties, and what is and is not covered.

- A. BROAD FORM NAMED INSURED
- B. BLANKET ADDITIONAL INSURED
- C. EMPLOYEE HIRED AUTO
- D. EMPLOYEES AS INSURED
- E. SUPPLEMENTARY PAYMENTS INCREASED LIMITS
- F. HIRED AUTO LIMITED WORLDWIDE COV-ERAGE - INDEMNITY BASIS
- G. WAIVER OF DEDUCTIBLE GLASS

### PROVISIONS

A. BROAD FORM NAMED INSURED

The following is added to Paragraph A.1., Who is An insured, of SECTION II – COVERED AUTOS LIABILITY COVERAGE:

Any organization you newly acquire or form during the policy period over which you maintain 50% or more ownership interest and that is not separately insured for Business Auto Coverage. Coverage under this provision is afforded only until the 180th day after you acquire or form the organization or the end of the policy period, whichever is earlier.

### B. BLANKET ADDITIONAL INSURED

The following is added to Paragraph c. in A.1., Who is An insured, of SECTION II - COVERED AUTOS LIABILITY COVERAGE:

Any person or organization who is required under a written contract or agreement between you and that person or organization, that is signed and executed by you before the "bodily injury" or "property damage" occurs and that is in effect during the policy period, to be named as an additional insured is an "insured" for Covered Autos Liability Coverage, but only for damages to which

- H. HIRED AUTO PHYSICAL DAMAGE LOSS OF USE - INCREASED LIMIT
- I. PHYSICAL DAMAGE TRANSPORTATION EXPENSES - INCREASED LIMIT
- J. PERSONAL PROPERTY
- K. AIRBAGS
- L. NOTICE AND KNOWLEDGE OF ACCIDENT OR LOSS
- M. BLANKET WAIVER OF SUBROGATION
- N. UNINTENTIONAL ERRORS OR OMISSIONS

this insurance applies and only to the extent that person or organization qualifies as an "insured" under the Who is An insured provision contained in Section II.

### C. EMPLOYEE HIRED AUTO

1. The following is added to Paragraph A.1., Who is An insured, of SECTION II - COV-ERED AUTOS LIABILITY COVERAGE:

An "employee" of yours is an "insured" while operating an "auto" hired or rented under a contract or agreement in an "employee's" name, with your permission, while performing duties related to the conduct of your business.

- The following replaces Paragraph b. in B.5., Other Insurance, of SECTION IV – BUSI-NESS AUTO CONDITIONS:
  - b. For Hired Auto Physical Damage Coverage, the following are deemed to be covered "autos" you own:
    - (1) Any covered "auto" you lease, hire, rent or borrow; and
    - (2) Any covered "auto" hired or rented by your "employee" under a contract in an "employee's" name, with your

permission, while performing duties related to the conduct of your business.

However, any "auto" that is leased, hired, rented or borrowed with a driver is not a covered "auto".

### D. EMPLOYEES AS INSURED

The following is added to Paragraph A.1., Who is An insured, of SECTION II – COVERED AUTOS LIABILITY COVERAGE:

Any "employee" of yours is an "insured" while using a covered "auto" you don't own, hire or borrow in your business or your personal affairs.

- E. SUPPLEMENTARY PAYMENTS INCREASED LIMITS
  - 1. The following replaces Paragraph A.2.a.(2), of SECTION II – COVERED AUTOS LIABIL-ITY COVERAGE:
    - (2) Up to \$3,000 for cost of bail bonds (including bonds for related traffic law violations) required because of an "accident" we cover. We do not have to furnish these bonds.
  - The following replaces Paragraph A.2.a.(4), of SECTION II – COVERED AUTOS LIABIL-ITY COVERAGE:
    - (4) All reasonable expenses incurred by the "insured" at our request, including actual loss of earnings up to \$500 a day because of time off from work.
- F. HIRED AUTO LIMITED WORLDWIDE COV-ERAGE - INDEMNITY BASIS

The following replaces Subparagraph (5) in Paragraph B.7., Policy Period, Coverage Territory, of SECTION IV – BUSINESS AUTO CONDI-TIONS:

(5) Anywhere in the world, except any country or jurisdiction while any trade sanction, embargo, or similar regulation imposed by the United States of America applies to and prohibits the transaction of business with or within such country or jurisdiction, for Covered Autos Liability Coverage for any covered "auto" that you lease, hire, rent or borrow without a driver for a period of 30 days or less and that is not an "auto" you lease, hire, rent or borrow from any of your "employees", partners (if you are a partnership), members (if you are a limited liability company) or members of their households.

- (a) With respect to any claim made or "suit" brought outside the United States of America, the territories and possessions of the United States of America, Puerto Rico and Canada:
  - (I) You must arrange to defend the "insured" against, and investigate or settle any such claim or "suit" and keep us advised of all proceedings and actions.
  - (ii) Neither you nor any other involved "insured" will make any settlement without our consent.
  - (III) We may, at our discretion, participate in defending the "insured" against, or in the settlement of, any claim or "sult".
  - (iv) We will reimburse the "insured" for sums that the "insured" legally must pay as damages because of "bodily injury" or "property damage" to which this insurance applies, that the "insured" pays with our consent, but only up to the limit described in Paragraph C., Limits Of Insurance, of SECTION II – COVERED AUTOS LIABILITY COVERAGE.
  - (v) We will reimburse the "insured" for the reasonable expenses incurred with our consent for your investigation of such claims and your defense of the "insured" against any such "suit", but only up to and included within the limit described in Paragraph C., Limits Of Insurance, of SECTION II – COVERED AUTOS LIABILITY COVERAGE, and not in addition to such limit. Our duty to make such payments ends when we have used up the applicable limit of insurance in payments for damages, settlements or defense expenses.
- (b) This insurance is excess over any valid and collectible other insurance available to the "insured" whether primary, excess, contingent or on any other basis.
- (c) This insurance is not a substitute for required or compulsory insurance in any country outside the United States, its territories and possessions, Puerto Rico and Canada.

You agree to maintain all required or compulsory insurance in any such country up to the minimum limits required by local law. Your failure to comply with compulsory insurance requirements will not invalidate the coverage afforded by this policy, but we will only be liable to the same extent we would have been liable had you complied with the compulsory insurance requirements.

(d) It is understood that we are not an admitted or authorized insurer outside the United States of America, its territories and possessions, Puerto Rico and Canada. We assume no responsibility for the furnishing of certificates of insurance, or for compliance in any way with the laws of other countries relating to insurance.

### G. WAIVER OF DEDUCTIBLE - GLASS

The following is added to Paragraph D., Deductible, of SECTION III - PHYSICAL DAMAGE COVERAGE:

No deductible for a covered "auto" will apply to glass damage if the glass is repaired rather than replaced.

H. HIRED AUTO PHYSICAL DAMAGE - LOSS OF USE - INCREASED LIMIT

The following replaces the last sentence of Paragraph A.4.b., Loss Of Use Expenses, of SEC-TION III – PHYSICAL DAMAGE COVERAGE:

However, the most we will pay for any expenses for loss of use is \$65 per day, to a maximum of \$750 for any one "accident".

I. PHYSICAL DAMAGE - TRANSPORTATION EXPENSES - INCREASED LIMIT

The following replaces the first sentence in Paragraph A.4.a., Transportation Expenses, of SECTION III – PHYSICAL DAMAGE COVER-AGE:

We will pay up to \$50 per day to a maximum of \$1,500 for temporary transportation expense incurred by you because of the total theft of a covered "auto" of the private passenger type.

### J. PERSONAL PROPERTY

The following is added to Paragraph A.4., Coverage Extensions, of SECTION III – PHYSICAL DAMAGE COVERAGE:

### Personal Property

We will pay up to \$400 for "loss" to wearing apparel and other personal property which is:

(1) Owned by an "insured"; and

(2) In or on your covered "auto".

This coverage applies only in the event of a total theft of your covered "auto".

No deductibles apply to this Personal Property coverage.

### K. AIRBAGS

The following is added to Paragraph B.3., Exclusions, of SECTION III – PHYSICAL DAMAGE COVERAGE:

Exclusion 3.a. does not apply to "loss" to one or more alrbags in a covered "auto" you own that inflate due to a cause other than a cause of "loss" set forth in Paragraphs A.1.b. and A.1.c., but only:

- a. If that "auto" is a covered "auto" for Comprehensive Coverage under this policy;
- b. The airbags are not covered under any warranty; and
- c. The airbags were not intentionally inflated.

We will pay up to a maximum of \$1,000 for any one "loss".

### L. NOTICE AND KNOWLEDGE OF ACCIDENT OR LOSS

The following is added to Paragraph A.2.a., of SECTION IV - BUSINESS AUTO CONDITIONS:

Your duty to give us or our authorized representative prompt notice of the "accident" or "loss" applies only when the "accident" or "loss" is known to:

- (a) You (if you are an individual);
- (b) A partner (if you are a partnership);
- (c) A member (if you are a limited liability company);
- (d) An executive officer, director or insurance manager (if you are a corporation or other organization); or
- (e) Any "employee" authorized by you to give notice of the "accident" or "ioss".

### M. BLANKET WAIVER OF SUBROGATION

The following replaces Paragraph A.5., Transfer Of Rights Of Recovery Against Others To Us, of SECTION IV – BUSINESS AUTO CONDI-TIONS:

### 5. Transfer Of Rights Of Recovery Against Others To Us

We walve any right of recovery we may have against any person or organization to the extent required of you by a written contract signed and executed prior to any "accident" or "loss", provided that the "accident" or "loss" arises out of operations contemplated by such contract. The waiver applies only to the person or organization designated in such contract.

### N. UNINTENTIONAL ERRORS OR OMISSIONS

The following is added to Paragraph B.2., Concealment, Misrepresentation, Or Fraud, of SECTION IV - BUSINESS AUTO CONDITIONS: The unintentional omission of, or unintentional error in, any information given by you shall not prejudice your rights under this insurance. However this provision does not affect our right to collect additional premium or exercise our right of cancellation or non-renewal.

# **saif**corporation

### Carrier No:

20001

Endorsement No:

WC000313

Policy No:

810540

Agency:

EAGLE-ELSNER INC PO BOX 23294 PORTLAND, OR 97281-3294

MARC BRANDEBERRY ANCHOR INSURANCE & SURETY INC 1201 SW 12TH AVE STE 500 PORTLAND, OR 97205

### Waiver of Our Right to Recover From Others Endorsement

We have the right to recover our payments from anyone liable for an injury covered by this policy. We will not enforce our right against the person or organization named in the Schedule.

This agreement shall not operate directly or indirectly to benefit anyone not named in the Schedule.

	Schedule
Description:	ALL OPERATIONS
Contractor Name:	PERSONS AND/OR ORGANIZATIONS WITH WHOM THE
	INSURED-EMPLOYER IS REQUIRED BY WRITTEN
	CONTRACT TO WAIVE SUBROGATION RIGHTS

This endorsement does not alter the rights of an injured worker to pursue recovery from another party or SAIF to receive a statutory share of recoveries by an injured worker, even from the party listed in the schedule.

The premium charge for this endorsement is based on one (1) percent of your manual premium.

Effective Date: 10-01-2018 This endorsement is part of your policy. This endorsement amends and controls anything to the contrary. It is otherwise subject to all other terms of your policy.

Countersigned 10-15-2018 at Salem, Oregon

Kerry Barnett, President

and Chief Executive Officer

430b

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

# STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

ith an attorney is encouraged with ad to the particular circumstances of	To prepare supplementary conditions that are coordinated with the General Conditions, use EJCDC's Guide to the Preparation of Supplementary Conditions (EJCDC® C-800, 2013 Edition). The full EJCDC Construction series of documents is discussed in the Commentary on the 2013 EJCDC Construction VTRACT	Copyright © 2013:	National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882 www.nspe.org	Ame 1015	American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723 <u>www.asce.org</u>	<b>ers</b> ® The copyright for this document is owned jointly by the three sponsoring organizations listed above. The National Society of Professional Engineers is the Copyright Administrator for the EJCDC documents; please direct all inquiries regarding EJCDC copyrights to NSPE.	NOTE: EJCDC publications may be purchased at <u>www.eicdc.org</u> , or from any of the sponsoring organizations above.	uction Contract. 10 f Bigineering Companies, Copyright © 2013 National Society of Could Biotensional Enters, American Council of Enters, American Council of Enter road.
respect to its unportant regar consequences; consultation with an attorney is respect to its use or modification. This document should be adapted to the particular the contemplated Project and the controlling Laws and Regulations.	STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT	Prepared by	EJCDC ENGINEERS JOINT CONTRACT DOCUMENTS COMMITTEE	Issued and Published Jointly by AMERICAN COUNCIL OF EXCINEMING COMPANIES	ARREAN SOLIE / OF ONE MAINE HES	Professional Engineers		EICDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers. American Cound of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

ŧ

Article 2 – Preliminary Matters..... Article 1 – Definitions and Terminology .... Article 3 - Documents: Intent, Requirements, Reuse Article 4 – Commencement and Progress of the Work... Article 5 – Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental Conditions ..... 1.02 1.01 2.01 2.04 2.03 2.02 3.02 3.01 2.05 3.03 2.06 4.01 3.05 3.04 4.04 4.03 4.02 5.01 4.05 5.03 5.04 5.02 5.05 Defined Terms . Delivery of Bonds and Evidence of Insurance Terminology.. Reference Standards Electronic Transmittals. Initial Acceptance of Schedules. Preconstruction Conference; Designation of Authorized Representatives Before Starting Construction Copies of Documents.. Reporting and Resolving Discrepancies... Intent ... Delays in Contractor's Progress. Commencement of Contract Times; Notice to Proceed Reuse of Documents.. Requirements of the Contract Documents Starting the Work . Progress Schedule . **Reference** Points .. Use of Site and Other Areas .. Underground Facilities . Differing Subsurface or Physical Conditions. Subsurface and Physical Conditions. Availability of Lands. ECCC\* C.700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 Netional Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page i STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT ..... TABLE OF CONTENTS ÷ Page .... 12 . თ . 10 . 10 σ . m . Б 10 . ∞ ...12 ...11 : ∞ .10 .10 .10 6 : 15 . 14 :13 .. 12 ò

EJCDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page ii	
Legal Relationships	8.03
Coordination	8.02
01 Other Work	8.01
- Of	Article 8
Delegation of Professional Design Services	7.19
	7.18
	7.17
Shop Drawings, Samples, and Other Submittals	7.16
Emergencies	7.15
Hazard Communication Programs	7.14
Safety Representative	7.13
2 Safety and Protection	7.12
Record Documents	7.11
	7.10
	7.09
3 Permits	7.08
patent Fees and Royalties	7.07
Concerning Subcontractors, Suppliers,	7.06
Substitutes	7.05
"Or Equals"	7.04
aterials, and Equipment	7.03
Labor, Working Hours	7.02
intendence	7.01
	Article 7 –
Receipt and Application of Property Insurance Process	6.07
	6.06
Property Insurance	6.05
Owner's Liability Insurance	6.04
	6.03
Insurance—General Provisions	6.02
Performance, Payment, and Other Bonds19	4
Bonds and Insurance	Article 6 – B
Hazardous Environmental Conditions at Site17	5.06

P

7

7

ר

2

نہ ۔

Cont - O

Article 9 –	۰۵ مرم . Owner's Responsibilities
9.01	ractor
9.02	Replacement of Engineer
9.03	Furnish Data40
9.04	Pay When Due40
9.05	Lands and Easements, Reports, Tests, and Drawings
90.6	Insurance
9.07	Change Orders
9.08	Inspections, Tests, and Approvals41
60.6	Limitations on Owner's Responsibilities41
9.10	Undisclosed Hazardous Environmental Condition41
9.11	Evidence of Financial Arrangements41
9.12	Safety Programs
Article 10 –	- Engineer's Status During Construction41
10.01	Owner's Representative
10.02	Visits to Site
10.03	Project Representative
10.04	Rejecting Defective Work
10.05	Shop Drawings, Change Orders and Payments
10.06	Determinations for Unit Price Work42
10.07	Decisions on Requirements of Contract Documents and Acceptability of Work42
10.08	Limitations on Engineer's Authority and Responsibilities42
10.09	Compliance with Safety Program43
Article 11 –	Amending the Contract Documents; Changes in the Work
11.01	Amending and Supplementing Contract Documents43
11.02	Owner-Authorized Changes in the Work
11.03	Unauthorized Changes in the Work44
11.04	Change of Contract Price
11.05	Change of Contract Times45
11.06	Change Proposals45
11.07	Execution of Change Orders46
11.08	Notification to Surety47
Article 12 – (	Claims
	ELCDC* C.700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Courcil of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page III

12.01	Claims
Article 13	.cost of the Work; Allowances; Unit Price Work
13.01	Cost of the Work
13.02	Allowances
13.03	Unit Price Work
Article 14 –	Tests and Inspections; Correction, Removal or Acceptance of Defective Work
14.01	
14.02	Tests, Inspections, and Approvals52
14.03	Defective Work
14.04	Acceptance of Defective Work
14.05	Uncovering Work
14.06	Owner May Stop the Work54
14.07	Owner May Correct Defective Work54
Article 15 – I	Payments to Contractor; Set-Offs; Completion; Correction Period55
15.01	
15.02	Contractor's Warranty of Title
15.03	Substantial Completion
15.04	Partial Use or Occupancy
15.05	Final Inspection
15.06	Final Payment
15.07	Waiver of Claims
15.08	Correction Period
Article 16 – S	Suspension of Work and Termination62
16.01	Owner May Suspend Work
16.02	Owner May Terminate for Cause62
16.03	Owner May Terminate For Convenience63
16.04	Contractor May Stop Work or Terminate63
Article 17 – Fi	Final Resolution of Disputes64
17.01	Methods and Procedures64
Article 18 – N	Miscellaneous
18.01 0	Giving Notice
18.02	Computation of Times64
18.03 (	Cumulative Remedies64
3	EICDC <sup>4</sup> C-700 [Rev. 1], standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Fuerinear A marices Consult of Economic and
	errigen of companies outery or routessional cligitheers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page iv

18.08 Headings
18 N7 Controlling Law
18.06 Survival of Obligations
18.05 No Waiver
18.04 Limitation of Damages

# ARTICLE 1 – DEFINITIONS AND TERMINOLOGY

# 1.01 Defined Terms

- A. Wherever used in the Bidding Requirements or Contract Documents, a term printed with initial capital letters, including the term's singular and plural forms, will have the meaning indicated in the definitions below. In addition to terms specifically defined, terms with initial capital letters in the Contract Documents include references to identified articles and paragraphs, and the titles of other documents or forms.
- Addenda—Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract
- Documents. 2. Agreement—The written instrument, executed by Owner and Contractor, that sets forth the Contract Price and Contract Times, identifies the parties and the Engineer, and designates the specific items that are Contract Documents.
- Application for Poyment—The form acceptable to Engineer which is to be used by Contractor during the course of the Work in requesting progress or final payments and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
- Bid—The offer of a Bidder submitted on the prescribed form setting forth the prices
- for the Work to be performed. *Bidder*—An individual or entity that submits a Bid to Owner.
- Bidder—An inclutional or entry true summer errors
   Bidding Documents—The Bidding Requirements, the proposed Contract Documents,
- and all Addenda.
  Bidding Requirements—The advertisement or invitation to bid, instructions to Bidders,
  Bid Bond or other Bid security, if any, the Bid Form, and the Bid with any attachments.
- Change Order—A document which is signed by Contractor and Owner and authorizes an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, or other revision to the Contract, issued on or after the Effective Date of the Contract.
- 9. Change Proposal—A written request by Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment in Contract Price or Contract Times, or both; contesting an initial decision by Engineer concerning the requirements of the Contract Documents or the acceptability of Work under the Contract Documents; challenging a set-off against payments due; or seeking other Contract Documents; challenging a set-off against payments due; or seeking other
- relief with respect to the terms of the Contract. 10. *Claim*—(a) A demand or assertion by Owner directly to Contractor, duly submitted in compliance with the procedural requirements set forth herein: seeking an adjustment of Contract Price or Contract Times, or both; contesting an initial decision by Engineer of contract Price or Contract Times, or both; contesting an initial decision by Engineer under the Contract Documents of the Contract Documents or the acceptability of Work concerning the requirements of the Contractual issue that Engineer has decision regarding a Change under the Contract Documents; contesting Engineer's decision regarding a Change demand or assertion by Contractor directly to Owner, duly submitted in compliance demand or assertion by Contractor directly to Owner, ully submitted in compliance with the procedural requirements set forth herein, contractual issue that Engineer's decision with the procedural requirements or the construction Contract.

EICDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 1 of 65

ELCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page v

٦

24. <i>Liens</i> —Charges, security interests, or encumbrances upon Contract-related funds, real property, or personal property.	25. Milestone—A principal event in the performance of the Work that the Contract requires Contractor to achieve by an intermediate completion date or by a time prior to Substantial Completion of all the Work.	<ol> <li>Notice of Award—The written notice by Owner to a Bidder of Owner's acceptance of the Bid.</li> </ol>	27. Notice to Proceed—A written notice by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work.	28. <i>Owner</i> —The individual or entity with which Contractor has contracted regarding the Work, and which has agreed to pay Contractor for the performance of the Work, pursuant to the terms of the Contract.	29. <i>Progress Schedule—</i> A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising the Contractor's plan to accomplish the Work within the Contract Times.	30. <i>Project</i> —The total undertaking to be accomplished for Owner by engineers, contractors, and others, including planning, study, design, construction, testing, commissioning, and start-up, and of which the Work to be performed under the Contract Documents is a part.	31. Project Manual—The written documents prepared for, or made available for, procuring and constructing the Work, including but not limited to the Bidding Documents or other construction procurement documents, reotechnical and existine	conditions information, the Agreement, bond forms, General Conditions, Supplementary Conditions, and Specifications. The contents of the Project Manual may be bound in one or more volumes.	32. Resident Project Representative—The authorized representative of Engineer assigned to assist Engineer at the Site. As used herein, the term Resident Project Representative	or "KPK" includes any assistants or field staff of Resident Project Representative. 33. <i>Samples</i> —Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and that establish the standards by which	such portion of the Work will be judged. 34. <i>Schedule of Submittals—</i> A schedule, prepared and maintained by Contractor, of	required submittals and the time requirements for Engineer's review of the submittals and the performance of related construction activities. 35. Schedule of Volues-A schedule, prepared and maintained by Contractor, allocating	portions of the Contract Fride to Various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment. 36. Shop Drowings–All drawings, diagrams, illustrations, schedules, and other data or information that are schefficially interacted or assembled humber for conservance and	whether approved or not, are not Drawings and are not Contract Documents.	ELCDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 3 of 65
has declined to address. A demand for money or services by a third party is not a Claim.		Liability Act, 42 U.S.C. §59601 et seq. ("CERCLA"); (b) the Hazardous Materials Transportation Act, 49 U.S.C. §55101 et seq. ("CERCLA"); (b) the Hazardous Materials Transportation Act, 49 U.S.C. §55101 et seq.: (c) the Resource Conservation and	Recovery Act, 42 U.S.C. §§6901 et seq. ("RCRA"); (d) the Toxic Substances Control Act, 15 U.S.C. §§2601 et seq.; (e) the Clean Water Act, 33 U.S.C. §§1251 et seq.; (f) the Clean Air Act, 42 U.S.C. §§7401 et seq.; or (g) any other federal, state, or local statute,		<ol> <li>Contract— Ine entire, and integrated written contract between the Owner and Contractor concerning the Work.</li> <li>Contract Documents—Those items so designated in the Agreement, and which</li> </ol>	together comprise the Contract. <ol> <li>Contract Price—The money that Owner has agreed to pay Contractor for completion of the Work in accordance with the Contract Documents.</li> <li>Contract Times—The number of days or the dates by which Contract challed to be a set of the Contract Contract Times.</li> </ol>	achieve Milestones, if any; (b) achieve Substantial Completion; and (c) complete the Work. Work. 16. <i>Contractor</i> —The individual or entity with which Owner has contracted for	13.01 for definition.		<ol> <li>Effective Date of the Contract—The date, indicated in the Agreement, on which the Contract becomes effective.</li> <li>Engineer—The individual or entity named as such in the Agreement.</li> </ol>	21. <i>Field Order</i> —A written order issued by Engineer which requires minor changes in the Work but does not change the Contract Price or the Contract Times.	22. Hazordous Environmental Condition—The presence at the Site of Constituents of Concern in such quantities or circumstances that may present a danger to persons or property exposed thereto. The presence at the Site of materials that are necessary for	the execution of the Work, or that are to be incorporated in the Work, and that are controlled and contained pursuant to industry practices, Laws and Regulations, and the requirements of the Contract, does not establish a Hazardous Environmental Condition.	23. Laws and Regulations; Laws or Regulations—Any and all applicable laws, statutes, rules, regulations, ordinances, codes, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.	EICDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers, All rights reserved. Page 2 of 65

- 37. Site-Lands or areas indicated in the Contract Documents as being furnished by Owner such other lands furnished by Owner which are designated for the use of Contractor. upon which the Work is to be performed, including rights-of-way and easements, and
- 38 Specifications-The part of the Contract that consists of written requirements for and certain administrative requirements and procedural matters applicable to the materials, equipment, systems, standards, and workmanship as applied to the Work,
- 39. Subcontractor-An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work.

Work.

- 40. Substantial Completion—The time at which the Work (or a specified part thereof) has applied to all or part of the Work refer to Substantial Completion thereof. is intended. The terms "substantially complete" and "substantially completed" as that the Work (or a specified part thereof) can be utilized for the purposes for which it part thereof) is sufficiently complete, in accordance with the Contract Documents, so progressed to the point where, in the opinion of Engineer, the Work (or a specified
- 41. Successful Bidder—The Bidder whose Bid the Owner accepts, and to which the Owner makes an award of contract, subject to stated conditions.
- 42. Supplementary Conditions-The part of the Contract that amends or supplements these General Conditions.
- 43 Supplier-A manufacturer, fabricator, supplier, distributor, materialman, or vendor or equipment to be incorporated in the Work by Contractor or a Subcontractor. having a direct contract with Contractor or with any Subcontractor to furnish materials
- 44. Technical Data-Those items expressly identified as Technical Data in the of subsurface water levels, laboratory test results, and other factual, objective conditions at the Site, then the data contained in boring logs, recorded measurements If no such express identifications of Technical Data have been made with respect to (except Underground Facilities) or (b) Hazardous Environmental Conditions at the Site. or physical conditions relating to existing surface or subsurface structures at the Site Supplementary Conditions, with respect to either (a) subsurface conditions at the Site, Paragraphs 5.03, 5.04, and 5.06. environmental report prepared for the Project and made available to Contractor are information regarding conditions at the Site that are set forth in any geotechnical or hereby defined as Technical Data with respect to conditions at the Site under
- 45. Underground Facilities-All underground pipelines, conduits, ducts, cables, wires, encasements containing such facilities, including but not limited to those that convey manholes, vaults, tanks, tunnels, or other such facilities or attachments, and any water, other liquids or chemicals, or traffic or other control systems. communications, fiber optic transmissions, cable television, water, wastewater, storm electricity, gases, steam, liquid petroleum products, telephone or other
- 46. Unit Price Work—Work to be paid for on the basis of unit prices.
- 47. Work—The entire construction or the various separately identifiable parts thereof start-up, and commissioning, all as required by the Contract Documents. equipment into such construction; and may include related services such as testing, produce such construction; furnishing, installing, and incorporating all materials and result of performing or providing all labor, services, and documentation necessary to required to be provided under the Contract Documents. Work includes and is the

EICDC® C-700 (Rev. 1), Standard General Conditions of the Construction Control Con-Copyright © 2013 National Society of Professional Engineers, American Council of Foigineering Companies, and American Society of Could Engineers. All rights reserved. Page 4 of 65 and American Society of Civil Engineers. All rights reserved.

10

 $\overline{}$ 

3 J

1.13

لي

- 48. Work Change Directive-A written directive to Contractor issued on or after the ordering an addition, deletion, or revision in the Work. Effective Date of the Contract, signed by Owner and recommended by Engineer,
- 1.02 Terminology
- ₽ The words and terms discussed in the following paragraphs are not defined but, when used in the Bidding Requirements or Contract Documents, have the indicated meaning
- Intent of Certain Terms or Adjectives:
- The Contract Documents include the terms "as allowed," "as approved," "as ordered," the performance of the Work, or any duty or authority to undertake responsibility contrary to the provisions of Article 10 or any other provision of the Contract shall not be effective to assign to Engineer any duty or authority to supervise or direct indicating otherwise). The use of any such term or adjective is not intended to and shown or indicated in the Contract Documents (unless there is a specific statement Documents and with the design concept of the Project as a functioning whole evaluate, in general, the Work for compliance with the information in the Contract such exercise of professional judgment, action, or determination will be solely to describe an action or determination of Engineer as to the Work. It is intended that judgment by Engineer. In addition, the adjectives "reasonable," "suitable," "as directed" or terms of like effect or import to authorize an exercise of professional "acceptable," "proper," "satisfactory," or adjectives of like effect or import are used to Documents. g
- Day

۲

- The word "day" means a calendar day of 24 hours measured from midnight to the next midnight.
- 0 Defective:
- ۲ The word "defective," when modifying the word "Work," refers to Work that is unsatisfactory, faulty, or deficient in that it:
- 'n does not conform to the Contract Documents; or
- ē does not meet the requirements of any applicable inspection, reference standard, test, or approval referred to in the Contract Documents; or
- ņ has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner Substantial Completion in accordance with Paragraph 15.03 or 15.04). at
- 'n Furnish, Install, Perform, Provide
- ۲ The word "furnish," when used in connection with services, materials, or equipment, some other specified location) ready for use or installation and in usable or operable shall mean to supply and deliver said services, materials, or equipment to the Site (or condition.
- shall mean to put into use or place in final position said services, materials, or The word "install," when used in connection with services, materials, or equipment, equipment complete and ready for intended use.

Ν

ECOC<sup>e</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 Netional Society of Professional Engineers, American Council of Engineering Companies and American Society of Civil Engineers. All rights reserved. Page 5 of 65

- 4. If the Contract Documents establish an obligation of Contractor with respect to specific services, materials, or equipment, but do not expressly use any of the four words "furnish," "install," "perform," or "provide," then Contractor shall furnish and install said services, materials, or equipment complete and ready for intended use.
- F. Unless stated otherwise in the Contract Documents, words or phrases that have a wellknown technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

# **ARTICLE 2 – PRELIMINARY MATTERS**

# 2.01 Delivery of Bonds and Evidence of Insurance

- Bonds: When Contractor delivers the executed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner such bonds as Contractor may be required to furnish.
- B. Evidence of Contractor's Insurance: When Contractor delivers the executed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner, with copies to each named insured and additional insured (as identified in the Supplementary Conditions or elsewhere in the Contract), the certificates and other evidence of insurance required to be provided by Contractor in accordance with Article 6.
- C. Evidence of Owner's Insurance: After receipt of the executed counterparts of the Agreement and all required bonds and insurance documentation, Owner shall promptly deliver to Contractor, with copies to each named insured and additional insured (as identified in the Supplementary Conditions or otherwise), the certificates and other evidence of insurance required to be provided by Owner under Article 6.
- 2.02 Copies of Documents
- A. Owner shall furnish to Contractor four printed copies of the Contract (including one fully executed counterpart of the Agreement), and one copy in electronic portable document format (PDF). Additional printed copies will be furnished upon request at the cost of reproduction.
- B. Owner shall maintain and safeguard at least one original printed record version of the Contract, including Drawings and Specifications signed and sealed by Engineer and other design professionals. Owner shall make such original printed record version of the Contract available to Contractor for review. Owner may delegate the responsibilities under this provision to Engineer.
- 2.03 Before Starting Construction
- A. Preliminary Schedules: Within 10 days after the Effective Date of the Contract (or as otherwise specifically required by the Contract Documents), Contractor shall submit to Engineer for timely review:
- a preliminary Progress Schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract;
- a preliminary Schedule of Submittals; and

ELCDC<sup>®</sup> C.700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 6 of 65

- a preliminary Schedule of Values for all of the Work which includes quantities and prices of items which when added together equal the Contract Price and subdivides the Work into component parts in sufficient detail to serve as the basis for progress payments during performance of the Work. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work.
- 2.04 Preconstruction Conference; Designation of Authorized Representatives
- A. Before any Work at the Site is started, a conference attended by Owner, Contractor, Engineer, and others as appropriate will be held to establish a working understanding among the parties as to the Work and to discuss the schedules referred to in Paragraph 2.03.4, procedures for handling Shop Drawings, Samples, and other submittals, processing Applications for Payment, electronic or digital transmittals, and maintaining required records.
- B. At this conference Owner and Contractor each shall designate, in writing, a specific individual to act as its authorized representative with respect to the services and responsibilities under the Contract. Such individuals shall have the authority to transmit and receive information, render decisions relative to the Contract, and otherwise act on behalf of each respective party.
- 2.05 Initial Acceptance of Schedules
- A. At least 10 days before submission of the first Application for Payment a conference, attended by Contractor, Engineer, and others as appropriate, will be held to review for acceptability to Engineer as provided below the schedules submitted in accordance with Paragraph 2.03.A. Contractor shall have an additional 10 days to make corrections and adjustments and to complete and resubmit the schedules. No progress payment shall be made to Contractor until acceptable schedules are submitted to Engineer.
- The Progress Schedule will be acceptable to Engineer if it provides an orderly progression of the Work to completion within the Contract Times. Such acceptance will not impose on Engineer responsibility for the Progress Schedule, for sequencing, scheduling, or progress of the Work, nor interfere with or relieve Contractor from Contractor's full responsibility therefor.
- Contractor's Schedule of Submittals will be acceptable to Engineer if it provides a workable arrangement for reviewing and processing the required submittals.
- Contractor's Schedule of Values will be acceptable to Engineer as to form and substance if it provides a reasonable allocation of the Contract Price to the component parts of the Work.
- 2.06 Electronic Transmittals
- A. Except as otherwise stated elsewhere in the Contract, the Owner, Engineer, and Contractor may transmit, and shall accept, Project-related correspondence, text, data, documents, drawings, information, and graphics, including but not limited to Shop Drawings and other submittals, in electronic media or digital format, either directly, or through access to a secure Project website.
- B. If the Contract does not establish protocols for electronic or digital transmittals, then Owner, Engineer, and Contractor shall jointly develop such protocols.
- C. When transmitting items in electronic media or digital format, the transmitting party makes

no representations as to long term compatibility, usability, or readability of the items resulting from the recipient's use of software application packages, operating systems, or ELCO\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract.

ELCDC<sup>4</sup> C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

from those established in applicable transmittal protocols. computer hardware differing from those used in the drafting or transmittal of the items, or

# ARTICLE 3 – DOCUMENTS: INTENT, REQUIREMENTS, REUSE

## 3.01 Intent

- ₽ required by all The Contract Documents are complementary; what is required by one is as binding as if
- œ part thereof) to be constructed in accordance with the Contract Documents. It is the intent of the Contract Documents to describe a functionally complete project (or
- <u></u> Unless otherwise stated in the Contract Documents, if there is a discrepancy between the derived from such electronic or digital versions) and the printed record version, the printed electronic or digital versions of the Contract Documents (including any printed copies record version shall govern.
- Ģ The Contract supersedes prior negotiations, representations, and agreements, whether
- m Engineer will issue clarifications and interpretations of the Contract Documents as provided written or oral
- 3.02 **Reference Standards** herein.
- ₽ Standards Specifications, Codes, Laws and Regulations
- <u>ب</u> Reference in the Contract Documents to standard specifications, manuals, reference standard specification, manual, reference standard, code, or Laws or Regulations in Regulations, whether such reference be specific or by implication, shall mean the standards, or codes of any technical society, organization, or association, or to Laws or were no Bids), except as may be otherwise specifically stated in the Contract effect at the time of opening of Bids (or on the Effective Date of the Contract if there Documents.
- Ņ No provision of any such standard specification, manual, reference standard, or code, or any instruction of a Supplier, shall be effective to change the duties responsibility inconsistent with the provisions of the part of the Contract Documents supervise or direct the performance of the Work or any duty or authority to undertake partners, employees, agents, consultants, or subcontractors, any duty or authority to effective to assign to Owner, Engineer, or any of their officers, directors, members, Documents prepared by or for Engineer. No such provision or instruction shall be consultants, agents, or employees, from those set forth in the part of the Contract responsibilities of Owner, Contractor, or Engineer, or any of their subcontractors, prepared by or for Engineer. q
- 3.03 Reporting and Resolving Discrepancies
- P **Reporting Discrepancies**
- مر Contractor's Verification of Figures and Field Measurements: Before undertaking each and verify pertinent figures and dimensions therein, particularly with respect to part of the Work, Contractor shall carefully study the Contract Documents, and check knowledge of, and shall not proceed with any Work affected thereby until the conflict any conflict, error, ambiguity, or discrepancy that Contractor discovers, or has actual applicable field measurements. Contractor shall promptly report in writing to Engineer

Copyright © 2013 National Society of Professional Engineers, American Co and American Society of Civil Engineers. All rights reserved. ELCDC® C-700 (Rev. 1), Standard General Conditions of the Construction Contract. © 2013 National Society of Professional Engineers, American Council of Engineering Companies Page 8 of 65

> Engineer, or by an amendment or supplement to the Contract Documents issued error, ambiguity, or discrepancy is resolved, by a clarification or interpretation by pursuant to Paragraph 11.01.

- Ν Contractor's Review of Contract Documents: If, before or during the performance of reference standard, or code, or (d) any instruction of any Supplier, then Contractor the Work, Contractor discovers any conflict, error, ambiguity, or discrepancy within the conflict, error, ambiguity, or discrepancy is resolved, by a clarification Work affected thereby (except in an emergency as required by Paragraph 7.15) until shall promptly report it to Engineer in writing. Contractor shall not proceed with the Law or Regulation, (b) actual field conditions, (c) any standard specification, manual, the Contract Documents, or between the Contract Documents and (a) any applicable Documents issued pursuant to Paragraph 11.01. interpretation by Engineer, or by an amendment or supplement to the Contract õ
- ω Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor had actual knowledge thereof
- φ **Resolving Discrepancies:**
- Ŀ Except as may be otherwise specifically stated in the Contract Documents, the such provisions of the Contract Documents and: take precedence in resolving any conflict, error, ambiguity, or discrepancy between provisions of the part of the Contract Documents prepared by or for Engineer shall
- ω the provisions of any standard specification, manual, reference standard, or code, or the instruction of any Supplier (whether or not specifically incorporated by reference as a Contract Document); or
- <u>p</u> the provisions of any Laws or Regulations applicable to the performance of the would result in violation of such Law or Regulation). Work (unless such an interpretation of the provisions of the Contract Documents
- 3.04 Requirements of the Contract Documents
- ₽ During the performance of the Work and until final payment, Contractor and Owner shall Contract Documents, and judge of the acceptability of the Work thereunder. after such matters arise. Engineer will be the initial interpreter of the requirements of the Documents (sometimes referred to as requests for information or interpretation-RFIs), or submit to the Engineer all matters in question concerning the requirements of the Contract relating to the acceptability of the Work under the Contract Documents, as soon as possible
- φ binding on Contractor, unless it appeals by submitting a Change Proposal, and on Owner, Engineer will, with reasonable promptness, render a written clarification, interpretation, or unless it appeals by filing a Claim. Documents. Engineer's written clarification, interpretation, or decision will be final and decision on the issue submitted, or initiate an amendment or supplement to the Contract
- <u></u> If a submitted matter in question concerns terms and conditions of the Contract in question, either party may pursue resolution as provided in Article 12. written notice to Owner and Contractor that Engineer is unable to provide a decision or otherwise), or (3) other engineering or technical matters, then Engineer will promptly give Contract Documents, (2) the design (as set forth in the Drawings, Specifications, or Documents that do not involve (1) the performance or acceptability of the Work under the interpretation. If Owner and Contractor are unable to agree on resolution of such a matter

ELCO: C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineeting Companies, and American Society of Civil Engineers. All rights reserved. Page 9 of 65

J

Reuse of Documents
3.05

- A. Contractor and its Subcontractors and Suppliers shall not:
- have or acquire any title to or ownership rights in any of the Drawings, Specifications, or other documents (or copies of any thereof) prepared by or bearing the seal of Engineer or its consultants, including electronic media editions, or reuse any such Drawings, Specifications, other documents, or copies thereof on extensions of the Project or any other project withhout written consent of Owner and Engineer and specific written verification or adaptation by Engineer, or
- have or acquire any title or ownership rights in any other Contract Documents, reuse any such Contract Documents for any purpose without Owner's express written consent, or violate any copyrights pertaining to such Contract Documents.
  - B. The prohibitions of this Paragraph 3.05 will survive final payment, or termination of the Contract. Nothing herein shall preclude Contractor from retaining copies of the Contract Documents for record purposes.

# ARTICLE 4 – COMMENCEMENT AND PROGRESS OF THE WORK

- 4.01 Commencement of Contract Times; Notice to Proceed
- A. The Contract Times will commence to run on the thirtieth day after the Effective Date of the Contract or, if a Notice to Proceed is given, on the day indicated in the Notice to Proceed. A Notice to Proceed may be given at any time within 30 days after the Effective Date of the Contract. In no event will the Contract Times commence to run later than the sixtieth day after the day of Bid opening or the thirtieth day after the Effective Date of the Contract, whichever date is earlier.
- 4.02 Starting the Work
- A. Contractor shall start to perform the Work on the date when the Contract Times commence to run. No Work shall be done at the Site prior to such date.
  - 4.03 Reference Points
- A. Owner shall provide engineering surveys to establish reference points for construction which in Engineer's judgment are necessary to enable Contractor to proceed with the Work. Contractor shall be responsible for laying out the Work, shall protect and preserve the established reference points and property monuments, and shall make no changes or relocations without the prior written approval of Owner. Contractor shall report to Engineer whenever any reference point or property monument is lost or destroyed or require relocation because of necessary changes in grades or locations, and shall be responsible for the accurate replacement or relocation of such reference points or properly monuments by professionally qualified personnel.
- 4.04 Progress Schedule
- Contractor shall adhere to the Progress Schedule established in accordance with Paragraph 2.05 as it may be adjusted from time to time as provided below.
- Contractor shall submit to Engineer for acceptance (to the extent indicated in Paragraph 2.05) proposed adjustments in the Progress Schedule that will not result in changing the Contract Times.

- Proposed adjustments in the Progress Schedule that will change the Contract Times shall be submitted in accordance with the requirements of Article 11.
- B. Contractor shall carry on the Work and adhere to the Progress Schedule during all disputes or disagreements with Owner. No Work shall be delayed or postponed pending resolution of any disputes or disagreements, or during any appeal process, except as permitted by Paragraph 16.04, or as Owner and Contractor may otherwise agree in writing.
- 4.05 Delays in Contractor's Progress
- A. If Owner, Engineer, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in the Contract Times and Contract Price. Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract Times.
  - B. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for delay, disruption, or interference caused by or within the control of Contractor. Delay, disruption, and interference attributable to and within the control of a Subcontractor or Supplier shall be deemed to be within the control of Contractor.
- C. If Contractor's performance or progress is delayed, disrupted, or interfered with by unanticipated causes not the fault of and beyond the control of Owner, Contractor, and those for which they are responsible, then Contractor shall be entitled to an equitable adjustment in Contract Times. Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract Times. Such an adjustment shall be Contractor's sole and exclusive remedy for the delays, disruption, and interference described in this paragraph. Causes of delay, disruption, or interference that may give rise to an djustment in Contract Times under this paragraph include but are not limited to the following:
  - severe and unavoidable natural catastrophes such as fires, floods, epidemics, and earthquakes;
- abnormal weather conditions;
- acts or failures to act of utility owners (other than those performing other work at or adjacent to the Site by arrangement with the Owner, as contemplated in Article 8); and
- acts of war or terrorism.
- D. Delays, disruption, and interference to the performance or progress of the Work resulting from the existence of a differing subsurface or physical condition, an Underground Facility that was not shown or indicated by the Contract Documents, or not shown or indicated with reasonable accuracy, and those resulting from Hazardous Environmental Conditions, are governed by Article 5.
  - E. Paragraph 8.03 governs delays, disruption, and interference to the performance or progress of the Work resulting from the performance of certain other work at or adjacent to the Site.
- F. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for any delay, disruption, or interference if such delay is concurrent with a delay, disruption, or interference caused by or within the control of Contractor.

EICDC® C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

ELDC\* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Gvil Engineers. All rights reserved.

Ģ Contractor must submit any Change Proposal seeking an adjustment in Contract Price or Contract Times under this paragraph within 30 days of the commencement of the delaying. disrupting, or interfering event.

ARTICLE 5 - AVAILABILITY OF LANDS; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS

# ENVIRONMENTAL CONDITIONS

- 5.01 Availability of Lands P Owner shall furnish the Site. Owner shall notify Contractor of any encumbrances or restrictions not of general application but specifically related to use of the Site with which
- œ Upon reasonable written request, Owner shall furnish Contractor with a current statement are to be made and Owner's interest therein as necessary for giving notice of or filing a of record legal title and legal description of the lands upon which permanent improvements Contractor must comply in performing the Work. mechanic's or construction lien against such lands in accordance with applicable Laws and
- <u></u> Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment Regulations
- 5.02 Use of Site and Other Areas
- Limitation on Use of Site and Other Areas:

P

- Ŀ Contractor shall confine construction equipment, temporary construction facilities, the storage of materials and equipment, and the operations of workers to the Site, unreasonably encumber the Site and such other adjacent areas with construction otherwise, and other adjacent areas permitted by Laws and Regulations, and shall not adjacent areas that Contractor has arranged to use through construction easements or equipment or other materials or equipment. Contractor shall assume full responsibility provided that such damage or injuries result from the performance of the Work or injuries and losses sustained by the owners or occupants of any such land or areas; Contractor's operations; (c) damage to any other adjacent land or areas; and (d) for for (a) damage to the Site; (b) damage to any such other adjacent areas used for responsible from other actions or conduct of the Contractor or those for which Contractor is
- Ņ If a damage or injury claim is made by the owner or occupant of any such land or area the Contractor or those for which Contractor is responsible, Contractor shall (a) take such owner or occupant, or otherwise resolve the claim by arbitration or other dispute (b) promptly attempt to settle the claim as to all parties through negotiations with immediate corrective or remedial action as required by Paragraph 7.12, or otherwise; because of the performance of the Work, or because of other actions or conduct of resolution costs) arising out of or relating to any claim or action, legal or equitable, attorneys, and other professionals and all court or arbitration or other dispute and damages (including but not limited to all fees and charges of engineers, architects each and any of them from and against any such claim, and against all costs, losses, directors, members, partners, employees, agents, consultants and subcontractors of Regulations, indemnify and hold harmless Owner and Engineer, and the officers, resolution proceeding, or at law; and (c) to the fullest extent permitted by Laws and brought by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused directly or indirectly, in whole or in part

Copyright © 2013 National Society of Professional Engineers, American and American Society of Civil Engineers. All rights reserved EJCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. / of Professional Engineers, American Council of Engineering Companies y of Civil Engineers. All rights reserved. Page 12 of 65

12

3

3

لہ ،

ŝ

Ĵ,

j.

1

or conduct of the Contractor or those for which Contractor is responsible by, or based upon, Contractor's performance of the Work, or because of other actions

- Ρ Removal of Debris During Performance of the Work: During the progress of the Work the materials, rubbish, and other debris. Removal and disposal of such waste materials, Contractor shall keep the Site and other adjacent areas free from accumulations of waste rubbish, and other debris shall conform to applicable Laws and Regulations.
- <u>.</u> Contractor shall remove from the Site and adjacent areas all tools, appliances, construction the Work and make it ready for utilization by Owner. At the completion of the Work Cleaning: Prior to Substantial Completion of the Work Contractor shall clean the Site and property not designated for alteration by the Contract Documents. equipment and machinery, and surplus materials and shall restore to original condition all
- Ģ Loading of Structures: Contractor shall not load nor permit any part of any structure to be part of the Work or adjacent structures or land to stresses or pressures that will endanger loaded in any manner that will endanger the structure, nor shall Contractor subject any them.

Subsurface and Physical Conditions

- 5.03 ₽ Reports and Drawings: The Supplementary Conditions identify:
- <u>ب</u> those reports known to Owner of explorations and tests of subsurface conditions at or
- 2 those drawings known to Owner of physical conditions relating to existing surface or subsurface structures at the Site (except Underground Facilities); and adjacent to the Site;
- Technical Data contained in such reports and drawings.

ω

- çα Reliance by Contractor on Technical Data Authorized: Contractor may rely upon upon the accuracy of the Technical Data (as defined in Article 1) contained in any Documents. If no such express identification has been made, then Contractor may rely respect to such reports and drawings, but such reports and drawings are not Contract accuracy of the Technical Data expressly identified in the Supplementary Conditions with partners, employees, agents, consultants, or subcontractors, with respect to: make any claim against Owner or Engineer, or any of their officers, directors, members, geotechnical or environmental report prepared for the Project and made available to Contractor. Except for such reliance on Technical Data, Contractor may not rely upon or
- the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor, and safety precautions and
- Ņ other data, interpretations, opinions, and information contained in such reports or programs incident thereto; or
- shown or indicated in such drawings; or any Contractor interpretation of or conclusion drawn from any Technical Data or any
- such other data, interpretations, opinions, or information

μ

ECDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Gouncil of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 13 of 65

<ul> <li>Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract Times.</li> <li>Contractor shall not be entitled to any adjustment in the Contract Price or Contract</li> </ul>	Times with respect to a subsurface or physical condition if: a. Contractor knew of the existence of such condition at the time Contractor made a commitment to Owner with respect to Contract Price and Contract Times by the submission of a Bid or becoming bound under a negotiated contract, or otherwise; or		c. Contractor railed to give the written notice as required by Paragraph 5.04.A. 3. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, or both, then any such adjustment shall be set forth in a Change Order.	<ol> <li>Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, or both, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the subsurface or physical condition in question.</li> <li>5.05 Underground Facilities</li> <li>A. Contractor's Responsibilities: The information and data shown or indicated in the Contract</li> </ol>	Documents with respect to existing Underground Facilities at or adjacent to the Site is based on information and data furnished to Owner or Engineer by the owners of such Underground Facilities, including Owner, or by others. Unless it is otherwise expressly provided in the Supplementary Conditions: 1. Owner and Engineer do not warrant or guarantee the accuracy or completeness of any such information or data provided by others; and 2. the cost of all of the followine will be included in the Contract Direct and Contracts		<ul> <li>c. coordination of the Work with the owners (including Owner) of such Underground Facilities, during construction; and</li> <li>d. the safety and protection of all existing Underground Facilities at the Site, and repairing any damage thereto resulting from the Work.</li> <li>B. Notice by Contractor if Contractor believes that an Underground Facility that is uncovered OI revealed at the Site was not shown or indicated with reasonable accuracy, then Contractor shown or indicated with reasonable accuracy, then Contractor shall, promptly after</li> </ul>	ELCDC: C-700 (New. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 15 of 65
<ul> <li>5.04 Differing Subsurface or Physical Conditions</li> <li>A. Notice by Contractor: If Contractor believes that any subsurface or physical condition that is uncovered or revealed at the Site either:</li> <li>1. is of such a nature as to establish that any Technical Data on which Contractor is</li> </ul>	entitled to rely as provided in Paragraph 5.03 is materially inaccurate, or is of such a nature as to require a change in the Drawings or Specifications; or differs materially from that shown or indicated in the Contract Documents; or is of an unusual nature, and differs materially from conditions ordinarily encountered	and generally recognized as inherent in work of the character provided for in the Contract Documents; then Contractor shall, promptly after becoming aware thereof and before further disturbing the subsurface or physical conditions or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing about such condition. Contractor shall not further disturb such	condition or perform any Work in connection therewith (except with respect to an emergency) until receipt of a written statement permitting Contractor to do so. B. <i>Engineer's Review</i> : After receipt of written notice as required by the preceding paragraph, Engineer will promptly review the subsurface or physical condition in guestion; determine	the necessity of Owner's obtaining additional exploration or tests with respect to the condition; conclude whether the condition falls within any one or more of the differing site condition categories in Paragraph 5.04.A above; obtain any pertinent cost or schedule information from Contractor; prepare recommendations to Owner regarding the Contractor's resumption of Work in connection with the subsurface or physical condition in question and the need for any change in the Drawings or Specifications; and advise Owner in writing of Engineer's findings, conclusions, and recommendations.	C. Owner's Statement to Contractor Regarding Site Condition: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the subsurface or physical condition in question, addressing the resumption of Work in connection with such condition, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations, in whole or in part.	<ol> <li>Possible Price and Times Adjustments:</li> <li>Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times, or both, to the extent that the existence of a differing subsurface or physical condition, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:</li> </ol>	<ul> <li>a. such condition must fall within any one or more of the categories described in Paragraph 5.04.A;</li> <li>b. with respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03; and,</li> </ul>	ELOC © 4.00 (Fav. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Sciency of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 14 of 65

becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), identify the owner of such Underground Facility and give written notice to that owner and to Owner and Engineer.

- C. Engineer's Review: Engineer will promptly review the Underground Facility and conclude whether such Underground Facility was not shown or indicated in the Contract Documents, or was not shown or indicated with reasonable accuracy; obtain any pertinent cost or schedule information from Contractor; prepare recommendations to Owner regarding the Contractor's resumption of Work in connection with the Underground Facility in question; determine the extent, if any, to which a change is required in the Drawings or Specifications to reflect and document the consequences of the existence or location of the Underground Facility; and advise Owner in writing of Engineer's findings, conclusions, and recommendations. During such time, Contractor shall be responsible for the safety and protection of such Underground Facility.
- D. Owner's Statement to Contractor Regarding Underground Facility: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the Underground Facility in question, addressing the resumption of Work in connection with such Underground Facility, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations in whole or in part.
- E. Possible Price and Times Adjustments:
- 1. Contractor shall be entitled to an equitable adjustment in the Contract Price or Contract Times, or both, to the extent that any existing Underground Facility at the Site that was not shown or indicated in the Contract Documents, or was not shown or indicated with reasonable accuracy, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:
- Contractor did not know of and could not reasonably have been expected to be aware of or to have anticipated the existence or actual location of the Underground Facility in question;
- With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03;
- c. Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract Times; and
- Contractor gave the notice required in Paragraph 5.05.8.

a.

- If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, or both, then any such adjustment shall be set forth in a Change Order.
- 3. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, or both, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the Underground Facility in question.

E(COC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 16 of 65

3)

9

# 5.06 Hazardous Environmental Conditions at Site

- A. Reports and Drawings: The Supplementary Conditions identify:
- those reports and drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site; and
- Technical Data contained in such reports and drawings.

2

- B. Reliance by Contractor on Technical Data Authorized: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely on the accuracy of the Technical Data (as defined in Article 1) contained in any geotechnical or environmental report prepared for the Project and made available to Contractor. Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors with respect to:
- the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences and procedures of construction to be employed by Contractor and safety precautions and programs incident thereto; or
- other data, interpretations, opinions and information contained in such reports or shown or indicated in such drawings; or
- any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions or information.
- C. Contractor shall not be responsible for removing or remediating any Hazardous Environmental Condition encountered, uncovered, or revealed at the Site unless such removal or remediation is expressly identified in the Contract Documents to be within the scope of the Work.
- D. Contractor shall be responsible for controlling, containing, and duly removing all Constituents of Concern brought to the Site by Contractor, Subcontractors, Suppliers, or anyone else for whom Contractor is responsible, and for any associated costs; and for the costs of removing and remediating any Hazardous Environmental Condition created by the presence of any such Constituents of Concern.
- 'n If Contractor encounters, uncovers, or reveals a Hazardous Environmental Condition whose necessity for Owner to retain a qualified expert to evaluate such condition or take such notice in writing). Owner shall promptly consult with Engineer concerning the by Paragraph 7.15); and (3) notify Owner and Engineer (and promptly thereafter confirm with such condition and in any area affected thereby (except in an emergency as required immediately: (1) secure or otherwise isolate such condition; (2) stop all Work in connection responsible creates a Hazardous Environmental Condition, then Contractor shall within the scope of the Work, or if Contractor or anyone for whom Contractor is removal or remediation is not expressly identified in the Contract Documents as being question, then Owner may remove and remediate the Hazardous Environmental Condition, whom Contractor is responsible created the Hazardous Environmental Condition in Contractor the written notice required by Paragraph 5.06.F. If Contractor or anyone for actions as are necessary to permit Owner to timely obtain required permits and provide corrective action, if any. Promptly after consulting with Engineer, Owner shall take such and impose a set-off against payments to account for the associated costs.

EJCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Protessional Engineers, American Council of Engineering Companies and American Society of Civil Engineers. All rights reserved. Page 17 of 65

- F. Contractor shall not resume Work in connection with such Hazardous Environmental Condition or in any affected area until after Owner has obtained any required permits related thereto, and delivered written notice to Contractor either (1) specifying that such condition and any affected area is or has been rendered safe for the resumption of Work, or (2) specifying any special conditions under which such Work may be resumed safely.
- G. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, or both, as a result of such Work stoppage or such special conditions under which Work is agreed to be resumed by Contractor, then within 30 days of Owner's written notice regarding the resumption of Work, Contractor may submit a Change Proposal, or Owner may impose a set-off.
- H. If after receipt of such written notice Contractor does not agree to resume such Work based on a reasonable belief it is unsafe, or does not agree to resume such Work under such special conditions, then Owner may order the portion of the Work that is in the area affected by such condition to be deleted from the Work, following the contractual change procedures in Article 11. Owner may have such deleted portion of the Work performed by Owner's own forces or others in accordance with Article 8.
- I. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them from and against all claims, costs, losses, and other professionals and all court or and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition, provided that such Hazardous Environmental Condition, provided that such Hazardous environmental Condition (1) was not shown or indicated in the Drawings, Specifications, or other Contract Documents, identified as Technical Data entitled to limited to limited within the scope of the Work, and (2) was not created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.1 shall obligate Owner to individual's on midvidual or entity from and against the consequences of that individual's or entity's own negligence.
- 1. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the failure to control, contain, or remove a Constituent of Concern brought to the Site by Contractor or by anyone for whom Contractor is responsible, or to a Hazardous Environmental Condition created by Contractor or by anyone for whom Contractor to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- K. The provisions of Paragraphs 5.03, 5.04, and 5.05 do not apply to the presence of Constituents of Concern or to a Hazardous Environmental Condition uncovered or revealed at the Site.

EICDC® C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineens, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 18 of 65

# **ARTICLE 6 – BONDS AND INSURANCE**

# 6.01 Performance, Payment, and Other Bonds

- A. Contractor shall furnish a performance bond and a payment bond, each in an amount at least equal to the Contract Price, as security for the faithful performance and payment of all of Contractor's obligations under the Contract. These bonds shall remain in effect until one year after the date when final payment becomes due or until completion of the correction period specified in Paragraph 15.08, whichever is later, except as provided otherwise by Laws or Regulations, the Supplementary Conditions, or other specific provisions of the Contract. Contract. Solutions of the Contract. Contractor shall also furnish such other bonds as are required by the Supplementary Conditions or other specific provisions of the Contract.
- B. All bonds shall be in the form prescribed by the Contract except as provided otherwise by Laws or Regulations, and shall be executed by such sureties as are named in "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Circular 570 (as amended and supplemented) by the Financial Management Service, Surety Bond Branch, U.S. Department of the Treasury. A bond signed by an agent or attorney-in-fact must be accompanied know that it individual's authority to bind the surety. The evidence of authority shall show that it is effective on the date the agent or attorney-in-fact signed the accompanying bond.
- C. Contractor shall obtain the required bonds from surety companies that are duly licensed or authorized in the jurisdiction in which the Project is located to issue bonds in the required amounts.
- D. If the surety on a bond furnished by Contractor is declared bankrupt or becomes insolvent, or its right to do business is terminated in any state or jurisdiction where any part of the Project is located, or the surety ceases to meet the requirements above, then Contractor shall promptly notify Owner and Engineer and shall, within 20 days after the event giving rise to such notification, provide another bond and surety, both of which shall comply with the bond and surety requirements above.
- E. If Contractor has failed to obtain a required bond, Owner may exclude the Contractor from the Site and exercise Owner's termination rights under Article 16.
- F. Upon request, Owner shall provide a copy of the payment bond to any Subcontractor, Supplier, or other person or entity claiming to have furnished labor or materials used in the performance of the Work.
- 6.02 Insurance—General Provisions
- A. Owner and Contractor shall obtain and maintain insurance as required in this Article and in the Supplementary Conditions.
- B. All insurance required by the Contract to be purchased and maintained by Owner or Contractor shall be obtained from insurance companies that are duly licensed or authorized, in the state or jurisdiction in which the Project is located, to issue insurance policies for the required limits and coverages. Unless a different standard is indicated in the Supplementary Contract shall have an A.M. Best rating of A-VII or better.
- C. Contractor shall deliver to Owner, with copies to each named insured and additional insured (as identified in this Article, in the Supplementary Conditions, or elsewhere in the Contract), certificates of insurance establishing that Contractor has obtained and is Entry Contract (section as (

ELDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved.

maintaining the policies, coverages, and endorsements required by the Contract. Upon request by Owner or any other insured, Contractor shall also furnish other evidence of such required insurance, including but not limited to copies of policies and endorsements, and documentation of applicable self-insured retentions and deductibles. Contractor may block out (redact) any confidential premium or pricing information contained in any policy or endorsement furnished under this provision.

- D. Owner shall deliver to Contractor, with copies to each named insured and additional insured (as identified in this Article, the Supplementary Conditions, or elsewhere in the Contract), certificates of insurance establishing that Owner has obtained and is maintaining the policies, coverages, and endorsements required of Owner by the Contract (if any). Upon request by Contractor or any other insured, Owner shall also provide other evidence of such required insurance (if any), including but not limited to copies of policies and endorsements, and documentation of applicable self-insured retentions and deductibles. Owner may block out (redact) any confidential premium or pricing information contained in any policy or endorsement furnished under this provision.
- E. Failure of Owner or Contractor to demand such certificates or other evidence of the other party's full compliance with these insurance requirements, or failure of Owner or Contractor to identify a deficiency in compliance from the evidence provided, shall not be construed as a waiver of the other party's obligation to obtain and maintain such insurance.
- F. If either party does not purchase or maintain all of the insurance required of such party by the Contract, such party shall notify the other party in writing of such failure to purchase prior to the start of the Work, or of such failure to maintain prior to any change in the required coverage.
- G. If Contractor has failed to obtain and maintain required insurance, Owner may exclude the Contractor from the Site, impose an appropriate set-off against payment, and exercise Owner's termination rights under Article 16.
- H. Without prejudice to any other right or remedy, if a party has failed to obtain required insurance, the other party may elect to obtain equivalent insurance to protect such other party's interests at the expense of the party who was required to provide such coverage, and the Contract Price shall be adjusted accordingly.
- . Owner does not represent that insurance coverage and limits established in this Contract necessarily will be adequate to protect Contractor or Contractor's interests.
- The insurance and insurance limits required herein shall not be deemed as a limitation on Contractor's liability under the indemnities granted to Owner and other individuals and entities in the Contract.

# 6.03 Contractor's Insurance

- A. Workers' Compensation: Contractor shall purchase and maintain workers' compensation and employer's liability insurance for:
- claims under workers' compensation, disability benefits, and other similar employee benefit acts.
- 2. United States Longshoreman and Harbor Workers' Compensation Act and Jones Act
- craining rounds and the second of the second

ELCDC\* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 20 of 65

3

5

لد \_

لسب

i. A

أس

أهتان

Foreign voluntary worker compensation (if applicable)

4

- B. Commercial General Liability—Claims Covered: Contractor shall purchase and maintain commercial general liability insurance, covering all operations by or on behalf of Contractor, on an occurrence basis, against:
- . claims for damages because of bodily injury, sickness or disease, or death of any person other than Contractor's employees.
- claims for damages insured by reasonably available personal injury liability coverage.
- claims for damages, other than to the Work itself, because of injury to or destruction of tangible property wherever located, including loss of use resulting therefrom.
- C. Commercial General Liability—Form and Content: Contractor's commercial liability policy shall be written on a 1996 (or later) ISO commercial general liability form (occurrence form) and include the following coverages and endorsements:

Products and completed operations coverage:

H

- Such insurance shall be maintained for three years after final payment.
- Contractor shall furnish Owner and each other additional insured (as identified in the Supplementary Conditions or elsewhere in the Contract) evidence of continuation of such insurance at final payment and three years thereafter.
- Blanket contractual liability coverage, to the extent permitted by law, including but not limited to coverage of Contractor's contractual indemnity obligations in Paragraph
- Broad form property damage coverage.
- Broad form property dama
   Severability of interest.
- 5. Underground, explosion, and collapse coverage
- Personal injury coverage

.7 .6

- Additional insured endorsements that include both ongoing operations and products and completed operations coverage through ISO Endorsements CG 20 10 10 01 and CG 20 37 10 01 (together); or CG 20 10 07 04 and CG 20 37 07 04 (together); or their equivalent.
- For design professional additional insureds, ISO Endorsement CG 20 32 07 04, "Additional Insured—Engineers, Architects or Surveyors Not Engaged by the Named Insured" or its equivalent.
- D. Automobile liability: Contractor shall purchase and maintain automobile liability insurance against claims for damages because of bodily injury or death of any person or property damage arising out of the ownership, maintenance, or use of any motor vehicle. The automobile liability policy shall be written on an occurrence basis.
- Umbrella or excess liability: Contractor shall purchase and maintain umbrella or excess liability insurance written over the underlying employer's liability, commercial general liability, and automobile liability insurance described in the paragraphs above. Subject to industry-standard exclusions, the coverage afforded shall follow form as to each and every one of the underlying policies.

'n

F. Contractor's pollution liability insurance: Contractor shall purchase and maintain a policy covering third-party injury and property damage claims, including clean-up costs, as a result

ELCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 21 of 65

	of nollition conditions arising from Contractor's onerations and commissed constraint. This		
		6.04	Owner's Liability Insurance
ы С	Additional insureds: The Contractor's commercial general liability		A. In addition to the insurance required to be provided by Contractor under Paragraph 6.03, Owner, at Owner's option, may purchase and maintain at Owner's expense Owner's own
	umbrenia or excess, and pollution liability policies shall include and list as additional insureds Owner and Engineer, and any individuals or entities identified in the		liability insurance as will protect Owner against claims which may arise from operations under the Contract Documents.
	Supplementary Conditions; include coverage for the respective officers, directors, members narrows and supplementary and subscriptions and supplementary and subscriptions and supplementary and subscriptions and supplementary an		B. Owner's liability policies, if any, operate separately and independently from policies
	members, particles, emphysics, agence, consumance, and succontractors of each and any of all such additional insurveds, and the insurance afforded to these additional insurveds shall provide any conserved for all define concerned shored shored shored.		required to be provided by Contractor, and Contractor cannot rely upon Owner's liability policies for any of Contractor's obligations to the Owner, Engineer, or third parties.
	provide printiary coverage for an claims covered thereby (including as applicable those arising from both ongoing and completed operations) on a non-contributory basis.	6.05	Property Insurance
	Contractor shall obtain all necessary endorsements to support these requirements.		A. Builder's Risk: Unless otherwise provided in the Supplementary Conditions. Fontractor shall
Ξ			
	professional services under this Contract, through a delegation of professional design		in the amount of the full insurable replacement cost thereof (subject to such deductible
	services or outlet wise, then contractor shall be responsible for purchasing and maintaining applicable professional liability insurance. This insurance shall provide protection against		amounts as may be provided in the supplementary Conditions or required by Laws and Regulations), This insurance shall:
	claims arising out of performance of professional design or related services, and caused by		<ol> <li>include the Owner and Contractor as named insureds. and all Subcontractors and any</li> </ol>
	a negligent error, omission, or act for which the insured party is legally liable. It shall be maintained throughout the duration of the Contract and for a miximum of two verse sfee		
	Substantial Completion. If such professional design services are performed by a		such builder's risk policy, as insureds or named insureds. For purposes of the
	Subcontractor, and not by Contractor itself, then the requirements of this paragraph may		remainder of this raragraph 6.05, Paragraphs 6.06 and 6.07, and any corresponding Supplementary Conditions. the parties required to he insured shall collectively he
	be satistied through the purchasing and maintenance of such insurance by such Subcontractor.		referred to as "insureds."
<u></u>			<ol><li>be written on a builder's risk "all risk" policy form that shall at least include insurance for physical loss or damage to the Myork temperatural direct force and meta-side</li></ol>
	1. include at least the specific coverages provided in this Article.		and equipment in transit, and shall insure against at least the following perils or causes
	2. be written for not less than the limits of liability provided in this Article and in the		of loss: fire; lightning; windstorm; riot; civil commotion; terrorism; vehicle impact;
	Supplementary Conditions, or required by Laws or Regulations, whichever is greater.		aircraft; smoke; theft; vandalism and malicious mischief; mechanical breakdown, hoiler explosion and artificially generated electric current: eacthorustee volocity
	3. contain a provision or endorsement that the coverage afforded will not be canceled,		activity, and other earth movement; flood; collapse; explosion; debris removal;
	materially changed, or renewal refused until at least 10 days prior written notice has		demolition occasioned by enforcement of Laws and Regulations; water damage (other
	occur given to contractor. Writing the early of a support of any such written notice, Contractor shall provide a copy of the notice to Owner, Engineer, and each other		trian triat caused by hourd), and such other perils of causes of loss as may be specifically required by the Supplementary Conditions. If insurance against mechanical
	insured under the policy.		breakdown, boiler explosion, and artificially generated electric current; earthquake;
	4. remain in effect at least until final payment (and longer if expressly required in this		volcanic activity, and other earth movement; or flood, are not commercially available
	Article) and at all times thereafter when Contractor may be correcting, removing, or		under builder's risk policies, by endorsement or otherwise, such insurance may be
	replacing defective Work as a warranty or correction obligation, or otherwise, or		
	returning to the Site to conduct other tasks arising from the Contract Documents.		3. cover, as insured property, at least the following: (a) the Work and all materials,
	5. be appropriate for the Work being performed and provide protection from claims that		supplies, machinery, apparatus, equipment, fixtures, and other property of a similar nature that are to be inconversed into or used in the provension. Exhibition
	may arise out of or result from Contractor's performance of the Work and Contractor's		construction, erection, or completion of the Work, including Owner-furnished or
	other obligations under the contract up bocuments, whether it is to be performed by Contractors and Subcontractors of Supplier to the contract of the contractors of		assigned property; (b) spare parts inventory required within the scope of the Contract;
			and (c) temporary works which are not intended to form part of the permanent
	employed by any or them to periodin any or the work, or by anyone for whose acts any of them may be liable.		constructed Work but which are intended to provide working access to the Site, or to
-	The conservation matrix of a constitution of increments which has not burning and increase.		the Work under construction, or which are intended to provide temporary support for
÷	The coverage requirements for specific poincies of insurance must be met by such policies, and not hy reference to excess or timbriella insurance provided in other noticies		the Work under construction, including scatfolding, form work, fences, shoring,
	מומ ווחר הל ובובובוורב נה בצרבא הו מיווהי ביום ווזהת מוירב הוהאומרת זוו הרוובו ההוורבא.		taisework, and temporary structures.

and not by reference to excess or umbrella insurance provided in other policies.

ELCDC\* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 22 of 65

COLUMN DE LE COLUM

All contractions of the

And the second

ς.

- lit les
- nd ble ble ble
- he be
- وہ ב ב ב ב ב ב . מי ב ב ב ב ב ב . falsework, and temporary structures.
- cover expenses incurred in the repair or replacement of any insured property (including but not limited to fees and charges of engineers and architects). 4.

ELCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Cuil Engineers. All rights reserved. Page 23 of 65

'n extend to cover damage or loss to insured property while in temporary storage at the Site or in a storage location outside the Site (but not including property stored at the premises of a manufacturer or Supplier)-

extend to cover damage or loss to insured property while in transit

<u>م</u>

- 7. allow for partial occupation or use of the Work by Owner, such that those portions of the Work that are not yet occupied or used by Owner shall remain covered by the builder's risk insurance.
- allow for the waiver of the insurer's subrogation rights, as set forth below

œ

- <u>و</u> provide primary coverage for all losses and damages caused by the perils or causes of loss covered.
- 10 not include a co-insurance clause.
- 11 include an exception for ensuing losses from physical damage or loss with respect to any defective workmanship, design, or materials exclusions
- 12 include performance/hot testing and start-up.
- 13 be maintained in effect, subject to the provisions herein regarding Substantial Completion and partial occupancy or use of the Work by Owner, until the Work is complete.
- φ Notice of Cancellation or Change: All the policies of insurance (and the certificates or other such written notice, the purchasing policyholder shall provide a copy of the notice to each notice has been given to the purchasing policyholder. Within three days of receipt of any be canceled or materially changed or renewal refused until at least 10 days prior written Paragraph 6.05 will contain a provision or endorsement that the coverage afforded will not evidence thereof) required to be purchased and maintained in accordance with this other insured
- <u>.</u> Deductibles: The purchaser of any required builder's risk or property insurance shall pay for costs not covered because of the application of a policy deductible.
- Þ Partial Occupancy or Use by Owner: If Owner will occupy or use a portion or portions of the Owner (directly, if it is the purchaser of the builder's risk policy, or through Contractor) will Work prior to Substantial Completion of all the Work as provided in Paragraph 15.04, then used by Owner shall remain covered by the builder's risk insurance. come off the builder's risk policy, while those portions of the Work not yet occupied or occupancy; rather, those portions of the Work that are occupied or used by Owner may insurance shall not be canceled or permitted to lapse on account of any such partial use or provide notice of such occupancy or use to the builder's risk insurer. The builder's risk
- Ē Additional Insurance: If Contractor elects to obtain other special insurance to be included in Paragraph 6.05, it may do so at Contractor's expense. or supplement the builder's risk or property insurance policies provided under this
- m Insurance of Other Property: If the express insurance provisions of the Contract do not property item will be responsible for deciding whether to insure it, and if so in what employee of Contractor or a Subcontractor, then the entity or individual owning such equipment, or other personal property owned by Contractor, a Subcontractor, or an require or address the insurance of a property item or interest, such as tools, construction

ELCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies and American Society of Civil Engineers. All rights reserved. Page 24 of 65

10

1

7

2

73

: 13

4

# 6.06 Waiver of Rights

- ₽ damage the insurers will have no rights of recovery against any insureds thereunder, or All policies purchased in accordance with Paragraph 6.05, expressly including the builder's addition, waive all such rights against Engineer, its consultants, all Subcontractors, all individuals or entities identified in the Supplementary Conditions as insureds, and the against each other and the respective officers, directors, members, partners, employees employees, agents, consultants, or subcontractors. Owner and Contractor waive all rights against Engineer or its consultants, or their officers, directors, members, partners, risk policy, shall contain provisions to the effect that in the event of payment of any loss or the proceeds of insurance held by Owner or Contractor as trustee or fiduciary, or otherwise the above waivers shall extend to the rights that any party making such waiver may have to of each and any of them, under such policies for losses and damages so caused. None of officers, directors, members, partners, employees, agents, consultants, and subcontractors covered by such policies and any other property insurance applicable to the Work; and, in damages caused by, arising out of, or resulting from any of the perils or causes of loss agents, consultants, and subcontractors of each and any of them, for all losses payable under any policy so issued. and
- œ and any of them, for: directors, members, partners, employees, agents, consultants and subcontractors of each Owner waives all rights against Contractor, Subcontractors, and Engineer, and the officers,
- ٣ loss due to business interruption, loss of use, or other consequential loss extending beyond direct physical loss or damage to Owner's property or the Work caused by, arising out of, or resulting from fire or other perils whether or not insured by Owner;

and

- 2 loss or damage to the completed Project or part thereof caused by, arising out of, or insurance maintained on the completed Project or part thereof by Owner during resulting from fire or other insured peril or cause of loss covered by any property pursuant to Paragraph 15.03, or after final payment pursuant to Paragraph 15.06. partial occupancy or use pursuant to Paragraph 15.04, after Substantial Completion
- <u>.</u> Any insurance policy maintained by Owner covering any loss, damage or consequential loss members, partners, employees, agents, consultants, or subcontractors of each and any of recovery against Contractor, Subcontractors, or Engineer, or the officers, directors, payment of any such loss, damage, or consequential loss, the insurers will have no rights of referred to in Paragraph 6.06.B shall contain provisions to the effect that in the event of
- P Contractor shall be responsible for assuring that the agreement under which a them. and the officers, directors, members, partners, employees, agents, consultants, and Subcontractor waives all rights against Owner, Contractor, all individuals or entities Subcontractor performs a portion of the Work contains provisions whereby the insurance and any other property insurance applicable to the Work of, relating to, or resulting from any of the perils or causes of loss covered by builder's risk subcontractors of each and any of them, for all losses and damages caused by, arising out identified in the Supplementary Conditions as insureds, the Engineer and its consultants,

# 6.07 Receipt and Application of Property Insurance Proceeds

- ₽ paragraph 6.05 will be adjusted and settled with the named insured that purchased the

EICDC\* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 25 of 65

guarantees required by the Specifications shall expressly run to the benefit of Owner. If required by Engineer, Contractor shall furnish satisfactory evidence (including reports of required tests) as to the source, kind, and quality of materials and equipment.	<ul> <li>C. All materials and equipment shall be stored, applied, installed, connected, erected, protected, used, cleaned, and conditioned in accordance with instructions of the applicable Supplier, except as otherwise may be provided in the Contract Documents.</li> <li>7.04 "Or Equals"</li> </ul>	A. Whenever an item of material or equipment is specified or described in the Contract Documents by using the name of a proprietary item or the name of a particular Supplier, the Contract Price has been based upon Contractor furnishing such item as specified. The	specification or description of such an item is intended to establish the type, function, appearance, and quality required. Unless the specification or description contains or is followed by words reading that no like, equivalent, or "or equal" item is permitted,	contractor may request that togimeer authorize the use of other items of material or equipment, or items from other proposed suppliers under the circumstances described below.	1. If Engineer in its sole discretion determines that an item of material or equinoment	proposed by Contractor is functionally equal to that named and sufficiently similar so that no change in related Work will be required, Engineer shall deem it an "or equal" item. For the purposes of this paragraph, a proposed item of material or equipment will be considered functionally equal to an item so named if:	a. in the exercise of reasonable judgment Engineer determines that:	<ol> <li>it is at least equal in materials of construction, quality, durability, appearance, strength, and design characteristics;</li> </ol>		functioning whole; 3) It has a proven record of performance and availability of responsive service;	4) it is not objectionable to Owner.	b. Contractor certifies that, if approved and incorporated into the Work:	<ol> <li>there will be no increase in cost to the Owner or increase in Contract Times; and</li> </ol>	<ol><li>it will conform substantially to the detailed requirements of the item named in the Contract Documents.</li></ol>	B. Contractor's Expense: Contractor shall provide all data in support of any proposed "or equal" item at Contractor's expense.	C. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each "or-equal" request. Engineer may require Contractor to furnish additional data about the promosed "or-anial" from Engineer will be the orbit induct of acconduction.	Note that we have an equal internation of the property of the	evidenced by an approved Shop Drawing or other written communication. Engineer will be evidenced by an approved Shop Drawing or other written communication. Engineer will advise Contractor in written of an unservice determination.	ELCDC* C.700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 27 of 65
policy. Such named insured shall act as fiduciary for the other insureds, and give notice to such other insureds that adjustment and settlement of a claim is in progress. Any other insured may state its position regarding a claim for insured loss in writing within 15 days after notice of such dam.	B. Proceeds for such insured losses may be made payable by the insurer either jointly to multiple insureds, or to the named insured that purchased the policy in its own right and as fiduciary for other insureds, subject to the requirements of any applicable mortgage clause. A named insured receiving insurance proceeds under the builder's risk and other policies of		C. If no other special agreement is reached, the damaged Work shall be repaired or replaced, the money so received applied on account thereof, and the Work and the cost thereof covered by Change Order, if needed.	5	ğ	A. Contractor shall supervise, inspect, and direct the Work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to perform the Work in accordance with the Contract Documents. Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of construction.		B. At all times during the progress of the Work, Contractor shall assign a competent resident superintendent who shall not be replaced without written notice to Owner and Engineer except under extraordinary circumstances.	7.02 Labor; Working Hours	<ul> <li>Contractor shall provide competent, suitably qualified personnel to survey and lay out the Work and perform construction as required by the Contract Documents. Contractor shall at all times maintain good discipline and order at the Site.</li> </ul>	B. Except as otherwise required for the safety or protection of persons or the Work or property at the Site or adjacent therety and event as otherwise control is the Control	Documents, all Work at the Site shall be performed during regular working hours, Monday	through Fiday. Contractor will not perform. Work on a Saturday, Sunday, or any legal holiday. Contractor may perform Work outside regular working hours or on Saturdays, Sundays, or legal holidays only with Owner's written conceant which will not ho	unreasonably withheid.	A.	essume rull responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and	incuentials necessary for the performance, testing, start up, and completion of the Work, whether or not such items are specifically called for in the Contract Documents.	B. All materials and equipment incorporated into the Work shall be of good quality and new, except as otherwise provided in the Contract Documents. All special warranties and	ELCDC C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Gwil Engineers. All rights reserved. Page 26 of 65

. .
D. Effect of Engineer's Determination: Neither approval nor denial of an "or-equal" request shall result in any change in Contract Price. The Engineer's denial of an "or-equal" request shall be final and binding, and may not be reversed through an appeal under any provision of the Contract Documents.

Treatment as a Substitution Request: If Engineer determines that an item of material or equipment proposed by Contractor does not qualify as an "or-equal" item, Contractor may request that Engineer considered the proposed item as a substitute pursuant to Paragraph 7.05.

m

7.05 Substitutes

₽

- Unless the specification or description of an item of material or equipment required to be furnished under the Contract Documents contains or is followed by words reading that no substitution is permitted, Contractor may request that Engineer authorize the use of other items of material or equipment under the circumstances described below. To the extent possible such requests shall be made before commencement of related construction at the creation.
- Contractor shall submit sufficient information as provided below to allow Engineer to determine if the item of material or equipment proposed is functionally equivalent to that named and an acceptable substitute therefor. Engineer will not accept requests for review of proposed substitute items of material or equipment from anyone other than Contractor.
- The requirements for review by Engineer will be as set forth in Paragraph 7.05.B, as supplemented by the Specifications, and as Engineer may decide is appropriate under the circumstances.
- Contractor shall make written application to Engineer for review of a proposed substitute item of material or equipment that Contractor seeks to furnish or use. The

application:

- a. shall certify that the proposed substitute item will:
   1) perform adequately the functions and achieve the results called for by the
- general design,
- be similar in substance to that specified, and
- be suited to the same use as that specified.
- b. will state:

2)

- the extent, if any, to which the use of the proposed substitute item will necessitate a change in Contract Times,
- 2) whether use of the proposed substitute item in the Work will require a change in any of the Contract Documents (or in the provisions of any other direct contract with Owner for other work on the Project) to adapt the design to the proposed substitute item, and
- 3) whether incorporation or use of the proposed substitute item in connection with the Work is subject to payment of any license fee or royalty.

will identify:

<u></u>

all variations of the proposed substitute item from that specified, and

ELCDC® C.700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineeting Companies and American Society of Civil Engineers. All rights reserved. Page 28 of 65

1

لمدر

لير

2) available engineering, sales, maintenance, repair, and replacement services.

- d. shall contain an itemized estimate of all costs or credits that will result directly or indirectly from use of such substitute item, including but not limited to changes in Contract Price, shared savings, costs of redesign, and claims of other contractors affected by any resulting change.
- B. Engineer's Evoluation and Determination: Engineer will be allowed a reasonable time to evaluate each substitute request, and to obtain comments and direction from Owner. Engineer may require Contractor to furnish additional data about the proposed substitute item. Engineer will be the sole judge of acceptability. No substitute will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an acceptable substitute. Engineer's determination will be evidenced by a Field Order or a proposed Change Order accounting for the substitution itself and all related impacts, including changes in Contract Price or Contract. Times. Engineer will advise Contractor in writing of any negative determination.
- C. Special Guarantee: Owner may require Contractor to furnish at Contractor's expense a special performance guarantee or other surety with respect to any substitute.
- D. Reimbursement of Engineer's Cost: Engineer will record Engineer's costs in evaluating a substitute proposed or submitted by Contractor. Whether or not Engineer approves a substitute so proposed or submitted by Contractor, Contractor shall reimburse Owner for the reasonable charges of Engineer for evaluating each such proposed substitute. Contractor shall also reimburse Owner for the reasonable charges of Engineer for the reasonable charges of Engineer for the reasonable charges of Engineer the reasonable charges of Engineer for the reasonable charges of Engineer for making contractor shall also reimburse Owner for the reasonable charges of Engineer for making changes in the Contract Documents (or in the provisions of any other direct contract with Owner) resulting from the acceptance of each proposed substitute.
- E. Contractor's Expense: Contractor shall provide all data in support of any proposed substitute at Contractor's expense.
- F. Effect of Engineer's Determination: If Engineer approves the substitution request, Contractor shall execute the proposed Change Order and proceed with the substitution. The Engineer's denial of a substitution request shall be final and binding, and may not be reversed through an appeal under any provision of the Contract Documents. Contractor may challenge the scope of reimbursement costs imposed under Paragraph 7.05.D, by timely submittal of a Change Proposal.
- 7.06 Concerning Subcontractors, Suppliers, and Others
- A. Contractor may retain Subcontractors and Suppliers for the performance of parts of the Work. Such Subcontractors and Suppliers must be acceptable to Owner.
- B. Contractor shall retain specific Subcontractors, Suppliers, or other individuals or entities for the performance of designated parts of the Work if required by the Contract to do so.
- Subsequent to the submittal of Contractor's Bid or final negotiation of the terms of the Contract, Owner may not require Contractor to retain any Subcontractor, Supplier, or other individual or entity to furnish or perform any of the Work against which Contractor has

?

reasonable objection. D. Prior to entry into any binding subcontract or purchase order, Contractor shall submit to Owner the identity of the proposed Subcontractor or Supplier (unless Owner has already deemed such proposed Subcontractor or Supplier acceptable, during the bidding process or otherwise). Such proposed Subcontractor or Supplier shall be deemed acceptable to Owner unless Owner raises a substantive, reasonable objection within five days.

ELCDC\* C.700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineens, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 29 of 65

0. Nothing in the Contract Documents:	<ol> <li>shall create for the benefit of any such Subcontractor, Supplier, or other individual or entity any contractual relationship between Owner or Engineer and any such Subcontractor, Supplier, or other individual or entity; nor</li> </ol>	<ol> <li>shall create any obligation on the part of Owner or Engineer to pay or to see to the payment of any money due any such Subcontractor, Supplier, or other individual or entity except as may otherwise be required by Laws and Regulations.</li> </ol>	te	A. Contractor shall pay all license fees and royalties and assume all costs incident to the use in the performance of the Work or the incorporation in the Work of any invention, design, process, product, or device which is the subject of patent rights or copyrights held by others. If a particular invention, design, process, product, or device is specified in the Contract Documents for use in the performance of the Work and if, to the actual knowledge of Documents for use is subject to natent rights or convrishes calling to the performance of the Work and if.	the payment of any license fee or royalty to others, the existence of such rights shall be disclosed by Owner in the Contract Documents. B. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold	harmless Contractor, and its officers, directors, members, partners, employees, agents, consultants, and subcontractors from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professional and all counts or subserve.	out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device specified in the Contract Documents, but not	identified as being subject to payment of any license fee or royalty to others required by patent rights or copyrights. C. To the fullest extent permitted by Laws and Regulations. Contractor shall indemnify and	hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all daims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, actionneys, and other professionals and all court or arbitration or other disoute resolution, costs) arising out of or relations to and accuract	of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device not specified in the Contract Documents.	<ol> <li>Permits</li> <li>A. Unless otherwise provided in the Contract Documents, Contractor shall obtain and pay for all construction permits and licenses. Owner shall assist Contractor, when necessary, in obtaining such memits and licenses. Conservation of Automatic Sciences and a second seco</li></ol>	inspection fees meeting on avoids. Contractory shall pay all governmental charges and inspection fees meeting of the prosecution of the Work which are applicable at the time of the submission of Contractor's Bid (or when Contractor became bound under a negotiated contract). Owner shall pay all charges of utility owners for connections for providing permanent service to the Work	ELCDC* C-700 (Fev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers. American Council of Engineering Companies, and American Society of Guil Engineers. All rights reserved. Page 31 of 65
E. Owner may require the replacement of any Subcontractor, Supplier, or other individual or entity retained by Contractor to perform any part of the Work. Owner also may require	Contractor to retain specific replacements; provided, however, that Contractor hor require a replacement to which Contractor has a reasonable objection. If Contractor has submitted the identity of certain Subcontractors, Suppliers, or other individuals or entities for	acceptance by Owner, and Owner has accepted it (either in writing or by failing to make written objection thereto), then Owner may subsequently revoke the acceptance of any such Subcontractor, Supplier, or other individual or entity so identified solely on the basis of subtransition accounded account of the second	or sustaintive, reasonable objection arter due investigation. Contractor shall submit an acceptable replacement for the rejected Subcontractor, Supplier, or other individual or entity.	F. If Owner requires the replacement of any Subcontractor, Supplier, or other individual or entity retained by Contractor to perform any part of the Work, then Contractor shall be entitled to an adjustment in Contract Price or Contract Times, or both, with respect to the replacement; and Contractor shall initiate a Change Proposal for such adjustment within 30 days of Owner's requirement of replacement.	<ol> <li>No acceptance by Owner of any such Subcontractor, Supplier, or other individual or entity, whether initially or as a replacement, shall constitute a waiver of the right of Owner to the completion of the Work in accordance with the Contract Documents.</li> </ol>	<ol> <li>On a monthly basis Contractor shall submit to Engineer a complete list of all Subcontractors and Suppliers having a direct contract with Contractor, and of all other Subcontractors and Suppliers known to Contractor at the time of submittal.</li> </ol>	Contractor shall be fully responsible to Owner and Engineer for all acts and omissions of the Subcontractors, Suppliers, and other individuals or entities performing or furnishing any of the Work just as Contractor is responsible for Contractor's own acts and omissions.	Contractor shall be solely responsible for scheduling and coordinating the work of Subcontractors, Suppliers, and all other individuals or entities performing or furnishing any of the Work.	<ul> <li>Contractor shall restrict all Subcontractors, Suppliers, and such other individuals or entities performing or furnishing any of the Work from communicating with Engineer or Owner, except through Contractor or in case of an emergency, or as otherwise expressly allowed herein.</li> </ul>	The divisions and sections of the Specifications and the identifications of any Drawings shall not control Contractor in dividing the Work among Subcontractors or Suppliers or delineating the Work to be performed by any specific trade.	<ul> <li>All Work performed for Contractor by a Subcontractor or Supplier shall be pursuant to an appropriate contractual agreement that specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract Documents for the benefit of Owner and Engineer.</li> </ul>	Owner may furnish to any Subcontractor or Supplier, to the extent practicable, information about amounts paid to Contractor on account of Work performed for Contractor by the particular Subcontractor or Supplier.	EICDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 30 of 65

And the second second second second

والمتكرية للمنادمة

ά.

щ.

ы Ю Ξ.

<u>...</u>:

-; -;

**ن**ــ

Š

ż

7.09

Taxes ₽ Contractor shall pay all sales, consumer, use, and other similar taxes required to be paid by Contractor in accordance with the Laws and Regulations of the place of the Project which are applicable during the performance of the Work.

### 7.10 Laws and Regulations

- ₽ Contractor shall give all notices required by and shall comply with all Laws and Regulations monitoring Contractor's compliance with any Laws or Regulations. applicable to the performance of the Work. Except where otherwise expressly required by applicable Laws and Regulations, neither Owner nor Engineer shall be responsible for
- œ If Contractor performs any Work or takes any other action knowing or having reason to limited to all fees and charges of engineers, architects, attorneys, and other professionals directors, members, partners, employees, agents, consultants, and subcontractors of each losses, and shall indemnify and hold harmless Owner and Engineer, and the officers, know that it is contrary to Laws or Regulations, Contractor shall bear all resulting costs and but this shall not relieve Contractor of Contractor's obligations under Paragraph 3.03. the Work described in the Contract Documents is in accordance with Laws and Regulations, such Work or other action. It shall not be Contractor's responsibility to make certain that and all court or arbitration or other dispute resolution costs) arising out of or relating to and any of them from and against all claims, costs, losses, and damages (including but not
- 0 submission of Contractor's Bid (or after the date when Contractor became bound under a Owner or Contractor may give notice to the other party of any changes after the having an effect on procuring permits and on sales, use, value-added, consumption, and performance of the Work, including but not limited to changes in Laws or Regulations negotiated contract) in Laws or Regulations having an effect on the cost or time of the amount or extent, if any, of any adjustment in Contract Price or Contract Times other similar taxes. If Owner and Contractor are unable to agree on entitlement to or on Change Proposal, or Owner may initiate a Claim. resulting from such changes, then within 30 days of such notice Contractor may submit a

### 7.11 **Record Documents**

₽ Contractor shall maintain in a safe place at the Site one printed record copy of all Drawings, available to Engineer for reference. Upon completion of the Work, Contractor shall deliver construction. These record documents, together with all approved Samples, will be record documents in good order and annotate them to show changes made during interpretations and clarifications, and approved Shop Drawings. Contractor shall keep such Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, written these record documents to Engineer.

### 7.12 Safety and Protection

- Þ Contractor shall be solely responsible for initiating, maintaining, and supervising all safety necessary protection to prevent damage, injury, or loss to: Contractor shall take all necessary precautions for the safety of, and shall provide the performance of their work, nor for compliance with applicable safety Laws and Regulations. relieve Subcontractors of their responsibility for the safety of persons or property in the precautions and programs in connection with the Work. Such responsibility does not
- ۲ all persons on the Site or who may be affected by the Work;

ELCDC\* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 Nutional Society of Professional Engineers, American Gouncil of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 32 of 65

1

1.1

Ĵ.

3

all the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site; and

2

- ω other property at the Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, other work in progress, utilities, and Underground Facilities not designated for removal, relocation, or replacement in the course of construction.
- Contractor shall comply with all applicable Laws and Regulations relating to the safety of Contractor shall notify Owner; the owners of adjacent property, Underground Facilities, in the protection, removal, relocation, and replacement of their property or work in to the Site, when prosecution of the Work may affect them, and shall cooperate with them and other utilities; and other contractors and utility owners performing work at or adjacent loss; and shall erect and maintain all necessary safeguards for such safety and protection. persons or property, or to the protection of persons or property from damage, injury, or progress
- 0 any. Contractor shall comply with the applicable requirements of Owner's safety programs, if applicable to the Work. The Supplementary Conditions identify any Owner's safety programs that are
- Þ Contractor shall inform Owner and Engineer of the specific requirements of Contractor's safety program with which Owner's and Engineer's employees and representatives must comply while at the Site.
- i... All damage, injury, or loss to any property referred to in Paragraph 7.12.A.2 or 7.12.A.3 perform any of the Work, or anyone for whose acts any of them may be liable, shall be Supplier, or any other individual or entity directly or indirectly employed by any of them to caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor employed by any of them, or anyone for whose acts any of them may be liable, and not Drawings or Specifications or to the acts or omissions of Owner or Engineer or anyone remedied by Contractor at its expense (except damage or loss attributable to the fault of employed by any of them) Contractor or any Subcontractor, Supplier, or other individual or entity directly or indirectly attributable, directly or indirectly, in whole or in part, to the fault or negligence of
- .... Contractor's duties and responsibilities for safety and protection shall continue until such otherwise expressly provided in connection with Substantial Completion) Contractor in accordance with Paragraph 15.06.B that the Work is acceptable (except as time as all the Work is completed and Engineer has issued a notice to Owner and
- Ģ Contractor's duties and responsibilities for safety and protection shall resume whenever correction obligations, or to conduct other tasks arising from the Contract Documents. Contractor or any Subcontractor or Supplier returns to the Site to fulfill warranty or

### 7.13 Safety Representative

₽ Contractor shall designate a qualified and experienced safety representative at the Site and supervising of safety precautions and programs. whose duties and responsibilities shall be the prevention of accidents and the maintaining

7.14

- Hazard Communication Programs ₽ Contractor shall be responsible for coordinating any exchange of material safety data
- sheets or other hazard communication information required to be made available to or

ECDC<sup>2</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 Nutional Society of Professional Engineers, American Council of Engineering Companies and American Society of Civil Engineers. All rights reserved. Page 33 of 65

لتتاري

provide and to enable Engineer to review the information for the limited purposes required by Paragraph 7.16.D	2. Samples:	<ul> <li>a. Contractor shall submit the number of Samples required in the Specifications.</li> <li>b. Contractor shall clearly identify each Sample as to material, Supplier, pertinent data such as catalog numbers, the use for which intended and other data as Engineer may require to enable Engineer to review the submittal for the limited purposes required by Paragraph 7.16.D.</li> </ul>	<ol> <li>Where a Shop Drawing or Sample is required by the Contract Documents or the Schedule of Submittals, any related Work performed prior to Engineer's review and approval of the pertinent submittal will be at the sole avances and reconscisition, of</li> </ol>	č	C. Uther Submittais. Contractor shall submit other submittals to Engineer in accordance with the accepted Schedule of Submittals, and pursuant to the applicable terms of the Specifications. D. Engineer's Review:		incorporation in the Work, conform to the information given in the Contract Documents and be compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents.	<ol> <li>Engineer's review and approval will not extend to means, methods, techniques, sequences, or procedures of construction or to safety precautions or programs incident thereto.</li> </ol>	<ol> <li>Engineer's review and approval of a separate item as such will not indicate approval of the assembly in which the item functions.</li> </ol>	•• cligners's review and approval or a Shop Drawing or Sample shall not relieve Contractor from responsibility for any variation from the requirements of the Contract Documents unless Contractor has complied with the requirements of Paragraph 7.16.A.3 and Engineer has given written approval of each such variation by specific written notation thereof incorporated in or accompanying the Shop Drawing or Sample. Engineer will document any such approved variation from the requirements of the Contract Documents in a Field Order.	<ol> <li>Engineer's review and approval of a Shop Drawing or Sample shall not relieve Contractor from responsibility for complying with the requirements of Paragraph 7.16.A and B.</li> </ol>	6. Engineer's review and approval of a Shop Drawing or Sample, or of a variation from the requirements of the Contract Documents, shall not, under any circumstances, change the Contract Times or Contract Price, unless such changes are included in a	Change Order. 7. Neither Engineer's receipt, review, acceptance or approval of a Shop Drawing, Sample, or other submittal shall result in such item becoming a Contract Document.	EJCDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 35 of 65
exchanged between or among employers at the Site in accordance with Laws or Regulations.	5 Emergencies	A. In emergencies affecting the safety or protection of persons or the Work or property at the Site or adjacent thereto, Contractor is obligated to act to prevent threatened damage, injury, or loss. Contractor shall give Engineer prompt written notice if Contractor believes that any significant changes in the Work or variations from the Contract Documents have been caused thereby or are required as a result thereof. If Engineer determines that a change in the Contract Documents is required because of the action taken by Contractor in	6	<ul> <li>A. Shop Drawing and Sample Submittal Requirements:</li> <li>1. Before submitting a Shon Drawing or Samule Contractor chall have.</li> </ul>		<ul> <li>b. determined and verified all field measurements, quantities, dimensions, specified performance and design criteria, installation requirements, materials, catalog numbers, and similar information with respect thereto;</li> </ul>	<ul> <li>determined and verified the suitability of all materials and equipment offered with respect to the indicated application, fabrication, shipping, handling, storage, assembly, and installation pertaining to the performance of the Work; and</li> </ul>	d. determined and verified all information relative to Contractor's responsibilities for means, methods, techniques, sequences, and procedures of construction, and safety precautions and programs incident thereto.	<ol> <li>Each submittal shall bear a stamp or specific written certification that Contractor has satisfied Contractor's obligations under the Contract Documents with respect to Contractor's review of that submittal, and that Contractor approves the submitral</li> </ol>	3. With each submittal, Contractor shall give Engineer specific written notice of any variations that the Shop Drawing or Sample may have from the requirements of the Contract Documents. This notice shall be set forth in a written communication separate from the Shop Drawings or Sample submittal; and, in addition, in the case of Shop Drawings by a specific notation made on each Shop Drawing submitted to Engineer for review and approval of each such variation.	B. Submittal Procedures for Shop Drawings and Samples: Contractor shall submit Shop Drawings and Samples to Engineer for review and approval in accordance with the accepted Schedule of Submittals. Each submittal will be identified as Engineer'may require.	<ol> <li>Shop Drawings:         <ol> <li>Contractor shall submit the number of copies required in the Specifications.</li> </ol> </li> </ol>	b. Data shown on the Shop Drawings will be complete with respect to quantities, dimensions, specified performance and design criteria, materials, and similar data to show Engineer the services, materials, and equipment Contractor proposes to	EICDC® C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 34 of 65

7.15

A management of

V. Martine - ----

t

. . . . . . .

Emergencies

7.16

- œ Contractor shall perform the Work in compliance with the requirements and commitments set forth in approved Shop Drawings and Samples, subject to the provisions of Paragraph 7.16.D.4.
- m **Resubmittal Procedures:**
- ч Contractor shall make corrections required by Engineer and shall return the required revisions other than the corrections called for by Engineer on previous submittals. for review and approval. Contractor shall direct specific attention in writing to number of corrected copies of Shop Drawings and submit, as required, new Samples
- Ν Contractor shall furnish required submittals with sufficient information and accuracy will record Engineer's time for reviewing a fourth or subsequent submittal of a Shop to obtain required approval of an item with no more than three submittals. Engineer off against payments due to Contractor to secure reimbursement for such charges. responsible for Engineer's charges to Owner for such time. Owner may impose a set-Drawings, sample, or other item requiring approval, and Contractor shall be
- μ If Contractor requests a change of a previously approved submittal item, Contractor such charges, unless the need for such change is beyond the control of Contractor. may impose a set-off against payments due to Contractor to secure reimbursement for shall be responsible for Engineer's charges to Owner for its review time, and Owner
- 7.17 Contractor's General Warranty and Guarantee
- P Contractor warrants and guarantees to Owner that all Work will be in accordance with the members, partners, employees, agents, consultants, and subcontractors shall be entitled to Contract Documents and will not be defective. Engineer and its officers, directors, rely on Contractor's warranty and guarantee.
- μ Contractor's warranty and guarantee hereunder excludes defects or damage caused by:
- abuse, modification, or improper maintenance or operation by persons other than Contractor, Subcontractors, Suppliers, or any other individual or entity for whom Contractor is responsible; or
- Ņ normal wear and tear under normal usage.
- ņ Documents shall be absolute. None of the following will constitute an acceptance of Work Contractor's obligation to perform and complete the Work in accordance with the Contract obligation to perform the Work in accordance with the Contract Documents: that is not in accordance with the Contract Documents or a release of Contractor's
- observations by Engineer;
- recommendation by Engineer or payment by Owner of any progress or final payment;
- ω the issuance of a certificate of Substantial Completion by Engineer or any payment related thereto by Owner;
- 4 use or occupancy of the Work or any part thereof by Owner;
- ы any review and approval of a Shop Drawing or Sample submittal;
- <u></u>б the issuance of a notice of acceptability by Engineer
- 7. any inspection, test, or approval by others; or
- 00 any correction of defective Work by Owner.

ECCOC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 36 of 65

100

3

Ę

->

- Ģ If the Contract requires the Contractor to accept the assignment of a contract entered into by Owner, then the specific warranties, guarantees, and correction obligations contained in Owner for the Work described in the assigned contract. the assigned contract shall govern with respect to Contractor's performance obligations to
- 7.18 Indemnification
- Þ To the fullest extent permitted by Laws and Regulations, and in addition to any other employees, agents, consultants and subcontractors of each and any of them from and hold harmless Owner and Engineer, and the officers, directors, members, partners, obligations of Contractor under the Contract or otherwise, Contractor shall indemnify and than the Work itself), including the loss of use resulting therefrom but only to the extent injury, sickness, disease, or death, or to injury to or destruction of tangible property (other of the Work, provided that any such claim, cost, loss, or damage is attributable to bodily arbitration or other dispute resolution costs) arising out of or relating to the performance charges of engineers, architects, attorneys, and other professionals and all court or against all claims, costs, losses, and damages (including but not limited to all fees and the Work or anyone for whose acts any of them may be liable. any individual or entity directly or indirectly employed by any of them to perform any of caused by any negligent act or omission of Contractor, any Subcontractor, any Supplier, q
- φ In any and all claims against Owner or Engineer or any of their officers, directors, members, perform any of the Work, or anyone for whose acts any of them may be liable, the any Supplier, or any individual or entity directly or indirectly employed by any of them to survivor or personal representative of such employee) of Contractor, any Subcontractor, partners, employees, agents, consultants, or subcontractors by any employee (or the Contractor or any such Subcontractor, Supplier, or other individual or entity under workers' limitation on the amount or type of damages, compensation, or benefits payable by or for indemnification obligation under Paragraph 7.18.A shall not be limited in any way by any compensation acts, disability benefit acts, or other employee benefit acts.
- the liability of Engineer and Engineer's officers, directors, members, partners, employees, The indemnification obligations of Contractor under Paragraph 7.18.A shall not extend to agents, consultants and subcontractors arising out of:

<u>\_</u>

- ۲ the preparation or approval of, or the failure to prepare or approve maps, Drawings,
- giving directions or instructions, or failing to give them, if that is the primary cause of opinions, reports, surveys, Change Orders, designs, or Specifications; or
- 2 the injury or damage
- 7.19 Delegation of Professional Design Services
- P Contractor will not be required to provide professional design services unless such services provide professional services in violation of applicable Laws and Regulations. methods, techniques, sequences and procedures. Contractor shall not be required to such services are required to carry out Contractor's responsibilities for construction means, are specifically required by the Contract Documents for a portion of the Work or unless
- φ If professional design services or certifications by a design professional related to systems, materials, or equipment are specifically required of Contractor by the Contract Documents, licensed professional, whose signature and seal shall appear on all drawings, calculations, satisfy. Contractor shall cause such services or certifications to be provided by a properly Owner and Engineer will specify all performance and design criteria that such services must specifications, certifications, and other submittals prepared by such professional. Shop

ELCOC<sup>4</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 37 of 65

أند

1

professional, if prepared by others, shall bear such professional's written approval when Drawings and other submittals related to the Work designed or certified by such submitted to Engineer.

- Owner and Engineer shall be entitled to rely upon the adequacy, accuracy, and completeness of the services, certifications, or approvals performed by such design professionals, provided Owner and Engineer have specified to Contractor all performance and design criteria that such services must satisfy. ن
- design drawings will be only for the limited purpose of checking for conformance with Pursuant to this paragraph, Engineer's review and approval of design calculations and performance and design criteria given and the design concept expressed in the Contract Documents. Engineer's review and approval of Shop Drawings and other submittals (except design calculations and design drawings) will be only for the purpose stated in Paragraph 7.16.D.1. ä
- Contractor shall not be responsible for the adequacy of the performance or design criteria specified by Owner or Engineer. ய

## **ARTICLE 8 – OTHER WORK AT THE SITE**

- Other Work 8.01
- In addition to and apart from the Work under the Contract Documents, the Owner may perform other work at or adjacent to the Site. Such other work may be performed by Owner's employees, or through contracts between the Owner and third parties. Owner may also arrange to have third-party utility owners perform work on their utilities and facilities at or adjacent to the Site. ÷
- If Owner performs other work at or adjacent to the Site with Owner's employees, or through contracts for such other work, then Owner shall give Contractor written notice thereof prior to starting any such other work. If Owner has advance information regarding che start of any utility work at or adjacent to the Site, Owner shall provide such information to Contractor. æ.
- Contractor shall afford each other contractor that performs such other work, each utility employees, proper and safe access to the Site, and provide a reasonable opportunity for the introduction and storage of materials and equipment and the execution of such other work. Contractor shall do all cutting, fitting, and patching of the Work that may be required to properly connect or otherwise make its several parts come together and properly integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that Contractor may cut or alter others' work with the written consent of Engineer and the others whose owner performing other work, and Owner, if Owner is performing other work with Owner's work will be affected. ن
- performed by others under this Article 8, Contractor shall inspect such other work and work that render it unavailable or unsuitable for the proper execution and results of If the proper execution or results of any part of Contractor's Work depends upon work promptly report to Engineer in writing any delays, defects, or deficiencies in such other Contractor's Work. Contractor's failure to so report will constitute an acceptance of such other work as fit and proper for integration with Contractor's Work except for latent lefects and deficiencies in such other work. ġ.

### Coordination 8.02

- If Owner intends to contract with others for the performance of other work at or adjacent to the Site, to perform other work at or adjacent to the Site with Owner's employees, or to arrange to have utility owners perform work at or adjacent to the Site, the following will be set forth in the Supplementary Conditions or provided to Contractor prior to the start of any such other work: Ä
- the identity of the individual or entity that will have authority and responsibility for coordination of the activities among the various contractors; ÷
- an itemization of the specific matters to be covered by such authority and responsibility; and ż
- the extent of such authority and responsibilities. 'n
- Unless otherwise provided in the Supplementary Conditions, Owner shall have sole authority and responsibility for such coordination. ഷ്
- Legal Relationships 8.03
- Subcontractors, or delays, disrupts, interferes with, or increases the scope or cost of the performance of the Work, through actions or inaction, then Contractor shall be entitled to If, in the course of performing other work at or adjacent to the Site for Owner, the Owner's employees, any other contractor working for Owner, or any utility owner for whom the Owner is responsible causes damage to the Work or to the property of Contractor or its an equitable adjustment in the Contract Price or the Contract Times, or both. Contractor must submit any Change Proposal seeking an equitable adjustment in the Contract Price or the Contract Times under this paragraph within 30 days of the damaging, delaying, disrupting, or interfering event. The entitlement to, and extent of, any such equitable adjustment shall take into account information (if any) regarding such other work that was provided to Contractor in the Contract Documents prior to the submittal of the Bid or the final negotiation of the terms of the Contract. When applicable, any such equitable alle Contractor's rights against such other contractor or utility owner with respect to the damage, delay, disruption, or interference that is the subject of the adjustment. Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract adjustment in Contract Price shall be conditioned on Contractor assigning to Owner limes. ż
- contractor or utility owner, then Owner may impose a set-off against payments due to Contractor shall take reasonable and customary measures to avoid damaging, delaying, disrupting, or interfering with the work of Owner, any other contractor, or any utility owner performing other work at or adjacent to the Site. If Contractor fails to take such measures and as a result damages, delays, disrupts, or interferes with the work of any such other Contractor, and assign to such other contractor or utility owner the Owner's contractual rights against Contractor with respect to the breach of the obligations set forth in this paragraph. ഷ്
- When Owner is performing other work at or adjacent to the Site with Owner's employees, Contractor shall be liable to Owner for damage to such other work, and for the reasonable direct delay, disruption, and interference costs incurred by Owner as a result of Contractor's failure to take reasonable and customary measures with respect to Owner's other work. In response to such damage, delay, disruption, or interference, Owner may mpose a set-off against payments due to Contractor. ن

EICDC<sup>4</sup> C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Givii Engineers. All rights reserved.

EICDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Givii Engineers. All rights reserved.

D. If Contractor damages, delays, disrupts, or interferes with the work of any other contractor, or any utility owner performing other work at or adjacent to the Site, through Contractor's arising out of Contractor's actions, inactions, or negligence in performance of the Work at or adjacent to the Site is made by any such other contractor or utility owner against or adjacent to the Site is made by any such other contractor or utility owner against claim as to all parties through negotiations with such other contractor or utility owner, or otherwise resolve the claim by arbitration or other dispute resolution proceeding or at law, and (2) indemnify and hold harmless Owner and Engineer, and the officers, director's members, partners, employees, agents, consultants and subcontractors of each and any of including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such damage, delay, disruption, or interference.

# ARTICLE 9 - OWNER'S RESPONSIBILITIES

- 9.01 Communications to Contractor
- A. Except as otherwise provided in these General Conditions, Owner shall issue all communications to Contractor through Engineer.
- 9.02 Replacement of Engineer
- A. Owner may at its discretion appoint an engineer to replace Engineer, provided Contractor makes no reasonable objection to the replacement engineer. The replacement engineer's status under the Contract Documents shall be that of the former Engineer.
- 9.03 Furnish Data
- A. Owner shall promptly furnish the data required of Owner under the Contract Documents.
- 9.04 Pay When Due
- A. Owner shall make payments to Contractor when they are due as provided in the Agreement.
- 9.05 Lands and Easements; Reports, Tests, and Drawings
- A. Owner's duties with respect to providing lands and easements are set forth in Paragraph
- B. Owner's duties with respect to providing engineering surveys to establish reference points are set forth in Paragraph 4.03.
- C. Article 5 refers to Owner's identifying and making available to Contractor copies of reports of explorations and tests of conditions at the Site, and drawings of physical conditions relating to existing surface or subsurface structures at the Site.
- 9.06 Insurance
- A. Owner's responsibilities, if any, with respect to purchasing and maintaining liability and property insurance are set forth in Article 6.
- 9.07 Change Orders
- A. Owner's responsibilities with respect to Change Orders are set forth in Article 11.
- EJCDC® C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 40 of 65

23

3

3

j.

لمن

1

1

Inspections, Tests, and Approvals

9.08

- A. Owner's responsibility with respect to certain inspections, tests, and approvals is set forth in Paragraph 14.02.8.
- 9.09 Limitations on Owner's Responsibilities
- A. The Owner shall not supervise, direct, or have control or authority over, nor be responsible for, Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Owner will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- 9.10 Undisclosed Hazardous Environmental Condition
- A. Owner's responsibility in respect to an undisclosed Hazardous Environmental Condition is set forth in Paragraph 5.06.
- 9.11 Evidence of Financial Arrangements
- A. Upon request of Contractor, Owner shall furnish Contractor reasonable evidence that financial arrangements have been made to satisfy Owner's obligations under the Contract Documents (including obligations under proposed changes in the Work).
- 9.12 Safety Programs
- A. While at the Site, Owner's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Owner has been informed.
- Owner shall furnish copies of any applicable Owner safety programs to Contractor

# ARTICLE 10 - ENGINEER'S STATUS DURING CONSTRUCTION

φ

- 10.01 Owner's Representative
- A. Engineer will be Owner's representative during the construction period. The duties and responsibilities and the limitations of authority of Engineer as Owner's representative during construction are set forth in the Contract.
- 10.02 Visits to Site
- A. Engineer will make visits to the Site at intervals appropriate to the various stages of construction as Engineer deems necessary in order to observe as an experienced and qualified design professional the progress that has been made and the quality of the various aspects of Contractor's executed Work. Based on information obtained during such visits and observations, Engineer, for the benefit of Owner, will determine, in general, if the Work is proceeding in accordance with the Contract Documents. Engineer will not be required to make exhaustive or continuous inspections on the Site to check the quality or quantity of the Work. Engineer's efforts will be directed toward providing for Owner a greater degree of confidence that the completed Work will conform generally to the Contract Documents. On the basis of such visits and observations, Engineer will keep Owner informed of the progress of the Work and will endeavor to guard Owner against
- B. Engineer's visits and observations are subject to all the limitations on Engineer's authority and responsibility set forth in Paragraph 10.08. Particularly, but without limitation, during

defective Work.

EICDC\* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 41 of 65

or as a result of Engineer's visits or observations of Contractor's Work, Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work.

### 10.03 Project Representative

A. If Owner and Engineer have agreed that Engineer will furnish a Resident Project Representative to represent Engineer at the Site and assist Engineer in observing the progress and quality of the Work, then the authority and responsibilities of any such Resident Project Representative will be as provided in the Supplementary Conditions, and limitations on the responsibilities thereof will be as provided in Paragraph 10.08. If Owner designates another representative or agent to represent Owner at the Site who is not Engineer's consultant, agent, or employee, the responsibilities and authority and limitations thereon of such other individual or entity will be as provided in the Supplementary Conditions.

### 10.04 Rejecting Defective Work

Engineer has the authority to reject Work in accordance with Article 14.

# 10.05 Shop Drawings, Change Orders and Payments

- A. Engineer's authority, and limitations thereof, as to Shop Drawings and Samples, are set forth in Paragraph 7.16.
- B. Engineer's authority, and limitations thereof, as to design calculations and design drawings submitted in response to a delegation of professional design services, if any, are set forth in Paragraph 7.19.
- C. Engineer's authority as to Change Orders is set forth in Article 11.
- D. Engineer's authority as to Applications for Payment is set forth in Article 15.

# 10.06 Determinations for Unit Price Work

- Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor as set forth in Paragraph 13.03.
- 10.07 Decisions on Requirements of Contract Documents and Acceptability of Work
- A. Engineer will render decisions regarding the requirements of the Contract Documents, and judge the acceptability of the Work, pursuant to the specific procedures set forth herein for initial interpretations, Change Proposals, and acceptance of the Work. In rendering such decisions and judgments, Engineer will not show partiality to Owner or Contractor, and will not be liable to Owner, Contractor, or others in connection with any proceedings, interpretations, decisions, or judgments conducted or rendered in good faith.

# 10.08 Limitations on Engineer's Authority and Responsibilities

A. Neither Engineer's authority or responsibility under this Article 10 or under any other provision of the Contract, nor any decision made by Engineer in good faith either to exercise or not exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by Engineer, shall create, impose, or give rise to any duty in contract, tort, or otherwise owed by Engineer to Contractor, any Subcontractor, any Supplier, any other individual or entity, or to any survety for or employee or agent of any of them.

- B. Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Engineer will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- C. Engineer will not be responsible for the acts or omissions of Contractor or of any Subcontractor, any Supplier, or of any other individual or entity performing any of the Work.
- D. Engineer's review of the final Application for Payment and accompanying documentation and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection, tests and approvals, and other documentation required to be delivered by Paragraph 15.06.A will only be to determine generally that their content complies with the requirements of, and in the case of certificates of inspections, tests, and approvals, that the results certified indicate compliance with the Contract Documents.
  - E. The limitations upon authority and responsibility set forth in this Paragraph 10.08 shall also apply to the Resident Project Representative, if any.

## 10.09 Compliance with Safety Program

A. While at the Site, Engineer's employees and representatives will comply with the specific applicable requirements of Owner's and Contractor's safety programs (if any) of which Engineer has been informed.

# ARTICLE 11 – AMENDING THE CONTRACT DOCUMENTS; CHANGES IN THE WORK

# 11.01 Amending and Supplementing Contract Documents

- A. The Contract Documents may be amended or supplemented by a Change Order, a Work Change Directive, or a Field Order.
- Change Orders:
- a. If an amendment or supplement to the Contract Documents includes a change in the Contract Price or the Contract Times, such amendment or supplement must be set forth in a Change Order. A Change Order also may be used to establish amendments and supplements of the Contract Documents that do not affect the Contract Price or Contract Times.
- b. Owner and Contractor may amend those terms and conditions of the Contract Documents that do not involve (1) the performance or acceptability of the Work, (2) the design (as set forth in the Drawings, Specifications, or otherweise), or (3) other engineering or technical matters, without the recommendation of the Engineer. Such an amendment shall be set forth in a Change Order.
- 2. Work Change Directives: A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the modification ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order, following negotiations by the parties as to the Work Change Directive's effect, if any, on the Contract Price and Contract Times; or, if negotiations are unsuccessful, by a determination under the terms of the Contract

Documents governing adjustments, expressly including Paragraph 11.04 regarding change of Contract Price. Contractor must submit any Change Proposal seeking an

ELOC<sup>4</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Cuil Engineers. All rights reserved.

ELDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Coundi of Engineering Companies, and American Society of Civil Engineers, All rights reserved. Page 42 of 65

adjustment of the Contract Price or the Contract Times, or both, no later than 30 days after the completion of the Work set out in the Work Change Directive. Owner must submit any Claim seeking an adjustment of the Contract Price or the Contract Times, or both, no later than 60 days after issuance of the Work Change Directive.

3. Field Orders: Engineer may authorize minor changes in the Work if the changes do not involve an adjustment in the Contract Price or the Contract Times and are compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. Such changes will be accomplished by a Field Order and will be binding on Owner and also on Contractor, which shall perform the Work involved promptly. If Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, or both, then before proceeding with the Work at issue, Contractor shall submit a Change Proposal as provided herein.

11.02 Owner-Authorized Changes in the Work

A. Without invalidating the Contract and without notice to any surety, Owner may, at any time or from time to time, order additions, deletions, or revisions in the Work. Such changes shall be supported by Engineer's recommendation, to the extent the change involves the design (as set forth in the Drawings, Specifications, or otherwise), or other Owner and Contractor have agreed as to the effect, if any, of the change on Contract Times or .Contract Price; or by a Work Change Directive. Upon receipt of any such deletion in the Work, promptly proceed with the Work involved; or, in the case of a document, Contractor shall promptly proceed with the Work involved; or, in the case of a document. Nothing in this paragraph shall obligate Contractor to undertake work contract or reasonably concludes cannot be performed in a manner consistent with that Contractor's safety obligations under the Contract Documents or Laws and Regulations.

11.03 Unauthorized Changes in the Work

A. Contractor shall not be entitled to an increase in the Contract Price or an extension of the Contract Times with respect to any work performed that is not required by the Contract Documents, as amended, modified, or supplemented, except in the case of an emergency as provided in Paragraph 7.15 or in the case of uncovering Work as provided in Paragraph 14.05.

11.04 Change of Contract Price

- A. The Contract Price may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Price shall comply with the provisions of Paragraph 11.06. Any Claim for an adjustment of Contract Price shall comply with the provisions of Article 12.
- An adjustment in the Contract Price will be determined as follows:

œ

- where the Work involved is covered by unit prices contained in the Contract Documents, then by application of such unit prices to the quantities of the items
- involved (subject to the provisions of Paragraph 13.03); or 2. where the Work involved is not covered by unit prices contained in the Contract Documents, then by a mutually agreed lump sum (which may include an allowance for
- Documents, then by a mutually agreed unity sum (where a sum of the second and profit not necessarily in accordance with Paragraph 11.04.C.2); or overhead and profit not necessarily in accordance with prices contained in the Contract
- 3. where the Work involved is not covered by unit prices contained in the Contract Documents and the parties do not reach mutual agreement to a lump sum, then on

ELCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies and American Society of Civil Engineers. All rights reserved. Page 44 of 65

لحبت

1

the basis of the Cost of the Work (determined as provided in Paragraph 13.01) plus a Contractor's fee for overhead and profit (determined as provided in Paragraph 11.04.C).

C. Contractor's Fee: When applicable, the Contractor's fee for overhead and profit shall be determined as follows:

a mutually acceptable fixed fee; or

Ņ

- if a fixed fee is not agreed upon, then a fee based on the following percentages of the
- various portions of the Cost of the Work: a. for costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2, the Contractor's fee
- shall be 15 percent; b. for costs incurred under Paragraph 13.01.8.3, the Contractor's fee shall be five
- e. where one or more tiers of subcontracts are on the basis of Cost of the Work plus a fee and no fixed fee is agreed upon, the intent of Paragraphs 11.04.C.2.a and 11.04.C.2.b is that the Contractor's fee shall be based on: (1) a fee of 15 percent of the costs incurred under Paragraphs 13.01.A.1 and 13.01.A.2 by the Subcontractor that actually performs the Work, at whatever tier, and (2) with respect to Contractor itself and to any Subcontractors of a tier higher than that of the Subcontractor that actually performs the Work, a fee of five percent of the amount (fee plus underlying costs incurred) attributable to the next lower tier Subcontractor; provided, however, that for any such subcontracted work the maximum total fee to be paid by Owner shall be no greater than 27 percent of the costs incurred by the Subcontractor that actually performs the work;
- no fee shall be payable on the basis of costs itemized under Paragraphs 13.01.8.4, 13.01.8.5, and 13.01.C;
- the amount of credit to be allowed by Contractor to Owner for any change which results in a net decrease in cost will be the amount of the actual net decrease in cost plus a deduction in Contractor's fee by an amount equal to five percent of such net decrease; and
- when both additions and credits are involved in any one change, the adjustment in Contractor's fee shall be computed on the basis of the net change in accordance with Paragraphs 11.04.C.2.a through 11.04.C.2.e, inclusive.

11.05 Change of Contract Times

- A. The Contract Times may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Times shall comply with the provisions of Paragraph 11.06. Any Claim for an adjustment in the Contract Times shall comply with the provisions of Article
- 12. An adjustment of the Contract Times shall be subject to the limitations set forth in
- An adjustment of the contractor's progress. Paragraph 4.05, concerning delays in Contractor's progress.

11.06 Change Proposals

. Φ

A. Contractor shall submit a Change Proposal to Engineer to request an adjustment in the Contract Times or Contract Price; appeal an initial decision by Engineer concerning the requirements of the Contract Documents or relating to the acceptability of the Work under the Contract Documents; contest a set-off against payment due; or seek other relief under the Contract Documents; contest a set-off against payment due; or seek other relief under

ELCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Guil Engineers. All rights reserved. Page 45 of 65

the Contract. The Change Proposal shall specify any proposed change in Contract Times or Contract Price, or both, or other proposed relief, and explain the reason for the proposed	B. If Owner or Contractor refuses to execute a Change Order that is required to be executed
change, with citations to any governing or applicable provisions of the Contract Documents.	under the terms of this raragraph 11.07, it shall be deemed to be of full force and effect, as if fully executed.
1. Procedures: Contractor shall submit each Change Proposal to Engineer promptly (but	11.08 Notification to Surety
in no event later than 30 days) after the start of the event giving rise thereto, or after such initial decision. The Contractor shall submit supporting dara including the	A. If the provisions of any bond require notice to be given to a surety of any change affecting
proposed change in Contract Price or Contract Time (if any), to the Engineer and	the general scope of the Work or the provisions of the Contract Documents (including, but
Owner within 15 days after the submittal of the Change Proposal. The supporting data	tiou liittited to, contract Frice of contract limes), the giving of any such notice will be Contractor's responsibility. The amount of each anninghed hand will he adjusted to adjusted to adjusted
arian be accompanied by a written statement that the supporting data are accurate and complete, and that any requested time or price adjustment is the entire	the effect of any such change.
adjustment to which Contractor believes it is entitled as a result of said event.	ARTICLE 12 – CLAIMS
comments or response from Owner regarding the Change Proposal, and consider any comments or response from Owner regarding the Change Proposal.	10 Claims
2. Engineer's Action: Engineer will review each Change Proposal and, within 30 days after	
receipt of the Contractor's supporting data, either deny the Change Proposal in whole, approve it in whole. or deny it in part and approve it in part Such actions shall he in	
writing, with a copy provided to Owner and Contractor. If Engineer does not take	<ol> <li>Appeals by Owner or Contractor of Engineer's decisions regarding Change Proposals:</li> </ol>
action on the Change Proposal within 30 days, then either Owner or Contractor may at	2. Owner demands for adjustments in the Contract Price or Contract Times or other
any une thereatter submit a letter to the other party indicating that as a result of Engineer's inaction the Change Drongel is downed during the environment of	relief under the Contract Documents; and
time for appeal of the denial under Article 12.	3. Disputes that Engineer has been unable to address because they do not involve the
3. Binding Decision: Engineer's decision will be final and binding upon Owner and	design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of
Article 12.	b. submitted of Lucini: Ine party submitting a Claim shall deliver it directly to the other party to the Contrast account of the issue of the state of the sta
B. Resolution of Certain Change Proposals: If the Change Proposal does not involve the design	to the contract prompty jout in no event later than 30 days) after the start of the event giving rise therete: in the case of anneals reparding Change Provocals within 30 days of the
(as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or	decision under appeal. The party submitting the Claim shall also furnish a copy to the
outer engineering of technical matters, then engineer will notify the parties that the Engineer is unable to resolve the Change Pronocal For numbers of further resolution of	Engineer, for its information only. The responsibility to substantiate a Claim shall rest with
such a Change Proposity such notice shall be deemed a denial, and Contractor may choose to cash recolution underthe Armene of Article 10, 20	the party making the Claim. In the case of a Claim by Contractor seeking an increase in the Contract Times or Contract Price, or both, Contractor shall certify that the Claim is made in
11 07 Evolution of Channel Cutors	good faith, that the supporting data are accurate and complete, and that to the best of
	contractor s knowredge and bener the amount of time of money requested accurately reflects the full amount to which Contractor is entitled.
the changes in the Contract Drive or Contract Timor which are several to the	C. Review and Resolution: The party receiving a Claim shall review it thoroughly, giving full
	consideration to its merits. The two parties shall seek to resolve the Claim through the
accordance with a Work Change Directive;	exchange of information and direct negotiations. The parties may extend the time for resolving the Claim by mutual acreament all actions taken on a chaine deal tha matted is
2. changes in Contract Price resulting from an Owner set-off, unless Contractor has duly	writing and submitted to the other party, with a copy to Engineer.
	D. Mediation:
<ol> <li>changes in the Work which are: (a) ordered by Owner pursuant to Paragraph 11.02, (b) required because of Owner's acceptance of defective Work under Paragraph 14.04 or Owner's correction of defective Work under Paragraph 14.07, or (c) aereed to by the</li> </ol>	<ol> <li>At any time after initiation of a Claim, Owner and Contractor may mutually agree to mediation of the underlying dispute. The agreement to mediate shall stay the Claim</li> </ol>
parties, subject to the need for Engineer's recommendation if the change in the Work	
involves the design (as set forth in the Drawings, Specifications, or otherwise), or other provinged and state states and states an	<ol> <li>In Uwite and contractor agree to mediation, that after bud days from such agreement, aither Owner or Constructor multitheration, torarized the advector of the aither Owner or Constructor multitheration.</li> </ol>

Constant of

changes in the Contract Price or Contract Times, or other changes, which embody the substance of any final and binding results under Paragraph 11.06, or Article 12. 4

engineering or technical matters; and

ELCDC\* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 46 of 65

ELCDC® C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright @ 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 47 of 65

either Owner or Contractor may unilaterally terminate the mediation process, and the Claim submittal and decision process shall resume as of the date of the termination. If the mediation proceeds but is unsuccessful in resolving the dispute, the Claim

submittal and decision process shall resume as of the date of the conclusion of the mediation, as determined by the mediator.

- Owner and Contractor shall each pay one-half of the mediator's fees and costs.
- *Partial Approval*: If the party receiving a Claim approves the Claim in part and denies it in part, such action shall be final and binding unless within 30 days of such action the other party invokes the procedure set forth in Article 17 for final resolution of disputes.

ίщ

ω

- F. Denial of Claim: If efforts to resolve a Claim are not successful, the party receiving the Claim may deny it by giving written notice of denial to the other party. If the receiving party does not take action on the Claim within 90 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of the inaction, the Claim is deemed denied, thereby commencing the time for appeal of the denial. A denial of the Claim shall be final and binding unless within 30 days of the denial the other party invokes the procedure set forth in Article 17 for the final resolution of disputes.
- G. Final and Binding Results: If the parties reach a mutual agreement regarding a Claim, whether through approval of the Claim, direct negotiations, mediation, or otherwise; or if a Claim is approved in part and denied in part, or denied in full, and such actions become final and binding; then the results of the agreement or action on the Claim shall be incorporated in a Change Order to the extent they affect the Contract, including the Work, the Contract Times, or the Contract Price.

# ARTICLE 13 - COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

### 13.01 Cost of the Work

- A. Purposes for Determination of Cost of the Work: The term Cost of the Work means the sum of all costs necessary for the proper performance of the Work at issue, as further defined below. The provisions of this Paragraph 13.01 are used for two distinct purposes:
- To determine Cost of the Work when Cost of the Work is a component of the Contract Price, under cost-plus-fee, time-and-materials, or other cost-based terms; or
- 2. To determine the value of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price. When the value of any such adjustment is determined on the basis of Cost of the Work, Contractor is entitled only to those additional or incremental costs required because of the change in the Work or because of the event giving rise to the adjustment.
- B. Costs Included: Except as otherwise may be agreed to in writing by Owner, costs included in the Cost of the Work shall be in amounts no higher than those prevailing in the locality of the Project, shall not include any of the costs itemized in Paragraph 13.01.C, and shall include only the following items:
- 1. Payroll costs for employees in the direct employ of Contractor in the performance of the Work under schedules of job classifications agreed upon by Owner and Contractor. Such employees shall include, without limitation, superintendents, foremen, and other personnel employed full time on the Work. Payroll costs for employees not employed full time on the Work shall be apportioned on the basis of their time spent on the Work. Payroll costs shall include, but not be limited to, salaries and wages plus the cost of fringe benefits, which shall include social security contributions, unemployment, excise, and payroll taxes, workers' compensation, health and retirement benefits, bonuses, sick leave, and vacation and holiday pay applicable

EICDC<sup>4</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 48 of 65

9

ليه .

thereto. The expenses of performing Work outside of regular working hours, on Saturday, Sunday, or legal holidays, shall be included in the above to the extent authorized by Owner.

- 2. Cost of all materials and equipment furnished and incorporated in the Work, including costs of transportation and storage thereof, and Suppliers' field services required in connection therewith. All cash discounts shall accrue to Contractor unless Owner deposits funds with Contractor with which to make payments, in which case the cash discounts shall accrue to Owner. All trade discounts, rebates, and refunds and returns from sale of surplus materials and equipment shall accrue to Owner, and Contractor shall make provisions so that they may be obtained.
- 3. Payments made by Contractor to Subcontractors for Work performed by Subcontractors. If required by Owner, Contractor shall obtain competitive bids from subcontractors acceptable to Owner and Contractor and shall deliver such bids to Owner, who will then determine, with the advice of Engineer, which bids, if any, will be acceptable. If any subcontract provides that the Subcontractor is to be paid on the basis of Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee shall be determined in the same manner as Contractor's Cost of the Work and fee as provided in this Paragraph 13.01.
- 4. Costs of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys, and accountants) employed for services specifically related to the Work.
- Supplemental costs including the following:

'n

- a. The proportion of necessary transportation, travel, and subsistence expenses of Contractor's employees incurred in discharge of duties connected with the Work.
- b. Cost, including transportation and maintenance, of all materials, supplies, equipment, machinery, appliances, office, and temporary facilities at the Site, and hand tools not owned by the workers, which are consumed in the performance of the Work, and cost, less market value, of such items used but not consumed which remain the property of Contractor.
- c. Rentals of all construction equipment and machinery, and the parts thereof, whether rented from Contractor or others in accordance with rental agreements approved by Owner with the advice of Engineer, and the costs of transportation, loading, unloading, assembly, dismantling, and removal thereof. All such costs shall be in accordance with the terms of said rental agreements. The rental of any such equipment, machinery, or parts shall cease when the use thereof is no longer necessary for the Work.
- Sales, consumer, use, and other similar taxes related to the Work, and for which

ġ

- Contractor is liable, as imposed by Laws and Regulations. Deposits lost for causes other than negligence of Contractor, any Subcontractor,
- e. Deposits lost for causes other than negligence or contractory, any source or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, and royalty payments and fees for permits and licenses.
- f. Losses and damages (and related expenses) caused by damage to the Work, not compensated by insurance or otherwise, sustained by Contractor in connection with the performance of the Work (except losses and damages within the deductible amounts of property insurance established in accordance with Paragraph 6.05), provided such losses and damages have resulted from causes EUCC<sup>2</sup> C.700 (Rev. 1), Standard General Conditions of the Construction Contract.

Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 49 of 65

B. Cash Allowances: Contractor agrees that:	<ol> <li>the cash allowances include the cost to Contractor (less any applicable trade discounts) of materials and equipment required by the allowances to be delivered at the Site, and all applicable taxes; and</li> </ol>	<ol> <li>Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit, and other expenses contemplated for the cash allowances have been included in the Contract Price and not in the allowances, and no demand for additional payment on account of any of the foregoing will be valid.</li> </ol>	C. Contingency Allowance: Contractor agrees that a contingency allowance, if any, is for the sole use of Owner to cover unanticipated costs.	D. Prior to final payment, an appropriate Change Order will be issued as recommended by Engineer to reflect actual amounts due Contractor on account of Work covered by allowances, and the Contract Price shall be correspondingly adjusted. 13.03 Unit Price Work	A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement.	B. The estimated quantities of items of Unit Price Work are not guaranteed and are solely for the purpose of comparison of Bids and determining an initial Contract Price. Payments to Contractor for Unit Price Work will be based on actual quantities.	C. Each unit price will be deemed to include an amount considered by Contractor to be adequate to cover Contractor's overhead and profit for each separately identified item.	D. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor. Engineer will review with Contractor the Engineer's preliminary determinations on such matters before rendering a written decision thereon (by recommendation of an Application for Payment or otherwise). Engineer's written decision recommendation of an Application for Payment or otherwise). Engineer's written decision	urereon will be tinal and binding (except as modified by Engineer to reflect changed factual conditions or more accurate data) upon Owner and Contractor, subject to the provisions of the following paragraph.	E. Within 30 days of Engineer's written decision under the preceding paragraph, Contractor may submit a Change Proposal, or Owner may file a Claim, seeking an adjustment in the Contract Price if:	<ol> <li>the quantity of any item of Unit Price Work performed by Contractor differs materially and significantly from the estimated quantity of such item indicated in the Agreement;</li> </ol>	2. there is no corresponding adjustment with respect to any other item of Work; and	<ol><li>Contractor believes that it is entitled to an increase in Contract Price as a result of having incurred additional expense or Owner believes that Owner is entitled to a</li></ol>	decrease in Contract Price, and the parties are unable to agree as to the amount of any		ELCDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companiles, and American Society of Civil Engineers. All rights reserved. Page 51 of 65
other than the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose other and of them and the listed	Such losses shall include settlements made with the written consent and approval of Owner. No such losses, damages, and expenses shall be included in the Cost of the Work for the purpose of determining Contractor's fee.	<ul> <li>The cost of utilities, fuel, and sanitary facilities at the Site.</li> <li>Minor expenses such as communication service at the Site, express and courier services, and similar petty cash items in connection with the Work.</li> </ul>	i. The costs of premiums for all bonds and insurance that Contractor is required by the Contract Documents to purchase and maintain.	<ul> <li>C. Costs Excluded: The term Cost of the Work shall not include any of the following items:</li> <li>1. Payroll costs and other compensation of Contractor's officers, executives, principals (of partnerships and sole proprietorships), general managers, safety managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and</li> </ul>	contracting agents, expediters, timekeepers, clerks, and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office for general administration of the Work and not specifically included in the agreed upon schedule of Job classifications referent to in Paragraph 13.01.B.1 or specifically covered by Paragraph 13.01.B.4. The payroll costs and other compensation excluded here are to	be considered administrative costs covered by the Contractor's fee. 2. Expenses of Contractor's principal and branch offices other than Contractor's office at the Site.	<ol><li>Any part of Contractor's capital expenses, including interest on Contractor's capital employed for the Work and charges against Contractor for delinquent payments.</li></ol>	4. Costs due to the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, including but not limited to, the correction of defective Work, disposal of materials or equipment wrongly supplied, and making good any damage to property.	<ol><li>Other overhead or general expense costs of any kind and the costs of any item not specifically and expressly included in Paragraph 13.01.8.</li></ol>	D. Contractor's Fee: When the Work as a whole is performed on the basis of cost-plus, Contractor's fee shall be determined as set forth in the Agreement. When the value of any Work covered by a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price is determined on the basis of Cost of the Work. Contractor's fee shall be		E. Documentation: Whenever the Cost of the Work for any purpose is to be determined pursuant to this Article 13, Contractor will establish and maintain records thereof in	accordance with generally accepted accounting practices and submit in a form acceptable to Engineer an itemized cost breakdown together with supporting data.	13.02 Allowances	A. It is understood that Contractor has included in the Contract Price all allowances so named in the Contract Documents and shall cause the Work so covered to be performed for such sums and by such persons or entities as may be acceptable to Owner and Engineer.	ELCDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 50 of 65

Ł

All a model of A

. )

ARTICLE 14 – TESTS AND INSPECTIONS; CORRECTION, REMOVAL OR ACCEPTANCE OF DEFECTIVE WORK

### 14.01 Access to Work

. Owner, Engineer, their consultants and other representatives and personnel of Owner, independent testing laboratories, and authorities having jurisdiction will have access to the Site and the Work at reasonable times for their observation, inspection, and testing. Contractor shall provide them proper and safe conditions for such access and advise them of Contractor's safety procedures and programs so that they may comply therewith as applicable.

## 14.02 Tests, Inspections, and Approvals

- A. Contractor shall give Engineer timely notice of readiness of the Work (or specific parts thereof) for all required inspections and tests, and shall cooperate with inspection and testing personnel to facilitate required inspections and tests.
- B. Owner shall retain and pay for the services of an independent inspector, testing laboratory, or other qualified individual or entity to perform all inspections and tests expressly required by the Contract Documents to be furnished and paid for by Owner, except that costs incurred in connection with tests or inspections of covered Work shall be governed by the provisions of Paragraph 14.05.
- C. If Laws or Regulations of any public body having jurisdiction require any Work (or part thereof) specifically to be inspected, tested, or approved by an employee or other representative of such public body, Contractor shall assume full responsibility for arranging and obtaining such inspections, tests, or approvals, pay all costs in connection therewith, and furnish Engineer the required certificates of inspection or approval.
- D. Contractor shall be responsible for arranging, obtaining, and paying for all inspections and tests required:
- by the Contract Documents, unless the Contract Documents expressly allocate responsibility for a specific inspection or test to Owner;
- to attain Owner's and Engineer's acceptance of materials or equipment to be incorporated in the Work;
- by manufacturers of equipment furnished under the Contract Documents;
- for testing, adjusting, and balancing of mechanical, electrical, and other equipment to be incorporated into the Work; and
- for acceptance of materials, mix designs, or equipment submitted for approval prior to Contractor's purchase thereof for incorporation in the Work.

Such inspections and tests shall be performed by independent inspectors, testing laboratories, or other qualified individuals or entities acceptable to Owner and Engineer.

- E. If the Contract Documents require the Work (or part thereof) to be approved by Owner, Engineer, or another designated individual or entity, then Contractor shall assume full responsibility for arranging and obtaining such approvals.
- F. If any Work (or the work of others) that is to be inspected, tested, or approved is covered by Contractor without written concurrence of Engineer, Contractor shall, if requested by Engineer, uncover such Work for observation. Such uncovering shall be at Contractor's expense unless Contractor had given Engineer timely notice of Contractor's intention to

ELCDC\* C.700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 52 of 65

2

3

7

7

cover the same and Engineer had not acted with reasonable promptness in response to such notice.

### 14.03 Defective Work

- A. Contractor's Obligation: It is Contractor's obligation to assure that the Work is not defective.
- B. Engineer's Authority: Engineer has the authority to determine whether Work is defective, and to reject defective Work.
- C. Notice of Defects: Prompt notice of all defective Work of which Owner or Engineer has actual knowledge will be given to Contractor.
- D. Correction, or Removal and Replacement: Promptly after receipt of written notice of defective Work, Contractor shall correct all such defective Work, whether or not fabricated, installed, or completed, or, if Engineer has rejected the defective Work, remove it from the Project and replace it with Work that is not defective.
- E. Preservation of Warranties: When correcting defective Work, Contractor shall take no action that would void or otherwise impair Owner's special warranty and guarantee, if any, on said Work.
- F. Costs and Damages: In addition to its correction, removal, and replacement obligations with respect to defective Work, Contractor shall pay all claims, costs, losses, and damages arising out of or relating to defective Work, including but not limited to the cost of the inspection, testing, correction, removal, replacement, or reconstruction of such defective Work, fines levied against Owner by governmental authorities because the Work is defective, and the costs of repair or replacement of work of others resulting from defective Work. Prior to final payment, if Owner and Contractor are unable to agree as to the measure of such claims, costs, losses, and damages resulting from defective Work, then Owner may impose a reasonable set-off against payments due under Article 15.

## 14.04 Acceptance of Defective Work

₽ If, instead of requiring correction or removal and replacement of defective Work, Owner prefers to accept it, Owner may do so (subject, if such acceptance occurs prior to final Contractor shall pay all claims, costs, losses, and damages attributable to Owner's design intent and applicable engineering principles, and will not endanger public safety) payment, to Engineer's confirmation that such acceptance is in general accord with the necessary revisions in the Contract Documents with respect to the Work shall not otherwise paid by Contractor. If any such acceptance occurs prior to final payment, the by Engineer as to reasonableness), and for the diminished value of the Work to the extent evaluation of and determination to accept such defective Work (such costs to be approved defective Work occurs after final payment, Contractor shall pay an appropriate amount to impose a reasonable set-off against payments due under Article 15. If the acceptance of Contract Price, reflecting the diminished value of Work so accepted, then Owner may incorporated in a Change Order. If the parties are unable to agree as to the decrease in the Owner. be

### 14.05 Uncovering Work

₽

- Engineer has the authority to require additional inspection or testing of the Work, whether or not the Work is fabricated, installed, or completed.
- in which development for faithant

ELCDC<sup>2</sup> C-700 (Rev. 1) Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 53 of 65

ز

Ĵ

If any Work is covered contrary to the written request of Engineer, then Contractor shall, if include but not be limited to all costs of repair, or replacement of work of others destroyed requested by Engineer, uncover such Work for Engineer's observation, and then replace the replacement of Work. The covering, all at Contractor's expense.		5	15.01 Prog	all claims, costs, losses, and damages arising out of or relating to such uncovering, exposure, observation, inspection, and testing, and of satisfactory replacement or		of others); and pending Contractor's full discharge of this responsibility the Owner shall be entitled to impose a reasonable set-off against navments fure under Arricle 15.		increase in the Contract Price or an extension of the Contract Times, or both, directly	such uncovering, exposure, observation, inspection, testing, Leconstruction. If the parties are unable to agree as to the amount		Work completed as of the date of the Application and accompanied by such summeries were accompanied by such accompanied by such accompanies of the contract because is a summeries of the contract because is a summe	If the Work is defective, or Contractor fails to supply sufficient skilled workers or suitable materials or equipment, or fails to perform the Work in such a way that the completed		the Work, or any portion thereof, until the cause for such order has been eliminated; however, this right of Owner to stop the Work shall not ever rise to any duty on the part of	Owner to exercise this right for the benefit of Contractor, any Subcontractor, any Supplier, any other individual or earthy or any survey, for or conserved states therein, all of which must be satisfactory to Owner.	מי מווץ שמי בגיץ וסו, טי בוווףוטארכי טו מצבוור טו מווץ טו נוופווו. 2.			3. In e amount or retainage with respect to progress payments will be as stipulated in the Amount or retainage with respect to progress payments will be as stipulated in the Agreement.	inay, area seven uays written ribute to contractor, correct of remeay any such deficiency. In everyising the righte and remedias under this Demarch 14.07. Ourses shall accord	ex extensions, or character statements and agraph action. Owner and proceed and action for parment, including a second statement of a second statement including and action for parment. Including a second seco				stored elsewhere. Contractor shall allow Owner's representatives, agents and	access to 2.	will constitute a representation by Engineer to Owner,	All claims, costs, losses, and damages incurred or sustained by Owner in exercising the rights and remediate under this parastant 14.07 will be charged assigned for parametian for parametian for parametian and the		ELCDC* C-700 [Rev. 1], Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineeria, American Council of Engineering Companies, American Council of Engineering Companies, American Council of Engineering Companies,
B. If any Work is covered col requested by Engineer, u the covering, all at Contra	<ul> <li>C. If Engineer considers it ne or inspected or tested br expose, or otherwise mak</li> </ul>	require, that portion of th equipment.	1. If it is found that the	all claims, costs, loss exposure, observatic	reconstruction (inclu	of others); and penc shall be entitled to in	2. If the uncovered Wo	increase in the Contr	attributable to suc replacement, and rec	or extent thereof, th	14 D6 Owner May Stop the Work	A. If the Work is defective, c materials or equipment, c	Work will conform to the	the Work, or any portion however, this right of Owr	Owner to exercise this right any other individual or ant	14.07 Owner Mary Correct Defective Work	A. If Contractor fails within a	Contractor fails to perform	Contractor fails to comply	IIIay, arter seven uays write B In evercising the rights ar		Contractor from all or par	suspend Contractor's servi	and equipment stored at 1	stored elsewhere. Contra	employees, Owner's other the Site to enable Owner to		C. All claims, costs, losses, a rights and remedies under	offs against payments due	EICDC® C-700 (Rev. 1), St Copyright © 2013 National Society o

a, the Work has progressed to the point indicated;

- o, the quality of the Work is generally in accordance with the Contract Documents Unit Price Work under Paragraph 13.03, and any other qualifications stated in the Contract Documents, a final determination of quantities and classifications for Substantial Completion, the results of any subsequent tests called for in the (subject to an evaluation of the Work as a functioning whole prior to or upon recommendation); and
- to have been fulfilled in so far as it is Engineer's responsibility to observe the the conditions precedent to Contractor's being entitled to such payment appear
- ω By recommending any such payment Engineer will not thereby be deemed to have represented that:

Work.

è inspections made to check the quality or the quantity of the Work as it has been progress, or involved detailed inspections of the Work beyond the responsibilities performed have been exhaustive, extended to every aspect of the Work in

- . there may not be other matters or issues between the parties that might entitle Contractor to be paid additionally by Owner or entitle Owner to withhold specifically assigned to Engineer in the Contract; or
- Neither Engineer's review of Contractor's Work for the purposes of recommending payments nor Engineer's recommendation of any payment, including final payment payment to Contractor.
- è, to supervise, direct, or control the Work, or

will impose responsibility on Engineer:

- 9 for the means, methods, techniques, sequences, or procedures of construction. or the safety precautions and programs incident thereto, or
- ٩ for Contractor's failure to comply with Laws and Regulations applicable to Contractor's performance of the Work, or
- ġ. to make any examination to ascertain how or for what purposes Contractor has used the money paid on account of the Contract Price, or
- e. to determine that title to any of the Work, materials, or equipment has passed to Owner free and clear of any Liens
- 'n Engineer may refuse to recommend the whole or any part of any payment if, in Engineer's opinion, it would be incorrect to make the representations to Owner stated
- σ Engineer will recommend reductions in payment (set-offs) necessary in Engineer's in Paragraph 15.01.C.2.
- opinion to protect Owner from loss because:
- the Work is defective, requiring correction or replacement;
- e,
- ŗ the Contract Price has been reduced by Change Orders;
- ? Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
- d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible; or

Copyright ECOC C.700 [Rev. 1], Standard General Conditions of the Construction Contract. © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 56 of 65

27

7

- e. Engineer has actual knowledge of the occurrence of any of the events that would constitute a default by Contractor and therefore justify termination for cause under the Contract Documents.
- Ģ Payment Becomes Due:
- Ten days after presentation of the Application for Payment to Owner with Engineer's become due, and when due will be paid by Owner to Contractor. recommendation, the amount recommended (subject to any Owner set-offs) will
- m Reductions in Payment by Owner:
- ۲ In addition to any reductions in payment (set-offs) recommended by Engineer, Owner is entitled to impose a set-off against payment based on any of the following:
- ຸມ claims have been made against Owner on account of Contractor's conduct in the Regulations, and patent infringement; workplace injuries, adjacent property damage, non-compliance with Laws and the Work, including but not limited to claims, costs, losses, or damages from damages on account of Contractor's conduct in the performance or furnishing of performance or furnishing of the Work, or Owner has incurred costs, losses, or
- ē Contractor has failed to take reasonable and customary measures to avoid damage, delay, disruption, and interference with other work at or adjacent to the Site;
- ņ Contractor has failed to provide and maintain required bonds or insurance;
- <u>a</u> Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible;
- P Owner has incurred extra charges or engineering costs related to submittal reviews, evaluations of proposed substitutes, tests and inspections, or return
- the Work is defective, requiring correction or replacement visits to manufacturing or assembly facilities;
- ά Owner has been required to correct defective Work in accordance with Paragraph
- 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
- 5 the Contract Price has been reduced by Change Orders;
- an event that would constitute a default by Contractor and therefore justify a termination for cause has occurred
- Milestones, Substantial Completion, or final completion of the Work; liquidated damages have accrued as a result of Contractor's failure to achieve
- Liens have been filed in connection with the Work, except where Contractor has delivered a specific bond satisfactory to Owner to secure the satisfaction and
- discharge of such Liens;
- there are other items entitling Owner to a set off against the amount
- 2 If Owner imposes any set-off against payment, whether based on its own knowledge recommended.
- or on the written recommendations of Engineer, Owner will give Contractor immediate written notice (with a copy to Engineer) stating the reasons for such action and the specific amount of the reduction, and promptly pay Contractor any amount

ECCC<sup>2</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 57 of 65

\_

لنذ

remaining after deduction of the amount so withheld. Owner shall promptly pay Contractor the amount so withheld, or any adjustment thereto agreed to by Owner and Contractor, if Contractor remedies the reasons for such action. The reduction imposed shall be binding on Contractor unless it duly submits a Change Proposal contesting the reduction.

Upon a subsequent determination that Owner's refusal of payment was not justified, the amount wrongfully withheld shall be treated as an amount due as determined by Paragraph 15.01.C.1 and subject to interest as provided in the Agreement.

## 15.02 Contractor's Warranty of Title

A. Contractor warrants and guarantees that title to all Work, materials, and equipment furnished under the Contract will pass to Owner free and clear of (1) all Liens and other title defects, and (2) all patent, licensing, copyright, or royalty obligations, no later than seven days after the time of payment by Owner.

### 15.03 Substantial Completion

- A. When Contractor considers the entire Work ready for its intended use Contractor shall notify Owner and Engineer in writing that the entire Work is substantially complete and request that Engineer issue a certificate of Substantial Completion. Contractor shall at the same time submit to Owner and Engineer an initial draft of punch list items to be completed or corrected before final payment.
- B. Promptly after Contractor's notification, Owner, Contractor, and Engineer shall make an inspection of the Work to determine the status of completion. If Engineer does not consider the Work substantially complete, Engineer will notify Contractor in writing giving the reasons therefor.
- If Engineer considers the Work substantially complete, Engineer will deliver to Owner a preliminary certificate of Substantial Completion which shall fix the date of Substantial Completion. Engineer shall attach to the certificate a punch list of items to be completed or corrected before final payment. Owner shall have seven days after receipt of the to any provisions of the certificate or attached punch list. If, after considering the objections to the provisions of the preliminary certificate, Engineer concludes that the Work is not certificate to Owner, notify Contractor in writing that the Work is not substantially complete, stating the reasons therefor. If Owner does not object to the provisions of the substantially complete, Engineer will, within 14 days after submission of the preliminary certificate, or if despite consideration of Owner's objections Engineer concludes that the Work is substantially complete, then Engineer will, within said 14 days, execute and deliver to Owner and Contractor a final certificate of Substantial Completion (with a revised punch list of items to be completed or corrected) reflecting such changes from the preliminary certificate as Engineer believes justified after consideration of any objections from Owner. preliminary certificate during which to make written objection to Engineer as ن
- D. At the time of receipt of the preliminary certificate of Substantial Completion, Owner and Contractor will confer regarding Owner's use or occupancy of the Work following Substantial Completion, review the builder's risk insurance policy with respect to the end of the builder's risk coverage, and confirm the transition to coverage of the Work under a permanent property insurance policy held by Owner. Unless Owner and Contractor agree otherwise in writing. Owner shall bear responsibility for security, operation, protection of the Work, property insurance, maintenance, heat, and utilities upon Owner's use or occupancy of the Work.

- E. After Substantial Completion the Contractor shall promptly begin work on the punch list of items to be completed or corrected prior to final payment. In appropriate cases Contractor may submit monthly Applications for Payment for completed punch list items, following the progress payment procedures set forth above.
- F. Owner shall have the right to exclude Contractor from the Site after the date of Substantial Completion subject to allowing Contractor reasonable access to remove its property and complete or correct items on the punch list.

## 15.04 Partial Use or Occupancy

- A. Prior to Substantial Completion of all the Work, Owner may use or occupy any substantially completed part of the Work which has specifically been identified in the Contract Documents, or which Owner, Engineer, and Contractor agree constitutes a separately functioning and usable part of the Work that can be used by Owner for its intended purpose without significant interference with Contractor's performance of the remainder of the Work, subject to the following conditions:
  - At any time Owner may request in writing that Contractor permit Owner to use or occupy any such part of the Work that Owner believes to be substantially complete. If and when Contractor agrees that such part of the Work is substantially complete, Contractor, Owner, and Engineer will follow the procedures of Paragraph 15.03.A through E for that part of the Work.
- At any time Contractor may notify Owner and Engineer in writing that Contractor considers any such part of the Work substantially complete and request Engineer to issue a certificate of Substantial Completion for that part of the Work.
- 3. Within a reasonable time after either such request, Owner, Contractor, and Engineer shall make an inspection of that part of the Work to determine its status of completion. If Engineer does not consider that part of the Work to be substantially complete, Engineer will notify Owner and Contractor in writing giving the reasons therefor. If Engineer considers that part of the Work to be substantially complete, the provisions of Paragraph 15.03 will apply with respect to certification of Substantial Completion of that part of the Work and the division of Substantial Completion of that part of the Work and the division of substantial thereof and access thereto.
- No use or occupancy or separate operation of part of the Work may occur prior to compliance with the requirements of Paragraph 6.05 regarding builder's risk or other property insurance.

### 15.05 Final Inspection

A. Upon written notice from Contractor that the entire Work or an agreed portion thereof is complete, Engineer will promptly make a final inspection with Owner and Contractor and will notify Contractor in writing of all particulars in which this inspection reveals that the Work, or agreed portion thereof, is incomplete or defective. Contractor shall immediately take such measures as are necessary to complete such Work or remedy such deficiencies.
 15.06 Final Poyment

- A. Application for Payment:
- After Contractor has, in the opinion of Engineer, satisfactorily completed all corrections identified during the final inspection and has delivered, in accordance with the Contract Documents all maintenance and constitue instantian contained.

the Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of ECOC® C-700 (Rev. 1), Standard General Conditions of the Construction Contract.

EICDC<sup>®</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page S9 of 65

EICDC<sup>4</sup> C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 58 of 65

inspection, annotated record documents (as provided in Paragraph 7.11), and other documents, Contractor may make application for final payment.

The final Application for Payment shall be accompanied (except as previously delivered) by:

2

- all documentation called for in the Contract Documents;
- b. consent of the surety, if any, to final payment;
- c. satisfactory evidence that all title issues have been resolved such that title to all Work, materials, and equipment has passed to Owner free and clear of any Liens or other title defects, or will so pass upon final payment.
- a list of all disputes that Contractor believes are unsettled; and

<u>a</u>

- complete and legally effective releases or waivers (satisfactory to Owner) of all Lien rights arising out of the Work, and of Liens filed in connection with the Work.
- 3. In lieu of the releases or waivers of Liens specified in Paragraph 15.06.A.2 and as approved by Owner, Contractor may furnish receipts or releases in full and an affidavit of Contractor that: (a) the releases and receipts include all labor, services, material, and equipment for which a Lien could be filed; and (b) all payrolls, material and equipment bills, and other indebtedness connected with the Work for which Owner might in any way be responsible, or which might in any way result in liens or other burdens on Owner's property, have been paid or otherwise satisfied. If any Subcontractor or Supplier fails to furnish such a release or receipt in full, Contractor may furnish a bond or other collateral satisfactory to Owner to indemnify Owner against any Lien, or Owner at its option may issue joint checks payable to Contractor and specified Subcontractors and Suppliers.
- B. Engineer's Review of Application and Acceptance:
- ٠ If, on the basis of Engineer's observation of the Work during construction and final inspection, and Engineer's review of the final Application for Payment Contractor, indicating in writing the reasons for refusing to recommend final payment, Paragraph 15.07. Otherwise, Engineer will return the Application for Payment to respect to progress payments. At the same time Engineer will also give written notice Engineer's opinion to protect Owner from loss for the reasons stated above with payment and present the Application for Payment to Owner for payment. Such Application for Payment, indicate in writing Engineer's recommendation of final the Contract have been fulfilled, Engineer will, within ten days after receipt of the final satisfied that the Work has been completed and Contractor's other obligations under accompanying documentation as required by the Contract Documents, Engineer is in which case Contractor shall make the necessary corrections and resubmit the to Owner and Contractor that the Work is acceptable, subject to the provisions recommendation shall account for any set-offs against payment that are necessary in Application for Payment. and ę,
- C. Completion of Work: The Work is complete (subject to surviving obligations) when it is ready for final payment as established by the Engineer's written recommendation of final payment.
- D. Payment Becomes Due: Thirty days after the presentation to Owner of the final Application for Payment and accompanying documentation, the amount recommended by Engineer (less any further sum Owner is entitled to set off against Engineer's recommendation,

ECDC® C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers. American council of Engineering Companies and American Society of Civil Engineers. All rights reserved. Page 60 of 65

3

Þ

1

T.

including but not limited to set-offs for liquidated damages and set-offs allowed under the provisions above with respect to progress payments) will become due and shall be paid by Owner to Contractor.

### 15.07 Waiver of Claims

- A. The making of final payment will not constitute a waiver by Owner of claims or rights against Contractor. Owner expressly reserves claims and rights arising from unsettled Liens, from defective Work appearing after final inspection pursuant to Paragraph 15.05, from Contractor's failure to comply with the Contract Documents or the terms of any special guarantees specified therein, from outstanding Claims by Owner, or from Contractor's continuing obligations under the Contract Documents.
- B. The acceptance of final payment by Contractor will constitute a waiver by Contractor of all claims and rights against Owner other than those pending matters that have been duly submitted or appealed under the provisions of Article 17.

### 15.08 Correction Period

- A. If within one year after the date of Substantial Completion (or such longer period of time as may be prescribed by the terms of any applicable special guarantee required by the Contract Documents, or by any specific provision of the Contract Documents), any Work is found to be defective, or if the repair of any damages to the Site, adjacent areas that Contractor has arranged to use through construction easements or otherwise, and other adjacent areas used by Contractor as permitted by Laws and Regulations, is found to be defective, then Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions:
- correct the defective repairs to the Site or such other adjacent areas;
- correct such defective Work;

ώŅ

- if the defective Work has been rejected by Owner, remove it from the Project and replace it with Work that is not defective, and
- satisfactorily correct or repair or remove and replace any damage to other Work, to the work of others, or to other land or areas resulting therefrom.

σ

- If Contractor does not promptly comply with the terms of Owner's written instructions, or in an emergency where delay would cause serious risk of loss or damage. Owner may have the defective Work corrected or repaired or may have the rejected Work removed and replaced. Contractor shall pay all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or repair or such removal and replacement (including but not limited to all costs of repair or replacement of work of others).
- In special circumstances where a particular item of equipment is placed in continuous service before Substantial Completion of all the Work, the correction period for that Item may start to run from an earlier date if so provided in the Specifications.

0

D. Where defective Work (and damage to other Work resulting therefrom) has been corrected or removed and replaced under this paragraph, the correction period hereunder with respect to such Work will be extended for an additional period of one year after such correction or removal and replacement has been satisfactorily completed.

ELCDC\* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 61 of 65

and damages exceeds such unpaid balance, Contractor shall pay the difference to Owner. Such claims, costs, losses, and damages incurred by Owner will be reviewed by Engineer as to their reasonableness and, when so approved by Engineer, incorporated in a Chanee	Order. When exercising any rights or remedies under this paragraph, Owner shall not be required to obtain the lowest price for the Work performed.	F. Where Contractor's services have been so terminated by Owner, the termination will not	affect any rights or remedies of Owner against Contractor then existing or which may thereafter accrue, or any rights or remedies of Owner against Contractor or any surety under any payment bond or performance bond. Any retention or payment of money due Contractor by Owner will not release Contractor from liability.	G. If and to the extent that Contractor has provided a performance bond under the provisions of Paragraph 6.01.A, the provisions of that bond shall govern over any inconsistent provisions of Paragraphs 16.02.B and 16.02.D.	16.03 Owner May Terminate For Convenience	<ul> <li>Upon seven days written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. In such</li> </ul>	case, Contractor shall be paid for (without duplication of any items):		<ol><li>expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents</li></ol>	in connection with uncompleted Work, plus fair and reasonable sums for overhead and profit on such expenses; and		B. Contractor shall not be paid on account of loss of anticipated overhead, profits, or revenue, or other economic loss arising out of or resulting from such termination.	ut ut	<ul> <li>A. If, through no act or fault of Contractor, (1) the Work is suspended for more than 90 consecutive days by Owner or under an order of court or other public authority. or (2)</li> </ul>	Engineer fails to act on any Application for Payment within 30 days after it is submitted, or (3) Owner fails for 30 days to nay Contractor any sum finally determined to ho due theo	Contractor may, upon seven days written notice to Owner and Engineer, and provided Owner or Engineer do not remedy such suspension or failure within that time, terminate	the contract and recover from Owner payment on the same terms as provided in Paragraph 16.03.	B. In lieu of terminating the Contract and without prejudice to any other right or remedy. if	Engineer has failed to act on an Application for Payment within 30 days after it is submitted or Owner has failed for 20 days for 20 days for the control of the second second second second second	be due. Contractor may, seven days after written notice to Owner and Engineer, stop the	Work until payment is made of all such amounts due Contractor, including interest thereon. The provisions of this paragraph are not intended to provide the Contractor from one manual of	Change Proposal for an adjustment in Contract Price or Contract Times or otherwise for	ELCDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Scolety of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers, All rights reserved. Page 63 of 65
E. Contractor's obligations under this paragraph are in addition to all other obligations and warranties. The provisions of this paragraph shall not be construed as a substitute for, or a waiver of, the provisions of any applicable statute of limitation or repose.	_	16.01 Owner May Suspend Work	A. At any time and without cause, Owner may suspend the Work or any portion thereof for a period of not more than 90 consecutive days by written notice to Contractor and Engineer. Such notice will fix the date on which Work will be resumed. Contractor shall resume the Work on the date so fixed. Contractor shall be entitled to an adjustment in the Contract	Price or an extension of the Contract Times, or both, directly attributable to any such suspension. Any Change Proposal seeking such adjustments shall be submitted no later than 30 days after the date fixed for resumption of Work.	16.02 Owner May Terminate for Cause	<ul> <li>The occurrence of any one or more of the following events will constitute a default by Contractor and justify termination for cause:</li> </ul>	<ol> <li>Contractor's persistent failure to perform the Work in accordance with the Contract Documents (including, but not limited to, failure to supply sufficient skilled workers or suitable materials or equipment or failure to adhere to the Progress Schedule);</li> </ol>	2. Failure of Contractor to perform or otherwise to comply with a material term of the	Contract Documents; 3.	4. Contractor's repeated disregard of the authority of Owner or Engineer.	B. If one or more of the events identified in Paragraph 16.02.A occurs, then after giving Contractor (and any surety) ten days written notice that Owner is considering a declaration that Contractor is in default and termination of the contract. Owner may account and the contractor is in default and termination of the contract of the contractor.	1. declare Contractor to be in default, and give Contractor (and any surret) notice that the Contract is terminated and	2. enforce the rights available to Owner under any annlicable neformance hour	C. Subject to the terms and operation of any applicable performance bond, if Owner has	terminated the Contract for cause, Owner may exclude Contractor from the Site, take possession of the Work, incorporate in the Work all materials and equipment stored at the	Site or for which Owner has paid Contractor but which are stored elsewhere, and complete the Work as Owner may deem expedient.	D. Owner may not proceed with termination of the Contract under Paragraph 16.02.B if	contributed of writing seven days of receipt of notice of intent to terminate begins to correct its failure to perform and proceeds diligently to cure such failure.	E. If Owner proceeds as provided in Paragraph 16.02.B, Contractor shall not be entitled to	receive any turther payment until the Work is completed. If the unpaid balance of the Contract Price exceeds the rost to complete the Mork including all related claims mathe	losses, and damages (including but not limited to all fees and charges of engineers,	arcintects, attorneys, and other protessionals) sustained by Owner, such excess will be paid to Contractor. If the cost to complete the Work including such related claims, costs, losses,	ELCDC* C-700 (Rev. 1), Standard General Conditions of the Construction Contract. Copyright © 2013 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 62 of 65

A second second

transformer and

Copyright © 2013 National Society of Professional Engineers. American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 64 of 65	A. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are otherwise imposed or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract. The provisions of this paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right, and remedy to which they apply.	<ul> <li>13:02 Computation of times</li> <li>A. When any period of time is referred to in the Contract by days, it will be computed to exclude the first and include the last day of such period. If the last day of any such period falls on a Saturday or Sunday or on a day made a legal holiday by the law of the applicable jurisdiction, such day will be omitted from the computation.</li> <li>18.03 Cumulative Remedies</li> </ul>	)	<ul><li>18.01 Giving Notice</li><li>A. Whenever any provision of the Contract Documents requires the giving of written notice, it will be deemed to have been validly given if:</li></ul>	ARTICLE 18 – MISCELLANEOUS	<ol> <li>agree with the other party of considered for in the Supplementary Conditions or</li> <li>if no dispute resolution process is provided for in the Supplementary Conditions or mutually agreed to, give written notice to the other party of the intent to submit the dispute to a court of competent jurisdiction.</li> </ol>	<ol> <li>elect in writing to invoke the dispute resolution process provided for in the Supplementary Conditions; or</li> <li>across with the other narty to submit the dispute to another dispute resolution</li> </ol>	<ul> <li>full; and</li> <li>2. Disputes between Owner and Contractor concerning the Work or obligations under</li> <li>the Contract Documents, and arising after final payment has been made.</li> <li>B. Final Resolution of Disputes: For any dispute subject to resolution under this Article, Owner</li> <li>or Contractor may:</li> </ul>		expenses or damage directly attributable to Contractor's stopping the Work as permitted by this paragraph. ARTICLE 17 – FINAL RESOLUTION OF DISPUTES 17.01 Methods and Procedures
Copyright © 2015 National sourcey of Civil Engineers. All rights reserved. Page 65 of 65 and American Society of Civil Engineers. All rights reserved.	EICDC <sup>2</sup> C.700 (Rev. 1), Standard General Conditions of the Construction Contract.					18.08 Headings A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.		18.06 Survival of Obligations A. All representations, indemnifications, warranties, and guarantees made in, required by, or given in accordance with the Contract, as well as all continuing obligations indicated in the given in accordance with the Contract, as well as all continuing obligations indicated in the Contract, will survive final payment, completion, and acceptance of the Work or contract, will survive final payment, completion at the services of Contractor. termination or completion of the Contract or termination of the services of Contractor.	<ul> <li>18.05 No Waiver</li> <li>A. A party's non-enforcement of any provision shall not constitute a waiver of that provision, nor shall it affect the enforceability of that provision or of the remainder of this Contract.</li> </ul>	18.04 Limitation of Damages A. With respect to any and all Change Proposals, Claims, disputes subject to final resolution, and other matters at issue, neither Owner nor Engineer, nor any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, shall be liable to Contractor for any claims, costs, losses, or damages sustained by Contractor on or in connection with any other project or anticipated project.

P. . . .

**N** - J

<u>第</u>一 二 13

ر ار

<u>ل</u>

7

لد

Ŀ,

. .)

ل ا

د -لار

Ĵ

لي ا

لتت

1

ليستناق

J

SUPPLEMENTARY GENERAL CONDITIONS

### SUPPLEMENTARY GENERAL CONDITIONS

e ...

### SUPPLEMENTARY GENERAL CONDITIONS

The following Supplementary Conditions modify, change, delete from or add to the Standard General Conditions of the Construction Contract EJCDC (C-700, 2013 Edition). Where any Article of the General Conditions is modified, or any Paragraph, Subparagraph or Clause thereof is modified or deleted by these supplements, the unaltered provisions of that Article, Paragraph, Subparagraph or Clause shall remain in effect.

### 1. Article 6 - BONDS AND INSURANCE

- A. Add the following to paragraph 6.01.C: All bonds shall be written through companies rated with A.M. Best rating of A or better.
- Β. The limits of liability for insurance required by paragraph 6.03 of the General Conditions shall provide coverage for not less than the following amounts or greater where required by Laws and Regulations:
  - Workers' Compensation, and related coverages under paragraphs 6.03.A of the 1. General Conditions:

a.	State:	Statutory
b.	Applicable Federal	Statutory
	(e.g., Longshoremen's)	5
c.	Employer's Liability	\$500,000

Contractor's General Liability under paragraphs 6.03.B of the General 2. Conditions which shall include completed operations and product liability coverages and eliminate the exclusion with respect to property under the care, custody, and control of the Contractor:

a.	General Aggregate	\$2,000,000
b.	Products - Completed	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
	Operations Aggregate	\$1,000,000
c.	Personal and Advertising Injury	\$1,000,000
d.	Each Occurrence (Bodily Injury	
	and Property Damage)	
		\$1,000,000
e.	Property Damage liability	
	insurance will provide Explosion,	
	Collapse, and Underground	
	coverages where applicable.	\$1,000,000
f.	Excess or Umbrella Liability	
	1) General Aggregate	\$1,000,000
	2) Each Occurrence	\$1,000,000
	g. Contractor shall name the Owner and Engin	neer as an
additi	onal insured	

additional insured.

- 3. Automobile Liability under paragraph 6.03.D of the General Conditions:
- a. Combined Single Limit of \$1,000,000
- C. The CONTRACTOR shall purchase, maintain, and pay for the insurance required
- D. Add the following as item 6.08:

The CONTRACTOR, its subcontractors, if any, and all employers working under this Contract are subject to Oregon Workers' Compensation Law, and shall comply with ORS 656.017, which requires them to provide workers compensation coverage for all their subject workers.

- 2. Article 7 CONTRACTOR'S RESPONSIBILITIES
- A. Amend 7.04 "Or Equals" to note approval of an "Or Equal" may only be requested prior to the Bid as defined in the instruction to bidders. Following the expiration of the time to submit an "Or Equal", replacement of any specified component shall be considered under the requirements for a substitute as defined in 7.05 of the General Conditions.
- B. Add the following sentence the end of item 7.17 A:

The Warranty and Guarantee period shall extend for one (1) year following FINAL acceptance by the OWNER.

- 3. Article 15 PAYMENTS TO CONTRACTOR
- A. Modify the first sentence of item 15.01.B.1. to read that each application for payment shall be submitted 30 days prior to the date established in the Agreement for each progress payment.
- B. Delete 15.01.D and insert:

Within thirty days after receipt of each Application for Payment, the amount recommended will (subject to the provisions of the last sentence of paragraph 15.01.E) become due and when due, will be paid by OWNER to CONTRACTOR.

C. Add 15.06.A.2.f:

The Application for Final Payment shall also include CONTRACTOR'S Affidavit of Payment of Debts and Claims.

### 4. Add the following:

### Article 19. STATE OF OREGON PUBLIC CONTRACT REQUIREMENTS

19.01 In accordance with ORS 279C.505(1) the CONTRACTOR shall:

A. Make payment promptly, as due, to all persons supplying to such CONTRACTOR labor or material for the prosecution of the Work provided for in this Agreement.

- B. Pay all contributions or amounts due the Industrial Accident Fund from such CONTRACTOR or Subcontractor incurred in the performance of this Agreement.
- C. Not permit any lien or claim to be filed or prosecuted against the OWNER on account of any labor or material furnished.
- D. Pay to the Department of Revenue all sums withheld from employees pursuant to ORS 316.167.
- 19.02 Contractor shall demonstrate that an employee drug testing program is in place (ORS 279C.505(2)).
- 19.03 In accordance with ORS 279C.510(1) the Contractor shall salvage or recycle construction and demolition debris if feasible and cost effective.
- 19.04 In accordance with ORS279C.515(1) if the Contractor fails, neglects or refuses to pay promptly a person's claim for labor or services that the person provides to the contractor or a subcontractor in connection with the public improvement contract as the claim becomes due, the OWNER may pay the amount of the claim to the person that provides the labor or services and charge the amount of the payment against funds due or to become due the contractor by reason of the contract.
- 19.05 In accordance with ORS 279C.515(2) if the Contractor or a first-tier subcontractor fails, neglects or refuses to pay a person that provides labor or materials in connection with the public improvement contract within 30 days after receiving payment from the contracting agency or a contractor, the contractor or first-tier subcontractor owes the person the amount due plus interest charges that begin at the end of the 10-day period within which payment is due under ORS 279C.580 (4) and that end upon final payment, unless payment is subject to a good faith dispute as defined in ORS 279C.580. The rate of interest on the amount due is nine percent per annum. The amount of interest may not be waived.
- 19.06 In accordance with ORS 279C.515(3) if the Contractor or a subcontractor fails, neglects or refuses to pay a person that provides labor or materials in connection with the public improvement contract, the person may file a complaint with the Construction Contractors Board, unless payment is subject to a good faith dispute as defined in ORS 279C.580.

19.07 In accordance with ORS 279C.520 a person may not be employed for more than 10 emergency or when the public policy absolutely requires it, and in such cases, except in cases of contracts for personal services as defined in ORS 279C.100, the employee shall be paid at least time and a half pay:

A. For all overtime in excess of eight hours in any one day or 40 hours in any one week when the work week is five consecutive days, Monday through Friday; or

 B. For all overtime in excess of 10 hours in any one day or 40 hours in any one week when the work week is four consecutive days, Monday through Friday; and

C. For all work performed on Saturday and on any legal holiday specified in ORS 279C.540.

An employer must give notice in writing to employees who work on a public contract, either at the time of hire or before commencement of work on the contract, or by posting a notice in a location frequented by employees, of the number of hours per day and days per week that the employees may be required to work.

- 19.08 In accordance with 279C.530 the contractor shall promptly, as due, make payment to any person, copartnership, association or corporation furnishing medical, surgical and hospital care services or other needed care and attention, incident to sickness or pay for the employees of the contractor, of all sums that the contractor collected or deducted from the wages of employees under any law, contractor collected or number the contract must be either employees under any law, contractor spreement for the employers that are exempt under ORS 656.126. Contractor shall ensure that each of employers that are exempt under ORS 656.126. Contractor shall ensure that each of its subcontractors complies with these requirements.
- 19.09 In accordance with ORS 279C.545, any worker employed by the contractor shall be foreclosed from the right to collect for any overtime provided in ORS 279C.540 unless a claim for payment is filed with the Contractor within 90 days from the completion of the contract, providing the Contractor has (1) caused a circular be posted in a prominent place alongside the door of the timekeeper's office or in a similar place that is readily available and freely visible to workers employed on the work, and (2) maintained the circular continuously posted from the inception to the tompletion of the contract on which workers are or have been employed.
- 19.10 The Contractor shall comply with the requirements of ORS 279C.570 as they apply including prompt payment policy, progress payments and rate of interest.

19.11 The Contractor shall comply with the provisions of ORS 279C.580 as they apply with regard to Contractor's relation with subcontractors, including that the Contractor is required to include in each subcontract entered into with a first-tier subcontractor, including a material supplier, for the purpose of performing a construction contract:

A. A payment clause obligating the Contractor to pay the first-tier subcontractor for satisfactory performance under its subcontract within 10 days out of amounts the OWNER pays to the Contractor under the public improvement contract;

B. A clause that requires the Contractor to provide a first-tier subcontractor with a standard form for application for payment.

C. A clause that requires the Contractor to use the standard form and administrative procedures for payment for the entire term of the subcontract.

- E. An interest penalty clause conforming with ORS 279C.580(3)(b).
- 19.12 The Contractor shall comply with the provisions of ORS 279C.605 as they apply with regard to notice of claims.
- 19.13 This public works contract is subject to the requirements of the Oregon Prevailing Wage Rates. In accordance with 279C.830 Contractor shall comply with the existing state prevailing rate of wage that must be paid to workers in each trade or occupation required for the public works employed in the performance of the contract either by the Contractor or subcontractor or other person doing or contracting to do the whole or any part of the work contemplated by the contract. Workers shall be paid not less than the specified minimum hourly rate of wage.
- 19.14 The OWNER will pay the required fee to the Commissioner of the Bureau of Labor and Industries as provided in ORS 279C.825 (1), under the administrative rule of the commissioner.
- 19.15 The Contractor and every subcontractor must have a public works bond filed with the Construction Contractors Board before starting work on the project, unless exempt under ORS 279C.836 (7) or (8). Contractor shall also include in every subcontract a provision requiring the subcontractor to have a public works bond filed with the Construction Contractors Board before starting work on the project, unless exempt under ORS 279C.836 (7) or (8).
- 19.16 By Submission of a bid the Contractor certifies compliance with the Oregon tax laws in accordance with ORS 305.385.
- 5. Add the following:

Article 20. FEDERAL, STATE AND LOCAL STATUTES

20.01 CONTRACTOR is responsible for compliance with State and Federal safety and health acts, ORS 654.001 et. seq., and 29 USC 651 et. seq., and the regulations promulgated thereunder.

20.02 The CONTRACTOR shall comply with all federal, state and local laws, regulations, executive orders and ordinances applicable to the work under this agreement. Without limiting the generality of the foregoing, CONTRACTOR Act of 1964; (ii) the Americans with Disabilities Section V of the Rehabilitation Act of 1973; (iii) the Americans with Disabilities established pursuant to the foregoing laws; (v) all other applicable requirements of federal and state civil rights and rehabilitation statutes, rules and regulations; and (vi) the environmental laws and regulations enacted by appropriate public agencies.

An employer must give notice in writing to employees who work on a public contract, either at the time of hire or before commencement of work on the contract, or by posting a notice in a location frequented by employees, of the number of hours per day and days per week that the employees may be required to work.

### PREVAILING WAGE RATE FORMS

### PREVAILING WAGE RATE FORMS

.

12

 $\gamma$ 

~

ليہ ،

د : ز\_ر

لحتد ١٠

U

لأنذر

### PREVAILING WAGE RATES FOR PUBLIC WORKS CONSTRUCTION

5

This project is subject to the State of OREGON BOLI Prevailing Wage Rates, effective as of the date of the publication soliciting bids, which are in corporate by reference.

The current Prevailing Wage Rate Book is available online and can be viewed and printed in its entirety at: <u>http://www.oregon.gov</u>.

This Page Intentionally Left Blank

÷.

7

3

<u>ار</u> ر

 $\overline{\gamma}$ 

. ...

لبريا

لتد

لہ .

ليريد

1



**TECHNICAL SPECIFICATIONS** 

### **LECHNICYL SPECIFICATIONS**

Ĵ.

ز. .

ل. .



**DIVISION 1 - GENERAL REQUIREMENTS** 

### DIAISION I - CENEBVT BEONIBEMENLS

لد

لي

لتد


PLAN DRAWINGS

3

7)

# SECTION 01000 GENERAL REQUIREMENTS

# PART 1 - GENERAL

# 1.01 SUMMARY OF WORK

- A. The CONTRACTOR shall furnish all labor, material and equipment necessary to complete the Work in all respects as shown on the plans and/or as specified herein and under the terms of the Contract.
- B. The work in this Contract is as described in the Notice to Contractors, Contract for Construction and the Bid Proposal.

# 1.02 GENERAL CONSTRUCTION

- A. The intent of the Contract Documents is to ensure the systematic implementation of construction with a minimum of public inconvenience. The CONTRACTOR shall coordinate his work to minimize any inconvenience to private property owners.
- B. The CONTRACTOR shall have on the job, at all times, a qualified superintendent as his agent, who is capable of reading and understanding the Contract Documents and is thoroughly experienced in the type of work being performed. The Superintendent shall be responsible for the progress of construction and the CONTRACTOR'S operations.
- C. The CONTRACTOR shall be responsible for cooperation with other utilities and any other contractors which might be employed by the OWNER on the project.
- D. The CONTRACTOR is responsible for maintenance of all streets, roads and construction sites within which Work is done until final acceptance of the Work by the OWNER. The CONTRACTOR shall pay all costs of modifying existing utility systems to meet specific construction needs, if required.
- E. The CONTRACTOR shall perform the Work in accordance with the plans, specifications and Contract terms except as modified in writing by the ENGINEER. He shall perform all work determined by the ENGINEER as necessary to properly prosecute and complete the project.
- F. The CONTRACTOR is responsible for controlling stormwater, mud, debris and the disposition of construction related materials. These substances shall be positively prevented from entering the storm or sanitary sewer system. Clean up resulting from the improper handling of these substances will be the CONTRACTOR'S responsibility.

- G. Conflicts in the contract documents shall be resolved based on the following
- 1. Technical Specifications
- 2. Design Drawings
- 3. Supplementary General Conditions
- 4. General Conditions

# 1.03 CONSTRUCTION LAYOUT

- A. The Work exact limits will be located in the field by the OWNER or the ENGINEER.
- B. The CONTRACTOR shall be responsible for replacement of all permanent the construction. Recorded boundary markers shall be replaced by a Registered by the construction. Recorded boundary markers shall be replaced by a Registered to find Surveyor. No permanent, recorded boundary markers shall be replaced by a Registered by a removed or disturbed without authorization of the ENGINEER.

#### 1.04 WARRANTY

The CONTRACTOR shall make all necessary repairs and replacements to remedy in a manner satisfactory to and at no cost to the OWNER, any and all defects, breaks, or failures of the work occurring within **ONE** (I) **YEAR FOLLOWING THE DATE OF** workmanship, and for damage caused by settling, washing or slipping when such damage or disturbance is caused, in whole or in part, from activities of the CONTRACTOR in performing the duties and obligations under this contract. When such defects or damage or disturbance is caused in whole or in part, from activities of the CONTRACTOR in performing the duties and obligations under this contract. When such defects or damage occur within the time period described hereinbefore, in any part of the surface or subsurface or the period described hereinbefore, in any part of the surface or tanage improvement not included in the work under the contract, the CONTRACTOR shall repair the same and the one year maintenance period required shall, with relation to such the tequired repair, be extended one year from the date of completion of such repair.

# 1.05 CONTRACTOR RESPONSIBILITY

- A. The CONTRACTOR shall be fully informed on all Federal and State laws and all local laws, ordinances and regulations of bodies having jurisdiction or authority or which in any way might affect the conduct of the Work.
- B. The CONTRACTOR shall indemnify and protect the OWNER and ENGINEER against any claim or liability arising from a violation of any law, ordinance or regulation by himself, any subcontractor or employee of the CONTRACTOR or subcontractor.

- C. The CONTRACTOR shall obtain and pay for all licenses and permits and shall be responsible for all fees, taxes or payments required for the lawful and due performance of the Work, except as defined herein.
- D. The CONTRACTOR shall be solely responsible for any trespass on adjacent properties or injury thereto, resulting from his operations. All private property damaged by his operations shall be fully restored to preconstruction conditions.
- E. Public safety and convenience shall be paramount in the CONTRACTOR'S operations and shall be provided for in a satisfactory manner. All laws, rules, ordinances and regulations shall be strictly adhered to by the CONTRACTOR. The CONTRACTOR shall perform his operations so as to minimize public and commercial inconvenience.
- F. Wherever and whenever a possible public hazardous situation shall occur, the CONTRACTOR shall be responsible for whatever signing, barricades or other safety precautions are necessary to protect the public and employees on the project.
- G. No explosive shall be stored, handled or used on the project without the authorization of the ENGINEER. No unlicensed or unqualified person shall handle or use explosives, and any damage resulting from the use of explosives shall be the CONTRACTOR'S responsibility.
- H. All approved uses of explosives shall be accomplished with utmost care in order to not endanger life or property. The CONTRACTOR shall be responsible for notification of adjacent property owners, utility companies and others having property or facilities in proximity to the site. Sufficient advance notice shall be given to allow nearby property to be properly protected from damage.

# 1.06 UTILITIES COORDINATION

- A. All coordination with utilities, including but not limited to water, sewer, power, gas, television, telegraph and telephone, shall be accomplished by the CONTRACTOR prior to any construction. No extra costs for damages or delay will be approved as a result of the CONTRACTOR'S failure to contact utilities or to arrange sufficient time for private utility infrastructure construction.
- B. Locations of utilities are not shown on the plans. The CONTRACTOR is responsible for determining the exact location with the assistance of the utility companies and to properly account for the possible interference of utilities property with his operations.
- C. In the event of an interruption of utilities service by his operations, the CONTRACTOR is solely responsible for repair costs and/or penalties accrued as a result of the interruptions. All planned interruptions of service shall be coordinated with the OWNER and operators, and kept to a minimum.

232-1987 or 1-800-332-2344 or 811 952-001-0090. Copies of the rules may be obtained by calling the Center at 503-Notification Center. These rules are set forth in OAR 952-001-0010 through OAR The CONTRACTOR must follow the rules adopted by the Oregon Utility D'

#### TRAFFIC CONTROL AND PUBLIC NOTIFICATION: 70.1

- Manual of Uniform Traffic Control Devices. work zone is properly identified in compliance with the current edition of the The CONTRACTOR shall provide adequate signing and flaggers to ensure the .Α
- access restored within 4 hours of their closure. flaggers as needed to maintain traffic. Cul-de-sacs and dead end streets shall have his work and obtain approval/ concurrence. Provide advance warning signs, and shall coordinate with the City all traffic control impacts on all streets as a result of CONTRACTOR shall at no time close any intersection. The CONTRACTOR routes must be properly signed and pre-approved by the OWNER. The road closures and/or detour routes on all the side streets and cul-de-sacs. Detour OWNER at the preconstruction meeting. The plan shall include any proposed A detailed traffic control plan and paving schedule shall be supplied to the
- CONTRACTOR to immediately take steps to bring his work within compliance. ENGINEER, the ENGINEER may stop that portion of the project and require the approvals or given proper notice as may be required in these documents or by the measures or services, or where the CONTRACTOR has not obtained suitable suitable signs, barricades, warning lights, flag men or other suitable traffic control Where, in the opinion of the ENGINEER, the CONTRACTOR has not provided
- Contractor shall prepare the notice jointly and with the approval of the Engineer. means of emergency access will be maintained at all times in all work zones. The work will be performed, stating time of day and the date of street closure. A Notification shall also include the exact location of work and the times when prior to the start of the Work, include a list or map of the streets affected. given street. Notify as shown below the agencies/ companies one week (7 days) The CONTRACTOR is responsible for notifications prior to starting work on any D'

Tel 503-655-8211	
Canby, OR 97013	
anuary are WN STIT	
Candy Police Department	

Tel 503-266-3353 Сапру, ОК 97013 элү <sub>ри</sub>г MN 519 Attn:Jami Goodpaster Canby Post Office

Tel 503-266-0751

Canby, OR 97013

Attn: Julie Wehling

Canby Transit

195 S HazelDell Way, Ste C

Tel 503-266-3900	Tel 503-266-2064
Canby, OR 97013	С <sup>зиру,</sup> ОК 97013
1600 SE 4th Ave	bA qidamoT .2 0881
Attn: Amber Drevenhouer	Attn: Byron Searls
Canby Disposal	Student Transportation of America

Tel 503-266-5851

Canby, OR 97013

221 S. Pine Street

C.

B.

Canby Fire District

The CONTRACTOR will be required to coordinate garbage and recycling materials collection schedule with Canby Disposal.

- E. The CONTRACTOR shall provide the "NO PARKING" signs for use in posting streets in advance of the Work. Streets may be posted "NO PARKING" between the hours of 7:00 AM and 5:00 PM. Posted parking prohibitions shall be restricted to one working day. Parking prohibitions proposed outside of the listed time frame or for more than one working day must have prior approval, in writing, from the CITY. Streets not completed must be rescheduled.
- F. "NO PARKING" signs shall be placed no less than twenty-four (24) hours or no more than forty-eight (48) hours prior to commencement of work. These "NO PARKING" signs shall be mounted by the CONTRACTOR on a suitable pedestal, e.g. tripod, barricade, etc., provided by the CONTRACTOR. Signs shall be posted every 100 feet on both sides of the street affected. The signs shall be maintained by the CONTRACTOR for the duration of each project. The CONTRACTOR shall remove all "NO PARKING" signs when the Work is completed on each project.
- G. The CONTRACTOR shall also provide door-knob notices in sufficient quantities to notify residents a minimum of forty-eight (48) hours prior to the start of any work on each street affected by the Work. This includes streets on which no work will take place but will be affected by the Work on the adjacent streets that serve as the sole means of access. Submit copies of notices and/ or proposed notice language to the CITY at the pre-construction conference. The notices shall be left on or at the front door of each dwelling, apartment unit, or tenant of a commercial unit abutting any of the streets on the list. This should be done at the same time the "NO PARKING" signs are first placed in the area. Each day, prior to commencement of work, the CONTRACTOR shall verbally contact, whenever possible, the residents to notify them of impending work. Notification shall be given enough in advance to allow residents to move personal vehicles prior to work beginning.
- H. The CONTRACTOR shall adhere to the published paving schedule and without delays. In the event of delays or rescheduling of the Work beyond the control of the CONTRACTOR, the CONTRACTOR must re-notify the ENGINEER, OWNER and all residents, businesses, postal services and emergency responses of the delays, resubmit an updated paving schedule and follow the procedures as stipulated above in sections 1.07D to 1.07 G.

# 1.08 SPECIAL REQUIREMENTS:

A. The CONTRACTOR shall provide continuous access to all local residents/ businesses, postal deliveries as well as emergency rescue traffic. Under no circumstances may access to any residents/ businesses be blocked without prearranged approval. This closure time should be kept as short as possible.

01000 - 5

- B. If the CONTRACTOR or their subcontractors do not possess a current City Business License, they will be required to obtain one from the OWNER before the start of their work.
- C. The project to minimize inconvenience to the residents and businesses.
- D. The CONTRACTOR shall schedule the construction activity to provide a minimum of ten (10) day notice to the OWNER and accomplish all work within the specified performance period.
- E. The CONTRACTOR will be required to coordinate with Mr. Jerry Nelzen at 971-253-9173 for trimming tree limbs and hedges that interferes with their activities. Allow a minimum of 48 hours prior to the scheduled paving.
- F. The CONTRACTOR shall replace all the existing street striping to match preconstruction conditions as they pertain to the schedule of Work within 15days of completing the Work but not less than a week after the treatment application. The approximate measurements and type of striping are shown in the Bid Proposal.

أسرب

لمريا

- MOBR (ILEWS C'I LHBU C'):
  BA LHE OMNEB MITT BE WYDE LO LHE CONLBYCLOB FOB SUCH ONSIDEBED INCIDENTAT LO LHE BROJECT AND NO BYAMENT BE RECESSYBA COSLS FOB THE FOLLOWING WORK WHICH WILL BE GONSIDEBED INCIDENTAT INCLUDE IN LHEIB BID ATT LHE
  G. THE CONLBYCLOB WUGL INCLUDE IN LHEIB BID ATT LHE
- G.1 The CONTRACTOR shall provide all the Traffic Control and Public Notifications (section 1.07, items A thru G) as they pertain to the respective schedule of Work.
- G.2 The CONTRACTOR shall inventory, provide and deliver to the project site all manhole risers, cleanout risers and pull up or jack hammer around all the water valve boxes to be raised to the new asphalt pavement grades. The the exact number of water valve boxes, manhole and cleanout ring risers. The CONTRACTOR will also be responsible for walking the project and inventory the exact number of water valve boxes, manhole and cleanout ring risers. The contry. This work and any variation in the published quantities will be considered incidental and no extra payment will be made by the CITY to the incidental and no extra payment will be made by the CITY to the considered
- G.3 The CONTRACTOR will be responsible for the sweeping of each street prior to the asphalt overlay application and street asphalt tacking.

- G.4 The CONTRACTOR will be responsible for coordinating with the City all the V-loc bases installation and plumbing in the concrete and sign post replacements for all the signs located in the new sidewalk.
- G.5 The CONTRACTOR must use all available means at his disposal to maintain and ensure positive drainage to the existing storm drainage collection system and to prevent puddling by verifying the grades in the field.
- G.6 As directed and to the satisfaction of the City, the CONTRACTOR will be responsible for removing any mounted asphalt or obstructions that interfere with the slurry seal or the asphalt overlays applications.
- G.7 As directed and to the satisfaction of the CITY, the CONTRACTOR will be responsible for removing any surface obstructions such as asphalt/ concrete mounted against the curb or otherwise that interferes with the overlay applications.
- G.8 The CONTRACTOR will be responsible for all coordination, delivery or pick up of the necessary riser ring adjustments related to DirectLink manholes or adjustments to NW Natural valve rings.
- G.9 The CONTRACTOR will be responsible for all the site restoration caused by the curbs, sidewalks and ADA ramps construction. The restoration shall match preexisting conditions or better and may include bark chips, sod, top soil mix, replanting bushes, shrubs, transplanting plants up to 36" in height, landscape features, etc...
- H. Authorization to proceed is required from the ENGINEER or the CITY if all the required risers for manholes, cleanouts, monuments, gas valves, etc.. and water valve pull up boxes are not completed prior to paving or overlays schedule/s. It is the responsibility of the CONTRACTOR to ensure all the necessary tasks are completed before scheduling any paving or overlays (No Exceptions). If the CONTRACTOR fails to comply with this requirement, the CITY or the ENGINEER reserves the right to stop the paving Work until compliance is achieved.
- I. The CONTRACTOR will be responsible for coordinating the overlay schedule with the City to avoid conflicts with the upcoming summer events and festivities as some of those streets may be used as a route for accommodating these occasions.
- J. The CONTRACTOR will be required to arrange an advance meeting between the City representative/ ENGINEER and the concrete subcontractor to discuss the limits of the ADA ramps reconstruction.

# **\*\* END OF SECTION \*\***

#### 01000 - 7

Anola the Vilionoiineine spag sinT

4 -

# SECTION 01500 CONTRACTOR FACILITIES AND TEMPORARY CONTROLS

# **PART 1 - GENERAL**

#### 1.01 WATER

If water is used from fire hydrants, the CONTRACTOR shall maintain adequate cross connection control and coordinate with Canby Utility. Information on hydrant meter setup is available from Canby Utility at 503-266-1156.

# 1.02 SANITARY FACILITIES

The CONTRACTOR shall provide, as a minimum, portable outside sanitary facilities for the use of the CONTRACTOR'S personnel.

#### 1.03 CLEANUP

All cleanup and site restoration shall be accomplished concurrently with construction. The CONTRACTOR shall ensure that no construction debris, excess excavation, materials or other waste is left on the site.

# 1.06 NOISE CONTROL

- A. Construction involving noisy operations shall be restricted to the hours between 7:00 AM and 8:00 PM as much as possible. Noisy operations shall be scheduled to minimize their duration.
- B. CONTRACTOR shall comply with all local controls and noise level rules, regulations and ordinances which apply to any work performed pursuant to the contract.
- C. Each internal combustion engine, used for any purpose on the job or related to the job shall be equipped with a muffler of a type recommended by the manufacturer. No internal combustion engine shall be operated on the project without a muffler.

#### **\*\*END OF SECTION\*\***

Anola the Vilonoitnetal egal sidT

# SECTION 01700 PROJECT CLOSEOUT

# PART 1 - GENERAL

# 1.01 FINAL CLEANUP

Prior to final acceptance the CONTRACTOR shall clean, remove all refused and unused materials resulting from the work activities within the project limits.

# 1.02 FINAL SUBMITTAL

Prior to final acceptance, the CONTRACTOR shall provide:

1. A letter stating that all work has been substantially completed in accordance with these plans and specifications, and requesting final inspection.

Prior to final acceptance and the initiation of the warranty period , the CONTRACTOR shall provide:

- 1. Prior to final acceptance, the CONTRACTOR shall execute and provide all the necessary supporting documents for the following forms provided by the ENGINEER:
  - a. Consent of Surety to Final Payment.
  - b. Contractor's Affidavit of Payment of Debts and Claims.
  - c. Contractor's Affidavit of Release of Liens.
- 2. Final invoicing.

# 1.03 WARRANTY

The warranty period shall be 12 months from the date of final acceptance by the OWNER.

# **\*\* END OF SECTION \*\***

This Page Intentionally Left Blank

3

# SECTION 02100 SITE PREPARATION

# **PART 1 - GENERAL**

#### 1.01 DESCRIPTION

This section includes labor, materials, equipments and provisions for demolition and disposal of all objectionable material and related work necessary to prepare the site for construction operations. All work shall be in conformance with 2018 ODOT/APWA Standard Specifications for Construction. except as modified herein.

### 1.02 COORDINATION

- A. The CONTRACTOR shall coordinate his operations with affected property owners and utilities likely to be impacted by construction. A partial listing of agencies and utilities follows:
  - 1. Canby Public Works Department, Mr. Jerry Nelzen, Phone 971-253-9173 e for Pavement Striping 503-793-6461
  - Canby Utility Electrical; Mr. Gary Stockwell 503-266-1156 or Cell 971-563-6307
  - 3. Canby Utility Water; Mr. Doug Erkson, phone 503-263-4331 or Cell 971-281-4191 or Bill Makowski, phone 503-263-4309 or cell 971-563-6315
  - 4. DirectLink; Mr. Matt Downs 503-266-8252 or Cell 503-341-4357
  - 5. Wave Broadband; Mr. Travis Edge 503-899-3267
  - NW Natural; Mr. Darrell Hammond 503-585-6611, Ext 8137 or Cell 541-981-0164
  - 7. One Call Notification Center; Phone 1-800-332-2344 or 811
- B. The CONTRACTOR shall be responsible for coordinating his activities with affected utilities and property owners and other adjacent contractors. No additional costs are allowed for delays resulting from a lack of such coordination by the CONTRACTOR.

# 1.03 PROPERTY PROTECTION

A. All structures, utilities, and properties, whether inside or outside the project limits, shall be protected from damage or interruption by the CONTRACTOR'S activities. Responsibility for safety and protection of buildings near or in the project limits are the CONTRACTOR'S. The CONTRACTOR shall repair or replace damaged structures, utilities, and/or properties to the satisfaction of the OWNER.

02100 - 1

The location of underground utilities; power, telephone, cable, gas, etc., if shown on the Design Drawings are approximate. The actual locations may vary from those shown. The CONTRACTOR is responsible for verifying all utility locations. No extra costs are allowed for delays resulting from utility conflicts for which the CONTRACTOR is responsible. On questions regarding the disposition of existing utilities, improvements or buildings, the CONTRACTOR shall contact the ENGINEER for advisement.

PART 2 - PRODUCTS (NOT USED)

# **PART 3 - EXECUTION**

# 3.01 DEMOLITION

B.

- A. Demolition will be as shown on the Drawings or as directed.
- B. Pavement to be removed will be saw cut in straight clean lines.

# 3.02 DISPOSAL OF WASTE MATERIAL

Demolished and waste material shall be disposed of off-site by the CONTRACTOR in accordance with all federal, state, and local regulations.

# **\*\*END OE SECLION\*\***

SECTION 02200 EARTHWORK

# PART 1 - GENERAL

#### 1.01 DESCRIPTION

This section includes all labor, materials, equipment and methods to be used to perform all excavation, foundation preparation, backfilling, slope stabilization, roadway embankment and compaction required as shown on the Drawings in conformance with 2018 ODOT/APWA Standard Specifications for Construction.

# **PART 2 - PRODUCTS**

# 2.01 NATIVE BACKFILL

Native backfill shall be native material which is free of organic materials, non-plastic (no clay), reasonably evenly graded and capable of being compacted with 100% of the material passing a 1" screen.

# 2.02 SELECT BACKFILL

All select backfill shall be crushed rock with a maximum size of 1-1/2", uniformly graded from coarse to fine and complying with the requirements of the 2018 ODOT/APWA Standard Specifications for Construction. All materials shall be subject to the ENGINEER's approval.

# 2.03 EMBANKMENT

The CONTRACTOR shall provide for embankment crushed rock material with a maximum size of 4" pit run rock, uniformly graded from coarse to fine and complying with the requirements of the 2018 ODOT/APWA Standard Specifications for Construction. All materials shall be subject to the ENGINEER's approval.

# PART 3 - EXECUTION

# 3.01 GENERAL

- A. The CONTRACTOR shall perform all excavation of every description and of whatever materials encountered to the depth indicated on the Drawings or as specified below.
- B. The CONTRACTOR shall make proper allowance for the pipe thickness, pipe bedding and, if required, foundation stabilization.

# 3.02 PREPARATION

- A. As specified in Section 02100, Site Preparation, clearing and grubbing shall be completed and existing pavement shall be removed prior to excavating soils.
- B. Erosion Control

.2

- I. The CONTRACTOR is responsible for developing and maintaining an erosion control plan to prevent the transport of soil off-site via wind or rain for the duration of the project. The erosion control measures shall remain in place until all vegetation is established.
- Best Management Practices shall be utilized to control runoff at the site and to protect existing and proposed storm drain facilities. These methods include but are not limited to: hay bales, bio bags, silt fences, berms, stockpile covers, gravel construction entrances, etc. Erosion control measures shall be in conformance with local, state, and federal regulations.
- 3. Temporary erosion control measures shall be provided until the placement of permanent features and/or the establishment of grass. The temporary erosion control measures shall include ground cover or other acceptable means of all disturbed areas caused by the CONTRACTOR'S activities. Temporary erosion control measures shall be in conformance with local, state, and federal regulations.

# 3.03 COMMON EXCAVATION

- A. Excavations shall be classified as either common or rock excavation.
- B. Common excavation is defined as the removal of all material which is not classified as rock excavation.
- C. The limits of excavation shall be kept to a minimum.
- D. The CONTRACTOR shall take necessary precautions to prevent damage to adjacent structures or utilities.
- E. Sheeting and shoring may be necessary or may be required by regulatory authorities in accordance with safety laws.

9

F. CONTRACTOR shall also make provisions to prevent soil or water intrusion into the pipelines prior to completion of installation.

# 3.04 EMBANKMENT CONSTRUCTION

- A. Embankment construction shall include preparation of the areas upon which embankments are to be placed as shown on the construction plans. Placement of fill materials to be performed after the area receiving fill has been stripped from all organic and foreign material.
- B. The CONTRACTOR shall place embankments and fills in horizontal layers of 12 - 18 inches maximum depth and compact each layer to a minimum of 95% of maximum density per AASHTO T-180 standards or approved equivalent within the roadway and right of ways. All embankments and fills outside the roadway and right of ways shall be compacted to a minimum of 90% of maximum density per AASHTO T-180 standards.

# 3.05 ROCK EXCAVATION

A. Rock excavation is defined as a method of removal and not a geological formation, and includes removal of solid material, that in the opinion of the ENGINEER, cannot be removed by nominal 60,000 pound excavating equipment available to the CONTRACTOR in common use for the type of project undertaken, AND requires for its removal drilling and blasting, wedging, sledging, barring or breaking up with additional powered equipment.

No material that can be removed with a power operated excavator of 60,000 pound capacity with rip tooth or similar attachment will be defined as rock excavation.

- B. Boulder removal may be classified as open trench rock removal if at least one dimension of the removed boulder is greater than 3 feet. Concrete and reinforced concrete pipes, structures, thrust blocks or fill required to be removed or demolished will not be considered rock excavation.
- C. Limits of payment for rock excavation shall be in-place cubic yard defined as a maximum of pipe outside diameter plus 30" in width, from the observed level of rock to a trench invert 6" below the pipe invert in depth, and length defined by the extent of the rock. Boulder excavation will be based on the volume of the boulders removed.

# 3.06 FOUNDATIONS

A. No unsuitable material has been identified on-site. Any foundation material required will be paid under a negotiated change order, if required.

When, in the judgment of the ENGINEER, the existing material in the excavation is unsuitable for support, the CONTRACTOR shall excavate below grade, as directed in writing. The CONTRACTOR shall backfill the excavation to subgrade with foundation material and compact in layers not exceeding six inches deep. The foundation material in the trench shall be compacted by machine operated pneumatic or mechanical tampers. Compaction shall be approved by the stite

.9118

# 3.07 DEWATERING

.В.

- A. Removal and disposal of storm water from the excavation shall be the responsibility of the CONTRACTOR as approved by the ENGINEER. Such approval shall not imply any liability of the ENGINEER or OWNER for damage or extra costs incurred by the CONTRACTOR in handling or disposing of storm or tunoff surface water.
- B. Groundwater
- Contractor shall perform dewatering as needed to complete the construction. Costs for dewatering activities shall be considered incidental to the contract.
- 2. Water generated from dewatering activities shall be properly disposed of in accordance with the requirements of the Contract Documents, herein, and local, state, and federal regulations.

# 3.08 HANDLING EXCAVATED MATERIAL

All excavated material not required for backfilling or embankment on-site shall be promptly removed and disposed off-site by the CONTRACTOR in accordance with governing regulations.

# \*\*END OL SECLION\*\*

# SECTION 02500 PAVING AND SURFACING

# PART 1 - GENERAL

### 1.01 DESCRIPTION

- A. This section covers the work and materials for furnishing labor, materials, equipment to install asphaltic concrete surfaces, crack sealing, pavement grinding, pavement striping, existing utility structures adjustments, fabric and Portland Cement products as indicated in the March 15, 2019 Memorandum or as directed in the field by the OWNER. All work shall be in conformance of 2018 ODOT/APWA Standard Specifications for Construction.
- B. Asphalt cement material price escalation/de-escalation adjustments will not be in effect during the life of this contract. All costs will be paid at the bid unit prices per ton of asphalt.

# 1.02 QUALITY ASSURANCE

Work provided under this section shall conform to 2018 ODOT/APWA Standard Specifications for Construction, unless noted otherwise here in.

# **PART 2 - PRODUCTS**

# 2.01 BASE COURSE MATERIAL

Base course material shall be crushed rock aggregate with a maximum size of 1" minus, uniformly graded from coarse to fine. All materials shall comply with section 02630.10 of the 2018 ODOT/APWA Standard Specifications for Construction.

# 2.02 ASPHALT CONCRETE PAVEMENT

Asphalt Concrete shall be <sup>1</sup>/<sub>2</sub>" Asphalt Concrete Pavement complying with the requirements of section 00745 of the 2018 ODOT/APWA Standard Specifications for Construction. The mix shall be level 2 or 3 Performance Grade (PG), 64-22 binder.

# 2.03 PAVEMENT STRIPING

Thermoplastic products shall be from the latest qualified product list as published by ODOT. All legends such as crosswalks, stop bars, bike stencils and arrows shall be type "A" liquid, hot-applied thermoplastic or preformed type "B" thermoplastic film with intermixed traffic paint beads. All installations shall be in accordance with 2018 ODOT/APWA Standard Specifications for Construction, section 00865. All crosswalk bars shall have non-skid finish. Painted striping or legends will not be permitted.

# 2.04 ASPHALT TACK COATING

- A. Asphalt tack coating for overlays shall be Cationic slow to medium setting, low viscosity complying with the requirements of the 2018 ODOT/APWA Standard Specifications for Construction, section 00730.
- B. Sealant coat for fabric placement shall be level 2 Performance Grade (PG), 64-22 binder asphalt cement additives conforming with the requirements of the 2018 ODOT/APWA Standard Specifications for Construction, section 00745.

# 2.05 ASPHALT JOINT CRACK SEALANT

- A. All work shall conform with the requirements of the 2018 ODOT/APWA Standard Specifications for Construction, section 00746.
- B. Hot poured joint/ crack filler shall conform to the requirements of AASHTO M324, Type II (ASTM D 6690, Type II).

# 5.06 GEOTEXTILE FABRIC

- Geotextile fabric for asphalt overlays shall be Petromat Number 4599 by Amoco Fabrics and Fibers Company, Mirafi Mirapave 400 produced by Mirafi Products, Geotex 381 Pave Dry produced by SI Corporation, Terra Tex OL by Hanes Geo Components or approved equal.
- B. Geotex tile fabric for subgrade shall be Tencate Mirafi 180N, 500X, Propex Geotex 801, US Fabrics US 205NW, or approved equal.

# 5.07 CURBS AND SIDEWALKS

- A. Concrete materials shall conform to the requirements of the 2018 ODOT/APWA Standard Specifications for Construction, Sidewalks and Curbs, Section 00759.
- B. Commercial Grade Concrete shall be minimum 3,000 psi compressive strength at 28 days in conformance of 2018 ODOT/APWA Standard Specifications for Construction, Section 00440.

# 2.08 ADA TRUNCATED DOME TEXTURE

A. ADA ramps shall comply with the requirements the Americans with Disabilities Act. Place truncated dome detectable warning texture  $2' \times 4'$  in the lower throat of the ramp only. Arrange domes using in-line pattern only. Color shall be safety yellow.

B. Mats shall be cast-in-place. Cast-in-place tiles shall be Armor-Tile Vitrified Polymer Composite (VPC) product, telephone 1-800-682-2525.

# 2.09 SIGN POSTS

- A. All sign posts shall be a minimum of 2 x 2- inch galvanized "unistrut's telespar" or 12-gauge perforated posts or approved equal. All new posts shall be set with a V-LOC base provided by the CONTRACTOR as shown on the plans. V-Loc shall be Tapco, model 200-VS2
- B. A 2 x 2-inch 14-gauge galvanized "unistrut's telespar" or 12-gauge perforated posts or approved equal shall be used when a combination of signs is more than 36 inches in height. Sign post is 2-inch square tubing and must be embedded a minimum of 12 inches into the base. V-lock base wedges will be provided by the CONTRACTOR.

# PART 3 - EXECUTION

# 3.01 BASE COURSE MATERIAL

All base and leveling courses material shall be compacted to a minimum density of 95% maximum density as determined by AASHTO T-180 and installed in accordance with the requirements of the 2018 ODOT/APWA Standard Specifications for Construction.

# 3.02 ASPHALT CONCRETE PAVEMENT

- A. Preparation, placement and testing shall comply with the 2018 ODOT/APWA Standard Specifications for Construction, Sections 00744 and 00745. Asphalt Concrete Pavement and Asphalt Concrete Pavement Statistical Acceptance.
- B. Mixture shall be machine laid, as practicable, by experienced workman on a compacted crushed rock base. Special care shall be taken to avoid segregation of materials and to maintain specified grades, thickness and smoothness.
- C. All valve boxes, manhole frames, catch basin gratings and other utility appurtenances located within paved areas shall be set or raised to finish grade, unless otherwise noted. All surfaces shall be tack coated prior to paving.
- D. The final surface shall be smooth, true to grade, free draining and free of all surface defects.
- E. Repairs to cut or damaged asphaltic concrete surfaces shall be repaired as follows:
  - 1. All AC surfaces at the match lines shall all be saw cut. All edges shall be sealed and sanded upon completion.

لري

5

- Unless directed otherwise by the EUGINEER, the new pavement shall have the same plan grade as the pavement that existed prior to construction.
- 3. Patches must be diamond cut and paved to conform to adjacent existing pavement and provide a smooth continuous surface.
- F. Scheduling- The CONTRACTOR shall schedule the construction activity to provide a minimum of 10 days notice to the OWNER, and accomplish all work within the specified performance period.
- G. The exact limits of pavement removal shall be determined in the field by the around the perimeter of the area to be removed. Cuts shall be made with a power saw or other approved methods that leaves a straight and vertical face.
- H. After cutting, existing pavement shall be excavated as required. Any damage to adjoining pavement or the underlying base course, as a result of either the cutting or removal operation, will be repaired by the CONTRACTOR at no cost to the OWNER.
- I. Excavate and dispose all existing asphalt and base course in a permitted waste disposal site. Disposal must be according to federal, state and local regulations.
- Place an engineered geotextile fabric material over the entire excavated area prior to the placement of the base course. Base course material shall be as described above in section 2.01 and compacted as per section 3.01.
- K. Before replacement of the asphalt surface, apply a tack coat to all vertical surfaces of the existing pavement and allow the tack coat to cure. Compact the newly replaced asphalt surface with a vibratory compactor, either by hand or by rolling to achieve a smooth, uniform surface level with the adjoining pavement.

# 3.03 PAVEMENT STRIPING

- A. All pavement striping alignment and legend locations shall match and duplicate the existing striping and legend locations. No striping plan is provided other than to match the existing.
- B. All striping shall comply with the requirements of 2018 ODOT/APWA Standard Specifications for Construction.
- C. All pre-markers for pavement markings shall be laid out in the field by the CONTRACTOR and approved by the ENGINEER prior to final placement of striping and symbols.

# 3.04 ASPHALT TACK COATING

The asphalt tack coating binder for overlays shall be applied with a pressure distributor at rate of 0.05 to 0.10 gallons per square yard to ensure bonding between the existing asphalt and the new asphalt.

# 3.05 ASPHALT CRACK SEALANT

- A. All work shall conform to the requirements of the 2018 ODOT/APWA Standard Specifications for Construction, section 00746.
- B. As directed by the City or their representative in the field, the CONTRACTOR shall clean all cracks/ joints designated for sealing of loose and foreign matter. Use hot lance to perform this cleaning. Use this wand to both clean and dry the crack just prior to sealing.
- C. Install the sealant so that the in-place sealant is well bonded to the pavement and free of voids or entrapped air.
- D. All cracks shall be sealed from the bottom up in a neat manner, the complete surface of the sealant material shall be flushed to 3/16" below the adjacent pavement surface. All low areas must be spotted and refilled.
- E. All sealant material shall be leveled and flushed to the surface with a "V" shaped squeegee device. Squeegee excess material so that it doesn't exceed 1<sup>1</sup>/<sub>2</sub>" on either side of the crack. All excess sealant shall be distributed over the crack in return motion.
- F. Cover the sealed cracks/ joints with clean sand material, then sweep the pavement and leave in clean condition.
- G. Do not allow any traffic or construction equipment on the new sealed cracks/ joints for at least one (1) hour after the sealant placement and refilling has been completed.

# 3.06 GEOTEXTILE FABRIC

All geotextile fabric material shall be laid straight along the subgrade stabilization allowing 6-inches minimum overlap over each fabric panel.

# 3.07 CURBS AND SIDEWALKS

A. Curbs and sidewalks construction shall comply with the requirements of the 2018 ODOT/APWA Standard Specifications for Construction.

- B. Contraction joints spacing to be no more than 15 feet. It shall be no less than 1/3" nor more than 1/4" wide.
- C. Surface finish shall be true and straight, free from humps, sags or other irregularities.
- D. Cure concrete by covering with burlap, canvas, sand or other acceptable material and keep moist for a sufficient amount of time to avoid cracking. Curing compounds may be used if applied immediately to the concrete surfaces.
- E. All flat work shall be scored in 5-foot squares typical and finished with a small radius tool, with  $1\sqrt{2}$  to 2" troweled shiner on all four edges of each panel, including at the curb line and back line of walk. Finish field with light broom finish.
- F. Curbs and sidewalks are permitted to be poured separately and cure a minimum of 24 hours prior to sidewalk. Requests to modify or deviate from this standard shall be submitted in advance with approval of the City Public Works Director or Supervisor.

# 3.08 ADA TRUNCATED DOME TEXTURE

- A. During surface preparation and mat installation procedures ensure that the installation meets all safety guidelines in accordance with the applicable industry and 2018 ODOT/APWA Standard Specifications for Construction.
- B. All truncated dome materials shall be installed in strict accordance with the guidelines set by the manufactures.

# STROG NHIR 60.5

All sign heights and locations shall conform to the requirements of the 2018 ODOT/APWA Standard Specifications for Construction and the Manual on Uniform Traffic Control Devices, latest edition and shall be inspected by the City.

# **\*\*END OE SECLION\*\***

7)

City of Canby - 2019 Street Asphalt Overlay List .ocation End Location Length (FT) Width (FT) Area (Sq. Ft.) Overlay Thickness	N. Juniper Street 265 58 15,370	N. Holly Street 255 58	N. Locust Street 1,040 50 52,000	Alley 400 40 16,000	N. Pine Street 1,200 40 48,000	South Side of Train Depot 430 24-36 13,000	SE 2nd Avenue 210 40 8,400	Cul De Sac 585 38 28,700	S. Fir Street 850 40 34,000	S. Teakwood Street 2,650 44 116,600	Cul De Sac 2,600 32 85,500	SE 13th Avenue 2,675 40 107,000		
City of Canb Begin Location	N. Ivy Street	NV ZNA Avenue N. Ivy Street N. Holly	N. Ivy Street		N. Pine Street		HWY 99 E	644 Residence	S. Elm Street	Nue S. LuPine Street	SE 13th Avenue	S. Fine Street S. Lownship Road SE 13th		

539,360 2.49 miles 13,160 Total **CRACK SEALING** 

NE Territorial Road SE 13th Avenue

N Holly Street S Lupine Street

Hwy 99E S Redwood Street

47,000 LF 3,000 LF

# ( ( \$ . Č

Reserved to other other

the first of the summary denomination of

لات \_\_\_\_\_ J \_J لد ب 2 ني ، ل: \_ . ل

March 15, 2019

# **MEMORANDUM**

- TO: Mr. Jerry Nelzen City of Canby
- FROM: Hassan Ibrahim, P.E. Curran-McLeod Consulting Engineers, Inc.

# RE: CITY OF CANBY 2019 STREET RESURFACING WORK SCOPE

This memorandum stipulates surface treatment recommendations on the streets depicted below based on our field findings. These recommendations were summarized to assist and guide the contractor in bidding and completing the work on each street segment and are as follows:

**NE 2<sup>ND</sup> AVENUE & NW 2<sup>ND</sup> AVENUE:** (from N Holly Street to N Juniper Street, the total estimated length is 550 lineal feet by 58 feet wide curb to curb):

- 1. Remove and replace an estimated 19'x2' concrete/pavers panel on each side of the midblock crossing or 38 square feet or (total 228 square feet) at the three crossings and install thirty (30) truncated dome mats. Remove and replace 4 lineal feet of existing curb and 7'x4' concrete ramp at the north and south ramps on each side of S Ivy Street (total 16 lineal feet of curb and 112 square feet of sidewalks) and reinstall a total of new four (4) ADA ramps with truncated dome mat. Remove and replace 4 lineal feet of existing curb and 7'x4' concrete ramp at the north and south ramps on east side of N Holly Street (total 8 lineal feet of curb and 56 square feet of sidewalks) and reinstall a total of new two (2) ADA ramps with truncated dome mat. Remove and replace 10 lineal feet of existing curb and 5'x4' concrete ramp near the two new ADA parking stalls as designated in the field (total 20 lineal feet of curb and 40 square feet of sidewalks) and reinstall a total of new two (2) ADA ramps with truncated dome mat. The prime contractor shall coordinate a meeting with the concrete contractor to determine the exact limits in the field prior to construction. See details.
- 2. Prior to the overlay application, grind existing asphaltic concrete to 2" depth between the project limits the **total grinding area is 23,753 square feet** and remove additional asphalt to 2" +/- depth by other excavation means for a **total excavated area of 6,386 square feet**. The contractor will be required to determine if needed to provide a durable, smooth and clean matching joint and coordinate the grinding limits with the Engineer.

3. Protect and adjust to finish grade; five (5) manhole ring risers and one (1) water valve box.

4. Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is **30,140 square feet**. Maintain 6-inches minimum curb exposure.

1

- 5. Restore continental sidewalk crossing (9' long x 2' wide) at the existing crossing near the west entrance to Credit Union, (126 square feet). See "Staggered Continental Crosswalk" detail <u>CW-SC</u>.
- 6. Restore 4" wide white thermoplastic parking stripes (1,100 lineal feet) including striped ADA islands. We have counted a total of 63 diagonal parking stalls @ 45 degree angle, stall typical dimensions are (9' x 15').
- 7. Restore two (2) thermoplastic international ADA legends, white symbol with blue background. See detail.

**WE 3<sup>RD</sup> AVENUE:** (from N. Ivy Street to N. Locust Street, the total estimated length is 1,040 lineal feet by 50 feet wide curb to curb):

- Remove and replace an estimated **30 lineal feet** of existing curb and **325 square feet** of existing sidewalk at the northeast and southeast corners of the intersection with N. Ivy Street and reinstall a total of new **two** (2) ADA ramps with truncated dome mat. Also at the intersection with N Juniper Street, remove and replace existing **eight** (8) 4'x4' concrete and reinstall a total of new **eight** (8) ADA ramps with truncated dome mat. The prime and reinstall a total of new **eight** (8) ADA ramps with truncated dome mat. The prime for the field prior to include **30 lineal feet of curbs** and **130 square feet of sidewalks** sections to include the eight (10) A'x4' concrete intersection with truncated dome mat. The prime intersection for the field prior to continue the eight (10) ADA ramps with truncated dome mat. The prime in the field prior to continue the exact limits in the field prior to construction. See details.
- 2. Remove and replace existing sign posts, salvage and transfer existing signs on new posts and install V-loc bases for all the posts located in the concrete sidewalk, **one** (1) each.
- 3. Prior to the overlay application, a 6-foot wide AC grinding panel to  $\Sigma$ " depth along the gutter and N Locust Street (2,080 lf) and a butt joint at the intersections with N Ivy Street and N Locust Street (285 lf) or as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. Estimated total 2,400 lineal feet.
- 4. Protect and adjust to finish grade; two (2) manhole ring risers, five (5) water valve boxes and one (1) NW Natural valve.
- 5. Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is 52,000 square feet. Maintain 4-inches minimum curb exposure.
- 6. Restore all existing striping to pre-paving conditions to include 2-4" double wide yellow thermoplastic (total 500 lineal feet); see "Narrow Double No-Pass Two 4" Yellow" detail <u>MD</u>, 1-4" wide skip yellow thermoplastic (total 170 lineal feet) see "Yellow Broken Line

Supplementation Reflectors with 4" Yellow Broken Lines" <u>detail YB/R-40</u>, 1-8" wide white thermoplastic stripe (**total 1,900 lineal feet**), see "8" White Line" <u>detail W-2</u>, 1-4" wide white thermoplastic stripe (**total 1,500 lineal feet**), see "4" White Line" <u>detail W, six (6)</u> white thermoplastic bike stencil, see "Bike Lane Standard Stencil (white)", <u>detail BS</u>, **two** (2) white thermoplastic left turn arrow, see "Left Turn Arrow (white)", <u>detail LA</u>, thermoplastic continental sidewalk crossing (9' long x 2' wide) at the intersection with N Ivy Street (**504 square feet**), see "Staggered Continental Crosswalk", <u>detail CW-SC</u>.

**<u>N. JUNIPER STREET</u>**: (from NE 4<sup>th</sup> Avenue to the Alley, the total estimated length is 400 lineal feet by 40 feet wide curb to curb):

- Remove and replace existing two (2) 4'x4' concrete ramp/sidewalk sections to include 10 lineal feet of curbs and 32 square feet of sidewalks and reinstall a total of new two (2) ADA ramps with truncated dome mat. The prime contractor shall coordinate a meeting with the concrete contractor to determine the exact limits in the field prior to construction. See details.
- 2. Prior to the overlay application, grind a 6-foot wide AC panel to 2" depth along the gutter lines along both sides of the streets and a butt joint at the match line with NE 4<sup>th</sup> avenue and the match line at the Alley Avenue (240 lf) or as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. **Estimated total 1,050 lineal feet**.
- 3. Protect and adjust to finish grade; one (1) sewer cleanout ring riser.
- 4. Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is **16,000 square feet**. Maintain 4-inches minimum curb exposure.
- 5. Install thermoplastic continental sidewalk crossing (9' long x 2' wide) at the crossings, (220 square feet). See "Staggered Continental Crosswalk", <u>detail CW-SC</u>.

**NE 13<sup>Th</sup> CRCLE:** (from N. Pine Street to Cul-De-Sac, the total estimated length is 1,200 lineal feet by 40 feet wide curb to curb):

- 1. Prior to the overlay application, grind a 6-foot wide AC panel to 2" depth along the gutter lines along both sides of the streets and a butt joint at the match line with N. Pine Street (215 lf) or as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. **Estimated total 2,600 lineal feet**.
- 2. Protect and adjust to finish grade; four (4) manhole riser rings and five (5) water valve boxes.

.4

- Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is 48,000 square feet. Maintain 4-inches minimum curb exposure.
- 4. Install 12" wide thermoplastic Stop Bar", detail S. long) at the intersection with N. Pine Street. (total area 12 square feet). See "Stop Bar", detail S.

**W. PINE STREET:** (from SE 4<sup>th</sup> Avenue (Train Depot) to the end of curvature, the total estimated length is 430 lineal feet by 24 to 36 feet wide curb to curb):

- 1. Protect and adjust to finish grade; two (2) water valve boxes.
- 2. Asphaltic concrete pre-leveling is needed. The limits can be determined in the field by the Engineer and the Contractor.
- 3. Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is 13,000 square feet.
- Restore all existing striping to pre-paving conditions to include 12" wide thermoplastic Stop Bar". Bar (24' long) at the intersection with NE  $4^{th}$  Avenue (**total 24 square feet**), See "Stop Bar" detail <u>S</u>, 2-4" double wide yellow thermoplastic (**total 860 lineal feet**); see "Varrow Double Wo-Pass Two 4" Yellow" detail <u>W</u>-2, **one** (**1**) white thermoplastic right turn arrow, see "Right Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic right turn arrow, see "Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic right turn arrow, see "Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic right turn arrow, see "Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic left turn arrow, see "Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic right turn arrow, see "Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic left turn arrow, see "Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic right turn arrow, see "Turn Arrow (**1**) white thermoplastic right turn arrow, see "Right Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic left turn arrow, see "Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic left turn arrow, see "Right Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic left turn arrow, see "Right Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic left turn arrow, see "Right Turn Arrow (white)", detail <u>RA</u>, **one** (**1**) white thermoplastic left turn arrow, see "Right Turn Arrow (white)", detail RA, **1** and **1**

**S. JUMPER STREET:** (from Hwy 99 E to SE  $2^{nd}$  Avenue, the total estimated length is 210 lineal feet by 40 feet wide curb to curb):

- 1. Remove and replace an estimated **40 lineal feet** of existing type "C" curb and an estimated **210 square feet** of sidewalk at the northeast and northwest corners of the intersection with  $SE 2^{nd}$  Avenue and reinstall **two** (2) ADA ramps with truncated dome mat. The prime contractor shall coordinate a meeting with the concrete contractor to determine the exact limits in the field prior to construction. See details.
- 2. Prior to the overlay application, grind a 6-foot wide AC panel to 2" depth along the gutter lines along both sides of the streets and a butt joint at the match line with Hwy 99E and SE 2<sup>nd</sup> Avenue (240 If) or as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. **Estimated total 650 lineal feet**.

- 3. Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is **8,400 square feet**. Maintain 4-inches minimum curb exposure.
- 4. Install 12" wide thermoplastic Stop Bar (12' long) at Hwy 99E and at SE 2nd Avenue. (total area 24 square feet). See "Stop Bar", <u>detail S</u>.

**S. KNOTT STREET/SE 7<sup>TH</sup> AVENUE:** (from 644 Residence to Cul-De-Sac, the estimated length is 585 lineal by 38 feet wide curb to curb and 45-foot radius Cul-De-Sac):

- 1. Pavement reconstruction of the following areas on SE 7<sup>th</sup> Avenue near the intersection with S. Larch Avenue (58'x17'). The total pavement reconstruction area is **986 square feet**.
- 2. Remove and replace an estimated **50 to 60 lineal feet** of existing type "C" curb and sidewalk at the northeast and northwest corners at the intersection of SE 7<sup>th</sup> Avenue and S Larch Street and reinstall **two (2) ADA ramps** with truncated dome mats. Restore the existing block wall at the back of the sidewalk. The prime contractor shall coordinate a meeting with the concrete contractor to determine the exact limits in the field prior to construction. See details.
- 3. Remove and replace existing sign posts, salvage and transfer existing signs on new posts and install V-loc bases for all the posts located in the concrete sidewalk, **two (2) each**.
- 4. Prior to the overlay application, a 6-foot wide AC grinding panel to 2" depth along the gutter lines on both sides of the street (1,380 lf)and a butt joint at the match line/the new joint (644 residence, 120 lf) <u>or</u> as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. **Estimated total 1,500 lineal feet.**
- 5. Protect and adjust to finish grade; five (5) manhole riser rings and five (5) water valve boxes.
- 6. Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is **28,700 square feet**. Maintain 4-inches minimum curb exposure.

**SE 11<sup>TH</sup> AVENUE:** (from S Elm Street to S Fir Street, the estimated length is 850 lineal by 40 feet wide curb to curb):

1. Remove and replace an estimated **50 to 60 lineal feet** of existing type "C" curb and sidewalk at the northeast and northwest corners at the intersection of S Fir Court and SE 11<sup>th</sup> Avenue and reinstall **two (2) ADA ramps** with truncated dome mats. The prime contractor shall coordinate a meeting with the concrete contractor to determine the exact limits in the field prior to construction. See details.

.1

- 2. Remove and replace existing sign posts, salvage and transfer existing signs on new posts and install V-loc bases for all the posts located in the concrete sidewalk, **one (1) each**.
- 3. Prior to the overlay application, a 6-foot wide AC grinding panel to 2" depth along the gutter lines on both sides of the street (1,700 lf) and a butt joint at the intersections with S Elm Street and S Fir Street (270 lf) or as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. Estimated total 2,000 lineal feet.
- 4. Protect and adjust to finish grade; five (5) manhole riser rings, six (6) water valve boxes and one (1) cleanout riser ring.
- Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is 34,000 square feet. Maintain 4-inches minimum curb exposure.

**SE 13<sup>TH</sup> AVENUE:** (from S. LuPine Street to S. Teakwood Street, the total estimated length is 2,650 lineal feet by 44 feet wide curb to curb and 40 feet curb to edge of pavement):

- Pavement reconstruction of the following areas (some are between S Ivy Street and S Lupine (50'x2.5'), (5'x6'), (5'x5'), (27'x5'), (27'x5'), (27'x5'), (7'x6'), (5'x6'), (5'x5'), (5'x6'), (7'x4'), (17'x4'), (4.5'x2'), (50'x2.5'), (5'x7.5'), (27'x2.5'), (27'x2.5'), (9'x5.5'), (12'x6.5'), (12'x6.5'), (73'x4'), (12'x6'), (12'x6.5'), (15.5'x4'), (
- 2. Remove and replace an estimated **32 square feet** of asphalt with (4) of (2'x4') concrete pade with truncated dome mat at two islands in the north and the south directions at the intersection with Pine Street and Ponderosa Street. The prime contractor shall coordinate a meeting with the concrete contractor to determine the exact limits in the field prior to construction.
- 3. Prior to the overlay application, grind a 6-foot wide AC panel to 2" depth along the gutter line on both sides of the street and a butt joint at the match line with S Lupine just east of the Speed Hump Street and S Teakwood Street (270 lf) or as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. Estimated total 5,570 lineal feet.

3

فمذ

4. Protect and adjust to finish grade; seven (7) manhole riser rings, fourteen (14) water valve boxes and one (1) NW Vatural valve.

- 5. Place 2-inches thick of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is **116,600 square feet**. Maintain 4-inches minimum curb exposure.
- 6. Restore all existing striping to pre-paving conditions to include 2-4" double wide yellow thermoplastic (total 5,625 lineal feet); see "Narrow Double No-Pass Two 4" Yellow" detail <u>ND</u>, 1-8" wide white thermoplastic stripe (total 4,960 lineal feet), see "8" White Line" detail <u>W-2</u>, 1-4" wide white thermoplastic stripe (total 1,300 lineal feet), see "4" White Line" detail <u>W</u>, five (5) white thermoplastic bike stencil, see "Bike Lane Standard Stencil (white)", detail BS, one (1) white thermoplastic SLOW legend, see detail, one (1) white thermoplastic HUMP legend, see detail, thermoplastic continental sidewalk crossing (9' long x 2' wide) at the intersection with Pine Street and Ponderosa Street (432 square feet), see "Staggered Continental Crosswalk", detail CW-SC.

**S. ELM STREET:** (from SW 13<sup>th</sup> Avenue to Cul-De-Sac, the estimated length is 2,600 lineal by 32 feet wide curb to curb and 45-foot radius Cul-De-Sac):

- 1. Prior to the overlay application, a 6-foot wide AC grinding panel to 2" depth along the gutter lines on both sides of the street (4,350 lf) and a butt joint at The intersection with SW 13<sup>th</sup> Avenue (150 lf) No gutter grinding at approximately 50 feet south of the entrance to Village on the Lochs to approximately 50 feet east of the first driveway on the right side of the street **or** as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. **Estimated total 4,500 lineal feet.**
- 2. Protect and adjust to finish grade; ten (10) manhole riser rings and twenty-two (22) water valve boxes and one utility vault.
- 3. Place 3-inches thick <u>placed in two lifts</u> of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is **85,500 square feet**. Maintain 4-inches minimum curb exposure.
- 4. Install thermoplastic continental sidewalk crossing (9' long x 2' wide) at the crossings, (**126** square feet). See "Staggered Continental Crosswalk", <u>detail CW-SC</u>.

**<u>S. PINE STREET</u>**: (from SE Township Road to SE 13<sup>th</sup> Avenue, the total estimated length is 2,675 lineal feet by 40 feet wide curb to curb):

1. Remove and replace an estimated **400 to 480 lineal feet** of existing type "C" curb and sidewalk and construct an estimated **320 lineal feet** of type "C" short curb at each of the following intersections: S Oak Place (N & S), SE 7<sup>th</sup> Avenue, SE 8<sup>th</sup> Avenue, SE 9<sup>th</sup> Avenue and SE 10<sup>th</sup> Avenue. A total of **sixteen (16) ADA ramps** with truncated dome mat. The prime

contractor shall coordinate a meeting with the concrete contractor to determine the exact limits in the field prior to construction. See details.

- 2. Prior to the overlay application, grind a 6-foot wide AC panel to 2" depth along the gutter lines along both sides of the streets and a butt joint at the match line with Township Rd and SE 13<sup>th</sup> Avenue (240 If) or as determined in the field. The contractor is required to determine where additional grinding may be needed and coordinate the grinding limits with the Engineer. Estimated total 5,570 lineal feet.
- 3. Protect and adjust to finish grade; fifteen (15) manhole ring risers and seventeen (17) water valve boxes.
- 4. Place 3-inches thick placed in two lifts of new AC overlay over the entire street width between the curb lines. The estimated AC overlay area is 107,000 square feet. Maintain 4-inches minimum curb exposure.

j

5. Install 12" wide thermoplastic Stop Bar (12' long) on the north and south side of the intersection with SE  $10^{th}$  Avenue. (total area 24 square feet). See "Stop Bar", <u>detail S</u>.

# **LOGGING ROAD TRAIL:**

1. Prep and pave 1" to 1.5" depth of asphaltic concrete at the approach to SE 13<sup>th</sup> Avenue with Sequoia Parkway. The total pavement area is 30'x9' or (270 square feet).

#### :eston suosnellsseiM

- The quantities listed in the Basic Bid schedule are estimated and may fluctuate. The payment for each item will be made based on the exact field measurements at the contract bid unit prices.
- 2. All restriping work shall be performed to match prepaved conditions, adjoining existing atriping to remain must be protected during the asphalt overlay placement. The contractor must inventory lane configurations, legends, lane widths, stripe sizes and kind, preliminary layout in the field and coordinate all the restriping work with the City by contacting Mr. Jerry Nelzen.
- Asphalt preleveling work will be performed if deemed necessary in the field at the bid unit price.
- 4. Asphalt tack coat shall be applied to every street to be overlaid prior to HMAC placement.
- 5. All asphalt joints at the match lines must be sand sealed after completion of the work.

- 6. The exact work limits as identified in the "Bid Proposal" will be identified in the field by the Owner or Engineer prior to the start of any work. Contractor to coordinate.
- 7. Where in the judgement of the contractor or the engineer, a 6' wide AC grinding panel may be needed to maintain curb and driveway exposure or a better joint connection, the contractor must coordinate limits by contacting the Engineer at 503-684-3478. The exact quantities will be adjusted based on the field findings and paid at the bid unit prices.



J:\Canby\1009\1009-STREET MAINT\2019 STREET MAINTENANCE\2019-STREET MAINTENANCE.dwg, 3/26/2019 12:12:38 PM, DWG To PDF.pc3





AltaLink B8055 PS (11x17).pc3 3/21/2019 8:17:23 AM, Xerox DETAIL.dwg, STREET MAINTENANCE J:\Canby\1009\1009-STREET MAINT\2019 STREET MAINTENANCE\2019

	DATMARCH/20	019	
	P/N 10	009	
		HAI	
	D/S	JVB	
1	CADI 1009-	-C1	
1			
J			
1	$\mathbf{i}$		

